J. Inst. Math. Jussieu (2023), **22**(4), 1655–1702 1655 doi:10.1017/S1474748021000499 © The Author(s), 2021. Published by Cambridge University Press.

AUTOMORPHIC LEFSCHETZ PROPERTIES FOR NONCOMPACT ARITHMETIC MANIFOLDS

ARVIND N. NAIR D AND ANKIT RAI

School of Mathematics, Tata Institute of Fundamental Research, Mumbai 400005, India (arvind@math.tifr.res.in; ankit@math.tifr.res.in)

(Received 15 May 2020; revised 15 September 2021; accepted 16 September 2021; first published online 18 October 2021)

Abstract We prove the injectivity of Oda-type restriction maps for the cohomology of noncompact congruence quotients of symmetric spaces. This includes results for restriction between (1) congruence real hyperbolic manifolds, (2) congruence complex hyperbolic manifolds, and (3) orthogonal Shimura varieties. These results generalize results for compact congruence quotients by Bergeron and Clozel [Quelques conséquences des travaux d'Arthur pour le spectre et la topologie des variétés hyperboliques, *Invent. Math.* **192** (2013), 505–532] and Venkataramana [Cohomology of compact locally symmetric spaces, *Compos. Math.* **125** (2001), 221–253]. The proofs combine techniques of mixed Hodge theory and methods involving automorphic forms.

Keywords: automorphic Lefschetz properties; arithmetic manifolds

2020 Mathematics subject classification: Primary 11F75 Secondary 11G18; 14G35

Contents

1	Introduction		
2	2 Preliminaries		
3	Restriction between Shimura varieties		
4	Lie-algebra cohomology computations		
5	Congruence real hyperbolic manifolds		
6	Congruence complex hyperbolic manifolds		
7	Orthogonal Shimura varieties		
A		me facts about the L^2 cohomology of arithmetic man- olds	1693
A	ppendix B Re	esidual Eisenstein cohomology in corank 1	1697

Appendix C Chern classes of automorphic vector bundles	1699		
Acknowledgements			
References	1700		

1. Introduction

1.1. Main results

Let G be a connected semisimple algebraic group over \mathbb{Q} and X be the symmetric space of $G(\mathbb{R})$. We write $G(\mathbb{R})^{nc}$ for the product of the noncompact factors of $G(\mathbb{R})$ and d_G for the (real) dimension of X.

For congruence subgroups $\Gamma \subset G(\mathbb{Q})$, we consider the quotients $M_{\Gamma} = \Gamma \setminus X$ and their cohomology groups $\mathrm{H}^{i}(M_{\Gamma})$ with complex coefficients. The direct limit

$$\mathrm{H}^{i}(\mathscr{M}_{G}) := \mathrm{colim}_{\Gamma}\mathrm{H}^{i}(M_{\Gamma})$$

is a $G(\mathbb{Q})$ -module using pullback by the isomorphisms $M_{g\Gamma g^{-1}} \to M_{\Gamma}$ induced by g^{-1} on X.

For a semisimple subgroup $H \subset G$, let $\Gamma_H = \Gamma \cap H(\mathbb{Q})$ and $M_{H,\Gamma_H} = \Gamma_H \setminus X_H$. The totally geodesic embedding $X_H \subset X$ induces a proper map

$$M_{H,\Gamma_H} \to M_{\Gamma}$$

Pullback in cohomology defines an $H(\mathbb{Q})$ -equivariant map $\iota^* : \mathrm{H}^*(\mathcal{M}_G) \to \mathrm{H}^*(\mathcal{M}_H)$, and composing with the action of $G(\mathbb{Q})$ gives a map

$$\operatorname{Res}: \operatorname{H}^*(\mathscr{M}_G) \longrightarrow I_H^G \operatorname{H}^*(\mathscr{M}_H).$$

The target of Res (defined in §2.4) is a certain induced module contained in the product $\prod_{g \in G(\mathbb{Q})} H^*(\mathcal{M}_H)$, so that concretely we have that $\operatorname{Res}(\alpha) \neq 0$ if and only if $\iota^*(g^{-1} \cdot \alpha) \neq 0$ for some $g \in G(\mathbb{Q})$ – that is, some Hecke translate of α restricts nontrivially to \mathcal{M}_H .

Theorem 1.1. Suppose that $H \subset G$ are semisimple groups of the same \mathbb{Q} -rank and that $H(\mathbb{R})^{nc} \subset G(\mathbb{R})^{nc}$ is one of the following embeddings:

- (1) $SO(1,c) \subset SO(1,d)$ (the real hyperbolic case), with neither H nor G a triality form;
- (2) $SU(1,m) \subset SU(1,n)$ (the complex hyperbolic or ball quotient case);
- (3) $SO(2,m) \subset SO(2,n)$ (the orthogonal Shimura variety case).

Then the map Res: $\mathrm{H}^*(\mathscr{M}_G) \longrightarrow I_H^G \mathrm{H}^*(\mathscr{M}_H)$ is injective in degrees $< d_H/2$ (and also in degree $i = d_H/2$ in the case where $SO(1,c) \subset SO(1,d)$ with c even).

This automorphic Lefschetz property is well known if G is anisotropic (or equivalently, M_{Γ} is compact): the injectivity in case (1) was proved in [12], and in cases (2) and (3) it was proved in [47] in degrees $i \leq d_H/2$. In the noncompact situation, case (1) can be proved by adapting the methods of [7, 12] with some care, and case (2) was proved in [36] (and [13]), so that the most interesting new case is (3). It includes, for example, the most basic orthogonal Shimura varieties arising from quadratic forms over \mathbb{Q} of signature (2, n) over

1656

1700

 \mathbb{R} with $n \geq 4$. The treatment of the 'missing' degree $i = d_H/2$ in the noncompact SU(1,n)and SO(2,n) cases requires arithmetic information, and we leave it for another occasion. (Note that restriction in degree 2 from SO(2,3) to SO(2,2), which is not covered by Theorem 1.1, was treated by Weissauer [51] using a detailed automorphic understanding of H^2 of Siegel threefolds.)

For the inclusion $SO(1,n) \subset SU(1,n)$ we have the following, which is [12, Theorem 1.7] in the compact case. For i < n, the group $\mathrm{H}^{i}(M_{\Gamma})$ carries a pure Hodge structure of weight i, and we set $\mathrm{H}^{i,0}(\mathcal{M}_{G}) := \mathrm{colim}_{\Gamma}\mathrm{H}^{i,0}(M_{\Gamma})$.

Theorem 1.2. Suppose that $H \subset G$ are of the same \mathbb{Q} -rank, $H(\mathbb{R})^{nc} \subset G(\mathbb{R})^{nc}$ is $SO(1,n) \subset SU(1,n)$, and H is not a triality form. Then $\operatorname{Res} : \operatorname{H}^{i,0}(\mathscr{M}_G) \longrightarrow I_H^G \operatorname{H}^*(\mathscr{M}_H)$ is injective in degrees $i \leq n/2$.

We will discuss the proofs in $\S1.3$.

1.2. Some history

Restriction maps between congruence quotients have been studied by numerous authors for almost 40 years, starting with the pioneering work of Oda [38]. We refer the reader to the surveys [10, 48] for a discussion of this work, and restrict ourselves here to a brief review of the history of immediate relevance to us.

The first result of this type was proved by Oda [38], who introduced the restriction map Res and proved Theorem 1.1 for $SU(1,m) \subset SU(1,n)$ in degree i = 1. Weissauer [51] then proved the Lefschetz property for $SO(2,2) \subset SO(2,3)$ in degree i=2. Motivated by the general conjecture (of Langlands, Kottwitz, and Arthur) on the Galois representations appearing in the cohomology of Shimura varieties, Arthur [1, §9] raised the question of whether the nonprimitive cohomology of Shimura varieties can be related to smaller Shimura varieties. Harris and Li [28] applied the Burger–Sarnak [18] method to prove the Lefschetz property in degree i = 2 in the cases of compact complex hyperbolic and orthogonal Shimura varieties – that is, cases (2) and (3). They also conjectured injectivity in degrees $\leq d_H/2$ in these cases and showed that in case (2) it would follow from Arthur's conjectures [1] on the discrete spectrum. They also asked (when $d_G = d_H + 2$ in cases (2) and (3)) whether a linear combination of Hecke translates of the class of the divisor $M_{H,\Gamma}$ is the class of an ample divisor. Venkataramana [47] showed that this is true in cohomology rather than on the level of cycles – that is, a linear combination of translates of the cycle class in $\mathrm{H}^2(M_{\Gamma})$ is the hyperplane cohomology class in the Baily-Borel projective embedding – and used this to prove the conjecture of [28] – that is, Theorem 1.1 in compact cases (2) and (3).

The automorphic approach of [28] was taken up by Bergeron and Clozel [7, 8, 11, 12], who made the remarkable discovery that Lefschetz properties hold for congruence hyperbolic manifolds (i.e., case (1) of Theorem 1.1) even though there is no complex structure available. This allows for a common approach to Lefschetz properties in different contexts, by using the Burger–Sarnak method to reduce them to uniform (in the level Γ) bounds for the nonzero eigenvalues of the Laplacian on forms on the smaller locally symmetric space. This eigenvalue bound was then deduced in case (1) in [12] from Arthur's endoscopic classification [2] of automorphic forms on orthogonal groups, completing the proof of Theorems 1.1 and 1.2 in the compact case.

In the noncompact case it is less clear what should be true, although the analogues for singular varieties of the Lefschetz theorems in [24] are suggestive. Moreover, since the cohomology of noncompact quotients is influenced by the behavior of *L*-functions (for example, through Eisenstein series constructions of cohomology), one expects that the question is more subtle, and this is reflected by the omission of $i = d_H/2$ in cases (2) and (3) for now.

The complex hyperbolic case of Theorem 1.1 was proved in [13, 35, 36]; here we will prove the rest of the theorem, with the case of orthogonal Shimura varieties being the main new result. In fact, our proof shows that the Lefschetz property for congruence real hyperbolic groups arises as a sort of *local* Lefschetz property at infinity for the noncompact case of orthogonal Shimura varieties. We will comment further on this later.

1.3. On the proofs

We discuss the proofs of the main theorems and some intermediate results proved along the way.

There are, roughly speaking, three types of arguments involved:

- (a) automorphic arguments mainly the Burger–Sarnak method, as in [7, 12, 28], but also rank 1 residual Eisenstein cohomology;
- (b) geometric arguments the use of cycle classes, as in [47], and mixed Hodge theory and compactifications, as in [35, 36]; and
- (c) elementary arguments with Lie-algebra cohomology, as in [36].

The proof of Theorem 1.1 in the different cases uses these ingredients differently: Case (1) uses (a) and (c), case (2) uses (b) and (c), and case (3) uses (b) and (c) explicitly, but also (a) through the use of case (1). The proof of Theorem 1.2 uses mainly (a) and (c), with some mild input from (b). The use of results about weights in the topology of singular varieties in (b), which play a crucial role in our approach, constitutes the main technical novelty of this paper. This will be clear from the detailed sketch of the proof of Theorem 1.1, which we now give.

A basic role is played by the minimal compactification $M_{\Gamma} \hookrightarrow M_{\Gamma}^*$, which is the cusp compactification of the (real or complex) hyperbolic manifold in cases (1) and (2) and the Satake–Baily–Borel compactification in cases (2) and (3). This gives the basic exact sequence

$$0 \longrightarrow \mathrm{H}^{k}_{!}(\mathscr{M}_{G}) \longrightarrow \mathrm{H}^{k}(\mathscr{M}_{G}) \longrightarrow \mathrm{H}^{k}(i^{*}j_{*}\mathbb{C}), \tag{1.1}$$

here the interior cohomology $\mathrm{H}^{k}_{!}(\mathscr{M}_{G})$ is, by definition, the image of $\mathrm{H}^{i}_{c}(\mathscr{M}_{G}) := \mathrm{colim}_{\Gamma}\mathrm{H}^{i}_{c}(M_{\Gamma})$ in $\mathrm{H}^{k}(\mathscr{M}_{G})$ and the third term is the boundary cohomology. This sequence is functorial for the inclusions $H \subset G$ considered in Theorems 1.1 and 1.2 because $M_{H,\Gamma_{H}} \to M_{\Gamma}$ extends to a morphism $M^{*}_{H,\Gamma_{H}} \to M^{*}_{\Gamma}$ of minimal compactifications. The obvious approach is to treat the interior cohomology and the contribution from the boundary separately, and this is what we do. The Lefschetz property for interior cohomology is the following:

Theorem 1.3 (Theorem 5.1, Corollary 3.2). The map Res is injective on $\mathrm{H}_{!}^{k}(\mathscr{M}_{G})$ for $k \leq d_{H}/2$ in cases (1) and (2) and for $k < d_{H}/2$ in case (3).

The proof of this is different in the various cases. In the real hyperbolic case (1), we adapt the Burger–Sarnak approach of [7, 12, 28] to the noncompact case. This is a more or less straightforward matter of combining the method with well-known results about residual Eisenstein cohomology, but since the literature on this is less than satisfactory, we treat it in some detail. In cases (2) and (3) when there is a complex structure available, we adopt a different approach based on some mixed Hodge theory. (The complex hyperbolic case was treated in [35, 36], but the approach here is slightly different and simpler.) Theorem 1.3 in cases (2) and (3) is then a corollary of the following:

Theorem 1.4 (Theorem 3.1). The map Res is injective on $\operatorname{Gr}_k^W \operatorname{H}^k(\mathscr{M}_G^*)$ for $k \leq d_H/2$.

This result is deduced as a corollary of a general nonvanishing criterion (Theorem 3.11) for the map on top weight quotients (for the weight filtration)

$$\operatorname{Res}:\operatorname{Gr}_{i}^{W}\operatorname{H}^{i}\left(\mathscr{M}_{G}^{*}\right)\to I_{H}^{G}\operatorname{Gr}_{i}^{W}\operatorname{H}^{i}\left(\mathscr{M}_{H}^{*}\right)$$

for a morphism between Shimura varieties, given in terms of the compact dual. This generalizes the criterion of [47] in the compact case and has other applications (see Remark 3.14). The spirit of the proof of Theorem 3.11 is that given a functorial cohomology group, some Poincaré duality, and semisimplicity, the averaging argument of [47] can be used to show that a linear combination of $G(\mathbb{Q})$ -translates of the cycle class of the subvariety gives the class of the compact dual of H. The necessary ingredients are available thanks to some results in mixed Hodge theory (consequences of the weights and purity package of [6, 42], reviewed in §3.1) and the theory of Chern classes of automorphic vector bundles (results from [25, 32], reviewed in §3.2 and Appendix C). We remark that the purely automorphic (i.e., Burger–Sarnak) method cannot be made to work easily for interior cohomology in case (3) (see Remark 5.4 for details).

Having treated the interior cohomology, we deal with the cohomology at infinity. In cases (1) and (2) this is straightforward: Given sequence (1.1) and the identification of the boundary cohomology in terms of Lie-algebra cohomology, it reduces to an elementary computation with Kostant's theorem (as was already done in case (2) in [36]). The argument in case (3) of orthogonal Shimura varieties is more delicate: The boundary of M_{Γ}^* is more complicated, containing modular curves as well as cusps, and it is no longer true that the restriction is injective on the entire boundary cohomology. Instead, the argument is in two steps. First, using an elementary argument using Kostant's theorem as in the rank 1 cases, one extends injectivity from the interior cohomology $H_1^i(\mathcal{M}_G)$ to an intermediate subspace

$$\mathrm{H}^{i}_{!}(\mathscr{M}_{G}) \subset \mathrm{Gr}^{W}_{i} \mathbb{H}^{i}_{c}\left(\mathscr{M}^{1}_{G}, j_{*}^{1} \mathbb{C}\right) \subset \mathrm{H}^{i}(\mathscr{M}_{G}),$$

which takes into account some contributions from the one-dimensional boundary strata (see §7.1 for the notation). Next, one extends injectivity to all of $\mathrm{H}^{i}(\mathcal{M}_{G})$ by taking into

account contributions from the cusps. By arguments in which weights, purity, and the description of the restriction to strata of the direct image sheaves in M_{Γ}^* play a crucial role, this reduces to the Lefschetz property for real hyperbolic manifolds for the pair $SO(1,m-1) \subset SO(1,n-1)$ appearing in the Levi subgroups of SO(2,m) and SO(2,n) – that is, to the result from case (1), although with nontrivial coefficients. This completes the sketch of the proof of Theorem 1.1.

1.4. Further remarks

Nonvanishing results for cup products in cohomology, which amount to injectivity of Res for the diagonal embedding $G \subset G \times G$, are known in the compact cases [12, 47] and for noncompact complex hyperbolic cases [36]. The nonvanishing of cup products in $\mathrm{H}^*_!(\mathscr{M}_G)$ follows from the criterion of Theorem 3.11. The extension to $\mathrm{H}^*(\mathscr{M}_G)$ should be possible using the methods used here.

Injectivity in degree $i = d_H/2$ in the complex hyperbolic case and the case of orthogonal Shimura varieties remain to be resolved. The two cases are slightly different, since in the first we have injectivity on $\mathrm{H}^{d_H/2}_{!}(\mathscr{M}_G)$ but not on the boundary cohomology, while in the second case we do not know injectivity on $\mathrm{H}^{d_H/2}_{!}(\mathscr{M}_G)$. In both cases, the classes potentially in the kernel of Res are constructed by residues of Eisenstein series, and their existence is caused by the nonvanishing of an *L*-value, whereas their survival under Res is also related to an *L*-value. We would also like to consider cases where $rank_{\mathbb{Q}}(H) < rank_{\mathbb{Q}}(G)$, to which our arguments do not necessarily apply. (Although note that what we have treated is the 'generic' case, at least over \mathbb{Q} – for example, in case (3), if $m+2 \geq 6$, then we are in the situation treated here.) It appears that these questions are most naturally considered in the framework of branching laws, and we will consider them in a sequel.

Finally, the appearance of the Lefschetz property in the real hyperbolic case as a local Lefschetz property at the cusp singularities for the orthogonal Shimura variety suggests trying to reverse the logic and *deduce* the Lefschetz property in the real hyperbolic case from purely geometric facts. It seems likely that this would follow from showing that a linear combination of Hecke translates of the image of $M_{H,\Gamma_H}^* \to M_{\Gamma}^*$ is ample – that is, resolving the question raised in [28]. Perhaps [15] can be used profitably here.

1.5. Contents

We end the introduction with a brief discussion of the contents of the individual sections.

Section 2 introduces the congruence quotients of interest and their minimal compactifications, recalls some well-known results on their local geometry and cohomology at infinity, and introduces the restriction maps in detail.

Section 3 discusses restriction between (connected) Shimura varieties. We show using some standard mixed Hodge theory that there is a simple criterion for the injectivity of Res on the top weight quotient of $H^*(\mathcal{M}_G^*)$. We apply it to $SU(1,m) \subset SU(1,n)$ and $SO(2,m) \subset SO(2,n)$ to prove injectivity on the top weight quotient and on interior cohomology in these cases. Section 4 contains Lie-algebra cohomology computations using Kostant's theorem which are necessary to treat boundary contributions in the various cases. These are explicit elementary calculations with roots and weights.

Section 5 considers the congruence real hyperbolic case and contains the proof of case (1) of Theorem 1.1.

Section 6 considers the congruence complex hyperbolic case and contains the proof of case (2) of Theorem 1.1 and the proof of Theorem 1.2.

Section 7 considers the case of orthogonal Shimura varieties. The results of \$\$3, 4, and 5 are combined to prove the remaining case (3) of Theorem 1.1.

The three appendices contain some facts which are presumably well known but for which we could not find appropriate references in the literature. Appendix A contains some facts about the L^2 cohomology of arithmetic manifolds used in §§5 and 6. In fact, we only need a very special case of what is proven (Proposition A.1 in the case SO(1,d)for d odd), but the facts recorded here will be useful elsewhere. Appendix B records some well-known facts about the construction of cohomology classes via residual Eisenstein series, for use in §§5 and 6. And Appendix C discusses Chern classes of automorphic vector bundles, which are used in §3.

2. Preliminaries

2.1. Congruence arithmetic quotients

The general setup we work in is as follows. Let G be a semisimple algebraic group over \mathbb{Q} , K the maximal compact subgroup of $G(\mathbb{R})$, and $X = G(\mathbb{R})/K$ the symmetric space. For a congruence subgroup $\Gamma \subset G(\mathbb{Q})$, the quotient

$$M_{\Gamma} = \Gamma \backslash X$$

is noncompact when G is Q-isotropic. The following three cases will be the main ones of interest to us:

- (i) $G(\mathbb{R})^{nc} = SO(1,d)$ for $d \ge 2$, so that X is a real hyperbolic d-space and M_{Γ} is a congruence hyperbolic manifold;
- (ii) $G(\mathbb{R})^{nc} = SU(1,n)$ for $n \ge 2$, so that X is the complex unit n-ball and M_{Γ} is a congruence ball quotient (or congruence complex hyperbolic manifold);
- (iii) $G(\mathbb{R})^{nc} = SO(2,n)$ for $n \ge 3$, so that

$$X = SO(2,n)/S(O(2) \times O(n)) = SO_0(2,n)/SO(2) \times SO(n)$$

and M_{Γ} is a Hermitian locally symmetric space which we will refer to, by an abuse of terminology, as an orthogonal Shimura variety.

In cases (ii) and (iii) the symmetric space X has an Hermitian structure, so that M_{Γ} is a smooth complex manifold if Γ is small enough. We will also be interested in the general case when X has an Hermitian structure; by an abuse of terminology, we will then refer to M_{Γ} as a Shimura variety.

Example 2.1. The standard examples of congruence quotients of types (i)–(iii) are given by quadratic or Hermitian spaces over \mathbb{Q} or number fields. For example, if (V,q) is a quadratic space over \mathbb{Q} and $q_{\mathbb{R}}$ has signature (1,d), then G = SO(q) gives an example of (i), whereas if the signature is (2,n), then it gives an example of (iii). If the number of variables is at least ≥ 5 and $q_{\mathbb{R}}$ is indefinite, then G = SO(q) is necessarily \mathbb{Q} -isotropic and M_{Γ} is noncompact. More generally, if n+2 is odd and ≥ 5 , then the only examples of type (iii) are the obvious ones – that is, they come from quadratic forms. When the number of variables is even, there are more complicated examples – for example, for d+1=8 there are triality forms of type (i).

Let $G(\mathbb{R})^c$ be the compact real form of $G(\mathbb{R})$. The compact symmetric space dual to X is

$$X^c = G(\mathbb{R})^c / K.$$

In our three cases, X^c is (i) the *d*-sphere, (ii) the complex projective *n*-space \mathbb{P}^n , and (iii) a quadric in \mathbb{P}^{n+1} . In all three cases there is a natural embedding $X \subset X^c$ and the action of $G(\mathbb{R})$ extends to the closure of X in X^c : In cases (ii) and (iii) it is the familiar $G(\mathbb{R})$ -equivariant Borel embedding of X in the flag variety, and in case (i) it is clear, for example, from the upper half-space model of hyperbolic space.

2.2. Minimal compactification

In all three of our cases, there is a canonical open immersion

$$j: M_{\Gamma} \hookrightarrow M_{\Gamma}^*$$

into a compact space, which we will call the minimal compactification. For cases (i) and (ii) it is the obvious cusp compactification and also coincides with the reductive Borel–Serre compactification. For cases (ii) and (iii) and more generally, for any arithmetic quotient of an Hermitian symmetric domain, it is the Satake–Baily–Borel compactification of M_{Γ} as a projective variety, and we will describe it in some more detail in this generality.

The closure of X in X^c decomposes as a disjoint union of boundary components, which are (by definition) the maximal connected complex submanifolds of the closure. The stabilizer of a proper (i.e., $\neq X$) boundary component is a product of maximal parabolic subgroups of the simple factors of $G(\mathbb{R})$, and the boundary component is called rational if the stabilizer is defined over \mathbb{Q} , in which case it is a maximal \mathbb{Q} -parabolic of G. As a topological space, $M_{\Gamma}^{c} = \Gamma \setminus X^{*}$, where

$$X^* = \bigsqcup_{\text{Frational}} F \quad \subset X^c \tag{2.1}$$

is the union of all rational boundary components of X, equipped with the Satake topology. The action of $G(\mathbb{Q})$ on X extends to a continuous action on X^* ; the stabilizer of a rational boundary component F is a maximal \mathbb{Q} -parabolic subgroup (in which case F is proper – i.e., $F \subset X^* - X$) or G itself (the case F = X). The Baily–Borel theory [5] puts an analytic structure on M^*_{Γ} inducing the given holomorphic structure on each stratum, and this structure is unique. Moreover, M^*_{Γ} has a unique structure of projective algebraic

1662

1663

variety compatible with this analytic structure, and this gives a canonical quasi-projective structure on M_{Γ} . The decomposition (2.1) induces an algebraic stratification of M_{Γ}^* .

Example 2.2. If G is isotropic and $G(\mathbb{R})^{nc}$ is isogenous to SU(1,n), the boundary components are points and M^*_{Γ} is the cusp compactification of the complex hyperbolic manifold.

Example 2.3. If $G(\mathbb{R})^{nc}$ is isogenous to SO(2,n), the boundary components have complex dimension 1 (i.e., they are upper half-planes) or 0 (points). The natural filtration of X^* induces a filtration by Zariski open subsets

$$M_{\Gamma} \subset M_{\Gamma}^1 \subset M_{\Gamma}^*$$

with $Z_{\Gamma}^1 = M_{\Gamma}^1 - M_{\Gamma}$ a disjoint union of curves and $Z_{\Gamma}^0 = M_{\Gamma}^* - M_{\Gamma}^1$ a finite set of cusps. (Either or both of Z_{Γ}^0 and Z_{Γ}^1 may be empty.)

The local geometry of the stratification of M_{Γ}^{*} is closely tied to the structure of parabolic subgroups, as we now review (see, e.g., [3, Sections III.4.1–III.4.2], [5, Section 3], [31, Section 6.1], or [25, Sections 7.1–7.3]). We will assume that the adjoint group G^{ad} is \mathbb{Q} -simple. Let P be a maximal rational parabolic subgroup. The unipotent radical W is an extension $1 \to U \to W \to V \to 1$, where U is the center of W and V is abelian. For the Lie algebras $\mathfrak{w} = \operatorname{Lie} W(\mathbb{R}), \mathfrak{u} = \operatorname{Lie} U(\mathbb{R}), \mathfrak{v} = \operatorname{Lie} V(\mathbb{R})$, we have an extension $0 \to \mathfrak{u} \to \mathfrak{v} \to \mathfrak{v} \to 0$. The action of A on $\mathfrak{u} = \text{Lie}U(\mathbb{R})$ is by the square of the positive (with respect to P) generator χ of $X^*(A)$, and the action on $\mathfrak{v} = \text{Lie}V(\mathbb{R})$ is by χ (if $\mathfrak{v} \neq 0$). The Levi quotient M = P/W has a decomposition $M = M_{\ell}M_{h}A$, where $A \cong \mathbb{G}_{m}$ is the maximal Q-split central torus in M, M_{ℓ} and M_h commute, (any lift of) M_h centralizes U, M_h contains no nontrivial connected Q-anisotropic subgroup, and $M_h(\mathbb{R})$ gives an Hermitian symmetric space, which is the rational boundary component corresponding to P. (The relation with the 'five-factor decomposition' of $[3, \S4.1]$ is the following: If P is the stabilizer of F and $P = G_h(F)G_\ell(F)M(F)V(F)U(F)$ as in [3, §4.1], then $W = W(F), U = U(F), V \cong V(F), M_{\ell} \cong G_{\ell}(F)M(F), \text{ and } M_hA \cong G_h(F).$ Note that if G is simply connected, the same is true of the derived group of the Levi, so $M^{der} = M_{\ell}^{der} \times M_h$, and hence M_h is also simply connected.

Example 2.4. Let G = SO(q) for a quadratic form over \mathbb{Q} of signature (2,n). Assume that G has \mathbb{Q} -rank 2 (this is automatic for $n \ge 6$). The maximal proper \mathbb{Q} -parabolics of G are the stabilizers of isotropic subspaces in V, which are of dimension 1 or 2. We have the following:

- 1. If P is the stabilizer of an isotropic plane $I \subset V$, the unipotent radical is a nontrivial extension $1 \to \mathbb{G}_a \to W \to \mathbb{G}_a^{2(n-2)} \to 1$ and the Levi M is $GL(2) \times SO(I^{\perp}/I)$. Here $M_h A = GL(2)$ and $M_\ell = SO(I^{\perp}/I) \cong SO(n-2)$ is anisotropic over \mathbb{R} .
- 2. If P is the stabilizer of an isotropic line $I \subset V$, the unipotent radical is abelian $W \cong \mathbb{G}_a^{n-2}$ and the Levi is $M \cong \mathbb{G}_m \times SO(I^{\perp}/I)$. Here M_h is trivial, $M_{\ell} \cong SO(I^{\perp}/I)$ has \mathbb{Q} -rank 1 and $M_{\ell}(\mathbb{R}) = SO(1, n-1)$.

The corresponding strata of M_{Γ}^* are modular curves in case (1) and cusps in case (2).

Example 2.5. It can happen that $Z_{\Gamma}^0 = \emptyset$ –for example, for the Q-rank 1 inner form of $\operatorname{Sp}(4,\mathbb{R}) = \operatorname{Spin}(2,3)$ associated with an indefinite quaternion algebra D over Q and a rank 2 Hermitian space over D with respect to the involution of D extending the nontrivial Galois action on the maximal subfield of D (which is real quadratic). (This is the form denoted $C_{2,1}^{(2)}$ in [46, p. 57].) In this case the stabilizer of a cusp is the inner form D^{\times} of GL(2) and the boundary of M_{Γ}^{*} is a disjoint union of Shimura curves.

Example 2.6. An example with $Z_{\Gamma}^1 = \emptyset$ is that of Hilbert modular surfaces, which are forms of $Spin(2,2) \cong SL(2) \times SL(2)$ with Q-rank 1. The boundary of M_{Γ}^* consists of cusps.

2.3. Direct limits

1664

Let G be a semisimple Q-algebraic group, $X = G(\mathbb{R})/K$ the symmetric space, and $M_{\Gamma} = \Gamma \setminus X$ for congruence Γ . For $\Gamma' \subset \Gamma$ the covering map $M_{\Gamma'} \to M_{\Gamma}$ gives pullback maps in cohomology and compactly supported cohomology, so taking colimits over all congruence subgroups, we define

$$\begin{aligned}
\mathbf{H}^{i}(\mathcal{M}_{G}) &:= \operatorname{colim}_{\Gamma} \mathbf{H}^{i}(M_{\Gamma}), \\
\mathbf{H}^{i}_{c}(\mathcal{M}_{G}) &:= \operatorname{colim}_{\Gamma} \mathbf{H}^{i}_{c}(M_{\Gamma}), \\
\mathbf{H}^{i}_{!}(\mathcal{M}_{G}) &:= \operatorname{colim}_{\Gamma} \mathbf{H}^{i}_{!}(M_{\Gamma}),
\end{aligned} \tag{2.2}$$

where, as usual, $\mathrm{H}_{i}^{i}(M_{\Gamma}) = \mathrm{im}\left(\mathrm{H}_{c}^{i}(M_{\Gamma}) \to \mathrm{H}^{i}(M_{\Gamma})\right)$ is the interior cohomology. All these are smooth $G(\mathbb{Q})$ -modules, in the sense that the stabilizer of a vector is a congruence subgroup. The action of $g \in G(\mathbb{Q})$ on $\mathrm{H}^{i}(M_{\Gamma}) \subset \mathrm{H}^{i}(\mathcal{M}_{G})$ is given by the pullback $\mathrm{H}^{i}(M_{\Gamma}) \to \mathrm{H}^{i}\left(M_{g\Gamma g^{-1}}\right)$ by the isomorphism $M_{g\Gamma g^{-1}} \to M_{\Gamma}$ induced by left translation by g^{-1} on the universal cover X. The transition maps in the colimits are injective, and $\mathrm{H}^{*}(M_{\Gamma})$ can be recovered as the Γ -invariants in $\mathrm{H}^{*}(\mathcal{M}_{G})$. The same remarks apply to $\mathrm{H}_{c}^{*}(\mathcal{M}_{G})$ and $\mathrm{H}_{!}^{*}(\mathcal{M}_{G})$.

When the symmetric space X is Hermitian or $G(\mathbb{R})^{nc}$ is isogenous to SO(1,d), we also have the minimal compactification M^*_{Γ} as in §2.2, and we can define

$$\mathrm{H}^{i}(\mathscr{M}_{G}^{*}) := \mathrm{colim}_{\Gamma}\mathrm{H}^{i}(M_{\Gamma}^{*}).$$

$$(2.3)$$

This is a smooth $G(\mathbb{Q})$ -module, and in the Hermitian case it carries a mixed (ind-)Hodge structure. In particular, it has a weight filtration with weights $\leq i$ in degree *i*, and the graded pieces are

$$\operatorname{Gr}_{i}^{W}\operatorname{H}^{i}(\mathscr{M}_{G}^{*}) = \operatorname{colim}_{\Gamma}\operatorname{Gr}_{i}^{W}\operatorname{H}^{i}(M_{\Gamma}^{*})$$

by strictness of the weight filtration.

The inductive setup requires the use of nontrivial coefficients (at the boundary) to treat the case of trivial coefficients. A finite-dimensional algebraic representation E of $G(\mathbb{C})$ gives a local system on M_{Γ} which, for simplicity, we continue to denote E, and we can consider $\mathrm{H}^*(M_{\Gamma}, E)$, the colimit

$$\mathrm{H}^*(\mathscr{M}_G, E) = \operatorname{colim}_{\Gamma} \mathrm{H}^*(M_{\Gamma}, E),$$

and similarly $\mathrm{H}^*_c(\mathscr{M}_G, E)$ and $\mathrm{H}^*_!(\mathscr{M}_G, E)$. For minimal compactifications $j_{\Gamma} : M_{\Gamma} \hookrightarrow M^*_{\Gamma}$, we take the sheaf $\mathrm{H}^0(j_{\Gamma*}E)$ – this is the ordinary, underived push-forward – and define

$$\mathrm{H}^*\left(\mathscr{M}_G^*, E\right) := \operatorname{colim}_{\Gamma} \mathrm{H}^*\left(M_{\Gamma}^*, \mathrm{H}^0(j_{\Gamma*}E)\right).$$

2.4. Restriction maps

Now suppose that $H \subset G$ is an injective homomorphism of semisimple Q-groups. Choosing (as we may) a maximal compact K in $G(\mathbb{R})$ such that $K_H = K \cap G(\mathbb{R})$ is maximal compact, and letting $\Gamma_H = \Gamma \cap H(\mathbb{Q})$ and $M_{H,\Gamma_H} = \Gamma_H \setminus H(\mathbb{R})/K_H$, we get a map $\iota: M_{H,\Gamma_H} \longrightarrow M_{\Gamma}$ which is well known to be proper. Thus there are induced pullback maps $\mathrm{H}^i(M_{\Gamma}) \to$ $\mathrm{H}^i(M_{H,\Gamma_H})$ and $\mathrm{H}^i_c(M_{\Gamma}) \to \mathrm{H}^i_c(M_{H,\Gamma_H})$ (the latter because $M_{H,\Gamma_H} \to M_{\Gamma}$ is proper). These are compatible under the natural maps $\mathrm{H}^*_c(\cdot) \to \mathrm{H}^*(\cdot)$, forgetting supports, and hence induce $\mathrm{H}^i_!(M_{\Gamma}) \to \mathrm{H}^i_!(M_{H,\Gamma_H})$. In the limit over Γ we have $H(\mathbb{Q})$ -equivariant maps

$$\iota^* : \mathrm{H}^i_{(c)}(\mathscr{M}_G) \to \mathrm{H}^i_{(c)}(\mathscr{M}_H)$$

in cohomology and compactly supported cohomology. There are induced homomorphisms of smooth $G(\mathbb{Q})$ -modules

$$\operatorname{Res}: \operatorname{H}^{i}_{(c)}(\mathscr{M}_{G}) \longrightarrow I^{G}_{H} \operatorname{H}^{i}_{(c)}(\mathscr{M}_{H}),$$

where I_H^G is an induction functor such that for a smooth $H(\mathbb{Q})$ -module $U, I_H^G U$ consists of functions $f: G(\mathbb{Q}) \to U$ such that $f(gh) = h^{-1} \cdot f(g)$ and f is left-invariant by a congruence subgroup of $G(\mathbb{Q})$, and the action of $g \in G(\mathbb{Q})$ is by $(g \cdot f)(x) = f(g^{-1}x)$. Then I_H^G is exact, takes smooth modules to smooth modules, and is right adjoint to restriction (these facts are completely elementary, see [35, Section 3.1]). Explicitly, Res is given by

$$\operatorname{Res}(\alpha)(g) = \iota^* \left(g^{-1} \cdot \alpha \right) = \iota^*((g \cdot)^* \alpha).$$

Note that Res restricts to a map Res : $\mathrm{H}^{i}_{!}(\mathcal{M}_{G}) \to I^{G}_{H}\mathrm{H}^{i}_{!}(\mathcal{M}_{H})$ on interior cohomology.

When both H and G give Hermitian symmetric spaces and K is chosen (as it may be) so that the map $H(\mathbb{R})/K_H \to G(\mathbb{R})/K$ is holomorphic, the map $M_{H,\Gamma_H} \to M_{\Gamma}$ extends to a morphism $M^*_{H,\Gamma_H} \to M^*_{\Gamma}$ of varieties of minimal compactifications. This well-known general fact (compare [44] or [27, Section 3.3]) is easily seen in our primary cases of interest using the description of M^*_{Γ} given in §2.2. Pullback induces an $H(\mathbb{Q})$ -equivariant map $\mathrm{H}^i(\mathcal{M}^*_G) \to \mathrm{H}^i(\mathcal{M}^*_H)$, which gives a homomorphism of mixed Hodge structures

$$\operatorname{Res}: \operatorname{H}^{i}(\mathscr{M}_{G}^{*}) \longrightarrow I_{H}^{G}\operatorname{H}^{i}(\mathscr{M}_{H}^{*})$$

by adjunction. There is a similar mapping in the real hyperbolic cases where $H(\mathbb{R})^{nc} \subset G(\mathbb{R})^{nc}$ is $SO(1,c) \subset SO(1,d)$ (up to isogeny) and in the 'mixed' case $SO(1,n) \subset SU(1,n)$, coming from the obvious extension of $M_{H,\Gamma_H} \to M_{\Gamma}$ to minimal compactifications.

Now assume that $H \subset G$ is such that the restriction of finite-dimensional representations from G to H is multiplicity-free. The situations we will treat are well known to be of this type, by classical branching laws (e.g. [22, Section 8.1.1]). Choose Borel subgroups $B_H \subset B$ and maximal tori $T_H \subset B_H$ and $T \subset B$ of $H(\mathbb{C})$ and $G(\mathbb{C})$, and for E with Bhighest weight $\lambda \in X^*(T)$ let E_H be the unique summand of $E|_{H(\mathbb{C})}$ with highest weight

 $\lambda|_{T_H}$. The composition $\mathrm{H}^*(\mathscr{M}_G, E) \to \mathrm{H}^*(\mathscr{M}_H, E|_H) \to \mathrm{H}^*(\mathscr{M}_H, E_H)$ is $H(\mathbb{Q})$ -invariant and induces a restriction map

$$\operatorname{Res}: \operatorname{H}^*(\mathscr{M}_G, E) \longrightarrow \operatorname{H}^*(\mathscr{M}_H, E_H)$$

with coefficients. There are similar maps for $H^*_c(\mathcal{M}_G, E)$, $H^*(\mathcal{M}_G, E)$, and $H^*(\mathcal{M}^*_G, E)$.

The use of Res for several different maps should cause no confusion, as we will always specify the domain when discussing injectivity results. We will also frequently write $\mathrm{H}^{i}(\mathcal{M}),\mathrm{H}^{i}(\mathcal{M}^{*})$, and so on, for $\mathrm{H}^{i}(\mathcal{M}_{G}),\mathrm{H}^{i}(\mathcal{M}_{G}^{*})$, and so on – dropping the subscript G when it is clear from context.

2.5. Higher direct images in the minimal compactification

We will use a well-known description of the restriction of $j_*\mathbb{Q}_{M_{\Gamma}}$ to a stratum of M^*_{Γ} in the case of Shimura varieties. (Here by j_* we mean the push-forward on the level of derived categories – that is, Rj_* in old-fashioned notation.)

To fix notation, let $i_S : S \hookrightarrow M_{\Gamma}^*$ be a stratum of the minimal compactification. Choose a rational boundary component $F \twoheadrightarrow S$ and let P = MW be the stabilizer of F and $M = M_h M_\ell A$ as in §2.2. For the congruence subgroup Γ , let $\Gamma_W = \Gamma \cap W(\mathbb{Q}), \ \Gamma_P = \Gamma \cap P(\mathbb{Q}), \ \Gamma_M = \Gamma_P / \Gamma_W, \ \Gamma_{M_\ell} = \Gamma_M \cap M_\ell(\mathbb{Q}), \ \text{and} \ \Gamma_{M_h} = \Gamma_M / \Gamma_{M_\ell}$. These are all neat arithmetic subgroups when Γ is neat.

Proposition 2.7. For a stratum $i_S : S \hookrightarrow M^*_{\Gamma}$ we have the following:

(1) There is a natural isomorphism in the derived category

$$i_S^* j_* \mathbb{Q}_{M_{\Gamma}} = \bigoplus_k \mathrm{H}^k \left(i_S^* j_* \mathbb{Q}_{M_{\Gamma}} \right) [-k]$$
(2.4)

of sheaves on $S = \Gamma_{M_h} \backslash F$. The object $\mathrm{H}^k(i_S^* j_* \mathbb{Q}_{M_{\Gamma}})$ is the local system on S associated with the representation of M_h on

$$\mathrm{H}^{k}\left(i_{S}^{*}j_{*}\mathbb{Q}_{M_{\Gamma}}\right)_{s} \cong \bigoplus_{r+s=k} \mathrm{H}^{r}\left(\Gamma_{M_{\ell}}, \mathrm{H}^{s}(\mathfrak{w}, \mathbb{Q})\right)$$
(2.5)

for $s \in S$.

(2) The weight filtration on $\mathrm{H}^{k}(i_{S}^{*}j_{*}\mathbb{Q}_{M_{\Gamma}})_{s}$ is split by the action of A on $\mathrm{H}^{*}(\mathfrak{w},\mathbb{Q})$ – that is, $\mathrm{Gr}_{i}^{W}\mathrm{H}^{k}(i_{S}^{*}j_{*}\mathbb{Q}_{M_{\Gamma}})_{s}$ is identified with the subspace on which A acts by χ^{-i} (where $\chi \in X^{*}(A)$ is such that A acts on the center \mathfrak{u} of \mathfrak{w} by χ^{2} and on $\mathfrak{v} = \mathfrak{w}/\mathfrak{u}$ by χ ; compare §2.2).

The description (2.5) of cohomology sheaves can be found, for example, in [31, Proposition 5.6] or [23, Corollary 22.8]. The weight filtration on $\mathrm{H}^{k}(i_{S}^{*}j_{*}\mathbb{Q}_{M_{\Gamma}})_{s}$ comes from the theory of mixed Hodge modules, and the assertion in (2) is due to Looijenga and Rapoport [31, Proposition 5.6]. (The analogue in the *l*-adic setting is in [39].) The existence of the decomposition (2.4) in the derived category can in fact be deduced from this, but instead one can use [19, Theorem 2.9], which proves the direct sum decomposition (2.4) in the derived category of mixed Hodge modules and the identification of the graded in the context of Shimura varieties. The isomorphisms (2.4) and (2.5) are equivariant for the actions of $(M_{\ell}AW)(\mathbb{Q})$ on both sides (which factor through $(M_{\ell}A)(\mathbb{Q})$).

Remark 2.8. The real hyperbolic case SO(1,d) fits notationally into this setup by taking $M_h = \{e\}, M_\ell = SO(d-1)$, and then (1) remains true.

3. Restriction between Shimura varieties

In this section we assume that H and G both give rise to Shimura varieties, and consider the restriction Res : $\operatorname{Gr}_i^W \operatorname{H}^i(\mathscr{M}_G^*) \to I_H^G \operatorname{Gr}_i^W \operatorname{H}^i(\mathscr{M}_H^*)$ on the top weight quotient for the weight filtration. We prove a criterion (Theorem 3.11) for the nonvanishing of this restriction involving the compact dual symmetric space, which is the analogue of the criterion of [47] in this situation. It implies the following:

Theorem 3.1. If $H(\mathbb{R})^{nc} \subset G(\mathbb{R})^{nc}$ is $SU(1,m) \subset SU(1,n)$ or $SO(2,m) \subset SO(2,n)$, then Res is injective on $\operatorname{Gr}_i^W \operatorname{H}^i(\mathscr{M}_G^*)$ in degrees $\leq m$.

The unitary case of this is contained in [35, Theorem 3.17], although the proof here is slightly different (and more direct). As a corollary, we get the following injectivity statements for interior cohomology, the first of which was proved earlier in [35] (see also [13]):

Corollary 3.2. If $H(\mathbb{R})^{nc} \subset G(\mathbb{R})^{nc}$ is $SU(1,m) \subset SU(1,n)$, then Res is injective on $\mathrm{H}^{i}_{\mathrm{f}}(\mathscr{M}_{G})$ in degrees $\leq m$.

If $H(\mathbb{R})^{nc} \subset G(\mathbb{R})^{nc}$ is $SO(2,m) \subset SO(2,n)$, then Res is injective on $\mathrm{H}^{i}_{!}(\mathscr{M}_{G})$ in degrees $\leq m$ if $rk_{\mathbb{Q}}(H) \leq 1$ and in degrees $\leq m-1$ if $rk_{\mathbb{Q}}(H) = 2$.

3.1. Some cohomological facts

We will use some facts about the cohomology of (possibly) singular varieties, summarized in Proposition 3.3 and Lemma 3.5.

Recall that by [20] the rational cohomology $\mathrm{H}^*(X) = \mathrm{H}^*(X, \mathbb{Q})$ and homology $\mathrm{H}_*(X) = \mathrm{H}_*(X, \mathbb{Q})$ of a complex algebraic variety X carry rational mixed Hodge structures; in particular, they have weight filtrations. The theory of mixed Hodge modules ([42], especially §4) gives a relative version of mixed Hodge structures and allows for sheaf-theoretic arguments, mirroring the situation in *l*-adic cohomology over finite fields [6]. Let X be an irreducible complex variety of dimension d. Let \mathbb{Q}_X^H be the canonical lift of \mathbb{Q}_X to an object in the derived category of mixed Hodge modules on X – that is, $\mathbb{Q}_X^H = a_X^* \mathbb{Q}^H$, where $a_X : X \to Spec(\mathbb{C})$ and \mathbb{Q}^H is the trivial Hodge structure. The rational cohomology, compactly supported cohomology, homology, and Borel–Moore homology groups of X acquire mixed Hodge structures via

$$\mathrm{H}^{i}(X) = \mathbb{H}^{i}\left(X, \mathbb{Q}_{X}^{H}\right), \qquad \mathrm{H}^{i}_{c}(X) = \mathbb{H}^{i}_{c}\left(X, \mathbb{Q}_{X}^{H}\right), \qquad (3.1)$$

$$\mathbf{H}_{i}(X) = \mathbb{H}_{c}^{-i}\left(X, \mathbb{D}\mathbb{Q}_{X}^{H}\right), \qquad \qquad \mathbf{H}_{i}^{BM}(X) = \mathbb{H}^{-i}\left(X, \mathbb{D}\mathbb{Q}_{X}^{H}\right), \qquad (3.2)$$

where \mathbb{D} is the Verdier duality functor, normalized so that $\mathbb{DQ}_X^H = \mathbb{Q}_X^H[2d](d)$ if X is smooth. The weights are determined by the fact that \mathbb{Q}_X^H has weights ≤ 0 , so, for example,

 $\mathrm{H}_{c}^{i}(X)$ has weights $\leq i$ and $\mathrm{H}_{i}^{BM}(X)$ has weights $\geq -i$. When X is proper, which is the main case of interest to us, we have $\mathrm{H}^{i}(X) = \mathrm{H}_{c}^{i}(X)$ and $\mathrm{H}_{i}(X) = \mathrm{H}_{i}^{BM}(X)$, so that $\mathrm{H}^{i}(X)$ has weights $\leq i$ and $\mathrm{H}_{i}(X)$ has weights $\geq -i$.

We will also use the intersection complex $IC_X^H = (j_{!*}\mathbb{Q}_U^H[d])[-d]$, where $j: U \hookrightarrow X$ is the inclusion of an open dense smooth subset; this lifts the topological intersection complex $IC_X = (j_{!*}\mathbb{Q}_U[d])[-d]$ of X, and it is pure of weight 0. (Our notation is slightly different from [42], where $IC_X(\mathbb{Q}^H)$ is used for $j_{!*}\mathbb{Q}_U[d]$.) Replacing \mathbb{Q}_X^H by IC_X^H in equation (3.2) defines rational mixed Hodge structures

$$\operatorname{IH}^{i}(X), \quad \operatorname{IH}^{i}_{c}(X), \quad \operatorname{IH}_{i}(X), \quad \operatorname{IH}^{BM}_{i}(X)$$

on the intersection cohomology, intersection cohomology with compact support, intersection homology, and Borel–Moore intersection homology, respectively. When X is proper, these are all pure, $\operatorname{IH}^*(X) = \operatorname{IH}^*_c(X)$ and $\operatorname{IH}_*(X) = \operatorname{IH}^{BM}_*(X)$, and the isomorphism $\mathbb{D}IC_X^H = IC_X^H[2d](d)$ extending $\mathbb{D}\mathbb{Q}_U^H = \mathbb{Q}_U^H[2d](d)$ on any smooth open subset $U \subset X$ induces duality isomorphisms $\operatorname{IH}^i(X) \cong \operatorname{IH}_{2d-i}(X)(-d)$ for all *i*.

Proposition 3.3. If $f: X \to Y$ is a morphism of varieties, there are maps

$$f^*: \operatorname{Gr}_i^W \operatorname{H}^i(Y) \to \operatorname{Gr}_i^W \operatorname{H}^i(X),$$

$$f_*: W_{-i} \operatorname{H}_i(X) \to W_{-i} \operatorname{H}_i(Y)$$
(3.3)

for each i satisfying

$$f_*(f^*(\alpha) \cap \beta) = \alpha \cap f_*(\beta) \qquad \text{for } \alpha \in \mathrm{Gr}_i^W \mathrm{H}^i(Y), \beta \in W_{-j} \mathrm{H}_j(X).$$
(3.4)

If X is an irreducible proper variety of dimension d, then the following are true:

(1) $H^i(X)$ has weights $\leq i$, $H_i(X)$ has weights $\geq -i$, and the extreme weights are given by

$$Gr_i^W H^i(X) = im \left(H^i(X) \to IH^i(X) \right),$$

$$W_{-i} H_i(X) = im \left(IH_i(X) \to H_i(X) \right)$$
(3.5)

for all i, j.

(2) If $[X] \in H_{2d}(X)(-d)$ is the fundamental class of X, then

$$\cap [X] : \operatorname{Gr}_{i}^{W} \operatorname{H}^{i}(X) \longrightarrow W_{-(2d-i)} \operatorname{H}_{2d-i}(X)(-d)$$
(3.6)

is an isomorphism for all i.

(3) If $i: Z \hookrightarrow X$ is an irreducible closed subvariety of codimension c, then the cycle class

$$\xi_{X,Z} := (\cap[X])^{-1}(i_*[Z]) \in \mathrm{Gr}_{2c}^W \mathrm{H}^{2c}(X)(c)$$

has the property that if $\alpha \in \mathrm{H}^{2\dim Z}(X)$ with $i^*(\alpha) = \xi_{Z,pt}$, then $\alpha \cdot \xi_{X,Z} = \xi_{X,pt}$.

Proof. The statements about f^* and f_* are simply the functoriality of the weight filtration and the fact that when homology is considered as a module over the cohomology ring using the cap product, push-forward in homology is a module over pullback in

cohomology. Statements (1) and (2) are contained in [42, §4.5], but for the reader's convenience we outline the arguments.

For the standard cohomology functor H^0 on mixed Hodge modules (which corresponds to the perverse cohomology functor ${}^{p}\mathrm{H}^0$ on sheaves), we have dual natural isomorphisms

$$\operatorname{Gr}_{d}^{W}\operatorname{H}^{d}\left(\mathbb{Q}_{X}^{H}\right) = IC_{X}^{H}[d],$$

$$W_{-d}\operatorname{H}^{-d}\left(\mathbb{D}\mathbb{Q}_{X}^{H}\right) = IC_{X}^{H}d$$
(3.7)

(see [42, §4.5] for details). If X is proper, then these dual statements and the hypercohomology spectral sequence imply that

$$Gr_i^W H^i(X) = im \left(H^i(X) \to IH^i(X) \right),$$

$$W_{-j} H_j(X) = im \left(IH_j(X) \to H_j(X) \right),$$
(3.8)

as claimed in (1).

An irreducible variety X has a fundamental class in Borel-Moore homology,

$$[X] \in \mathrm{H}_{2d}^{BM}(X)(-d) = \mathbb{H}^0\left(X, \left(\mathbb{D}\mathbb{Q}_X^H\right)[-2d](-d)\right) = \mathrm{Hom}\left(\mathbb{Q}_X^H, \left(\mathbb{D}\mathbb{Q}_X^H\right)[-2d](-d)\right),$$

giving the fundamental class homomorphism $\mathbb{Q}_X^H \to (\mathbb{D}\mathbb{Q}_X^H)[-2d](-d)$, which is an isomorphism if X is smooth. By the identities (3.7) and standard facts about the t-structure and weights, it factors as

$$\mathbb{Q}_X^H \longrightarrow IC_X^H \longrightarrow \left(\mathbb{D}IC_X^H\right)[-2d](-d) \longrightarrow \left(\mathbb{D}\mathbb{Q}_X^H\right)[-2d](-d), \tag{3.9}$$

where the first arrow is the unique extension of the identity morphism $\mathbb{Q}_U \to \mathbb{Q}_U$ on U and the third arrow is its dual (up to a twist). The second is the Verdier duality isomorphism extending $\mathbb{Q}_U^H = (\mathbb{D}\mathbb{Q}_U^H) [-2d](-d)$, and induces Poincaré duality isomorphisms

$$\operatorname{IH}_{c}^{i}(X) = \operatorname{\mathbb{H}}_{c}^{i}\left(X, IC_{X}^{H}\right) \cong \operatorname{\mathbb{H}}_{c}^{i-2d}\left(X, \mathbb{D}IC_{X}^{H}\right)(-d) = \operatorname{IH}_{2d-i}(X)(-d) = \operatorname{IH}^{2d-i}(X)^{*}(-d)$$

and

$$\operatorname{IH}^{i}(X) = \mathbb{H}^{i}\left(X, IC_{X}^{H}\right) \cong \mathbb{H}^{i-2d}\left(X, \mathbb{D}IC_{X}^{H}\right)(-d) = \operatorname{IH}_{2d-i}^{BM}(X)(-d),$$

and hence a nondegenerate pairing $\operatorname{IH}^{i}(X) \times \operatorname{IH}_{c}^{2d-i}(X) \to \mathbb{Q}(-d)$. The fundamental class homomorphism induces the identity isomorphism $\operatorname{Gr}_{d}^{W}\operatorname{H}^{d}(\mathbb{Q}_{X}^{H}) = W_{-d}\operatorname{H}^{-d}(\mathbb{D}\mathbb{Q}_{X}^{H})(-d)$ from equation (3.7) (as indeed it must, since it is the unique extension of $\mathbb{Q}_{U}^{H}[d] = \mathbb{D}\mathbb{Q}_{U}^{H}d$ on any smooth open $U \subset X$).

Now assume that X is proper. Then the duality of the first and third arrows in expression (3.9) implies that the cap product with the fundamental class [X] induces an isomorphism

$$\cap [X] : \operatorname{Gr}_{i}^{W} \operatorname{H}^{i}(X) \xrightarrow{\simeq} W_{2d-i} \operatorname{H}_{2d-i}(X)(-d).$$
(3.10)

This proves (2). For (3), note that if $i^*(\alpha) = \xi_{Z,pt}$, then $[Z] \cap i^*(\alpha) = 1$, so that

$$1 = i_*(i^*(\alpha) \cap [Z])$$

= $\alpha \cap i_*[Z]$
= $[X] \cap (\alpha \cdot (\cap [X])^{-1}(i_*[Z]))$
= $[X] \cap \alpha \cdot \xi_{X,Z},$ (3.11)

and thus $\alpha \cdot \xi_{X,Z} = (\cap [X])^{-1}(1) = \xi_{X,pt}$.

Remark 3.4. The assertion in (2) is related to some facts in classical mixed Hodge theory which we will also use. If X is irreducible and proper and $Y \to X$ is a resolution of singularities, then

$$\operatorname{Gr}_{i}^{W}\operatorname{H}^{i}(X) = \operatorname{im}\left(\operatorname{H}^{i}(X) \to \operatorname{H}^{i}(Y)\right)$$

for all *i* by [20, Prop. 8.2.5]. This is equivalent to equation (3.5), because the pullback factors as $\mathrm{H}^{i}(X) \to \mathrm{IH}^{i}(X) \hookrightarrow \mathrm{H}^{i}(Y)$ for any inclusion $\mathrm{IH}^{*}(X) \subset \mathrm{H}^{*}(Y)$ coming from the decomposition theorem [6, 42]. Since one also has

$$\operatorname{Gr}_{i}^{W}\operatorname{H}_{c}^{i}(U) = \operatorname{im}\left(\operatorname{H}_{c}^{i}(U) \to \operatorname{IH}^{i}(X)\right) = \operatorname{im}\left(\operatorname{H}_{c}^{i}(U) \to \operatorname{H}^{i}(Y)\right),$$

where U is smooth dense open over which $Y \to X$ is an isomorphism, one sees that $\operatorname{Gr}_{i}^{W}\operatorname{H}_{c}^{i}(U) \subset \operatorname{Gr}_{i}^{W}\operatorname{H}^{i}(X)$ for all *i*.

The following purity lemma will be used later in §7:

Lemma 3.5. Let X be a normal complex variety of dimension d with $U \subset X^1 \subset X$ a filtration by open subsets such that U is smooth and open dense in X^1 , $Z^1 = X^1 - U$ is smooth of dimension 1, and $Z^0 = X - X^1$ is smooth of dimension 0. Let $j: U \hookrightarrow X$ and $i: Z^0 \hookrightarrow X$ be the inclusions. Then $H^i(i^{0*}j_*\mathbb{Q}_U^H)$ has weights $\leq i$ for $i \leq d-2$. If $Z^1 = \emptyset$, then $H^i(i^{0*}j_*\mathbb{Q}_U^H)$ has weights $\leq i$ for $i \leq d-1$.

Proof. Write $j = j^0 \circ j^1$ for $j^1 : U \hookrightarrow X^1$ and $j^0 : X^1 \hookrightarrow X$. Since $IC_X = \tau_{\leq d} j_*^0 \tau_{\leq d-1} j_*^1 \mathbb{C}_U$, it follows easily that $IC_X \to j_* \mathbb{C}_U$ induces an isomorphism on cohomology sheaves in degrees $\leq d-2$. The same then holds for $IC_X^H \to j_* \mathbb{Q}_U^H$. On the other hand, by pointwise purity of the intersection complex, $H^i (IC_X^H)_x$ has weights $\leq i$ in all degrees [6, 42]. This proves the first assertion of the lemma. If $Z^1 = \emptyset$, we have that $IC_X \to j_* \mathbb{C}_U$ induces isomorphisms on cohomology sheaves in degrees $\leq d-1$, and purity proves the second assertion.

3.2. Invariants and the compact dual

We will assume from now on that G is semisimple and $X = G(\mathbb{R})/K$ is an Hermitian symmetric domain. In addition, we assume in this subsection that G is simply connected. We return to the use of cohomology with complex coefficients and ignore Tate twists.

We will consider the $G(\mathbb{Q})$ -module $\operatorname{IH}^{i}(\mathscr{M}^{*}) := \operatorname{colim}_{\Gamma}\operatorname{IH}^{i}(M_{\Gamma}^{*})$, which is smooth and admissible. Note that equation (3.5) gives an inclusion

$$\mathrm{Gr}^W_*\mathrm{H}^*(\mathscr{M}^*) := \bigoplus\nolimits_i \mathrm{Gr}^W_i\mathrm{H}^i(\mathscr{M}^*) \subset \mathrm{IH}^*(\mathscr{M}^*)$$

of $G(\mathbb{Q})$ -modules.

1670

Proposition 3.6. The $G(\mathbb{Q})$ -modules $\operatorname{Gr}^W_* \operatorname{H}^*(\mathscr{M}^*)$ and $\operatorname{IH}^*(\mathscr{M}^*)$ are semisimple, and the summand of invariants is given by

$$\mathrm{Gr}^W_*\mathrm{H}^*(\mathscr{M}^*)^{G(\mathbb{Q})} = \mathrm{IH}^*(\mathscr{M}^*)^{G(\mathbb{Q})} = \mathrm{H}^*(X^c).$$

The embedding of $\mathrm{H}^*(X^c)$ in $\mathrm{Gr}^W_*\mathrm{H}^*(\mathscr{M}^*)$ is functorial: If $H \subset G$ gives a complex subdomain $X_H \subset X$, then the obvious diagram coming from $X^c_H \subset X^c$ and $M^*_{H,\Gamma_H} \to M^*_{\Gamma}$ commutes.

Proof. There is a natural isomorphism

$$\operatorname{IH}^{*}(\mathscr{M}^{*}) = \operatorname{H}^{*}\left(\mathfrak{g}, K, L^{2}_{dis}\left(G(\mathbb{Q}) \setminus G(\mathbb{A})\right)\right)$$
(3.12)

thanks to [30, 43] and [16]. Here L^2_{dis} is the L^2 discrete spectrum; see [35, Proposition 3.8] for a detailed discussion of equation (3.12). This proves the semisimplicity statements. The inclusion of the constants in L^2 functions induces an embedding of $H^*(X^c) = H^*(\mathfrak{g}, K, \mathbb{C})$ in $IH^*(\mathscr{M}^*)$. It follows from equation (3.12) using strong approximation and the density of $G(\mathbb{Q})$ in $G(\mathbb{R})$ (weak approximation) – see, for example, the proof [35, Proposition 3.8], which works verbatim here – that these are all the invariants.

To show that the invariants are actually in $\operatorname{Gr}_*^W \operatorname{H}^*(\mathscr{M}^*)$, we will use Chern classes of automorphic vector bundles [25, 32]. A finite-dimensional representation V of K gives a homogenous bundle \mathscr{V}^c on $X^c = G(\mathbb{R})^c/K$. Restricting by the Borel embedding $X \subset X^c$ (see §2.1) and dividing by Γ gives a bundle \mathscr{V}_{Γ} on $M_{\Gamma} = \Gamma \setminus X$ for any Γ . The bundle \mathscr{V}_{Γ} does not, in general, extend to a vector bundle on M_{Γ}^* (although see Example 3.7 for an important exception), but Goresky and Pardon [25] defined classes $c_k^*(\mathscr{V}_{\Gamma}) \in \operatorname{H}^{2k}(M_{\Gamma}^*)$ which behave like the Chern classes of a putative extension \mathscr{V}_{Γ}^* to M_{Γ}^* . The main property is that for $\pi: M_{\Gamma}^{\Sigma} \to M_{\Gamma}^*$ a smooth toroidal desingularization [3], the pullback $\pi^*(c_k^*(\mathscr{V}_{\Gamma})) = c_k(\mathscr{V}_{\Gamma}^{\Sigma})$ is the Chern class of Mumford's canonical extension $\mathscr{V}_{\Gamma}^{\Sigma}$ [27, 32]. It is a well-known consequence of Mumford's generalization of Hirzebruch proportionality that the classes $c_k(\mathscr{V}_{\Gamma}^{\Sigma})$ generate a copy of $\operatorname{H}^*(X^c)$ in $\operatorname{H}^*(M_{\Gamma}^{\Sigma}) = (-1)^k c_k(\mathscr{V}_{\Gamma}^{\Sigma})$ for all k, \mathscr{V} (see Lemma C.1 for a proof, following [34, Lemma 3.7.2]). They are contained in $\operatorname{Gr}_*^W \operatorname{H}^*(M_{\Gamma}^*) = \operatorname{im}\left(\operatorname{H}^*(M_{\Gamma}^*) \to \operatorname{H}^*(M_{\Gamma}^{\Sigma})\right)$, since $\pi^*(c_k^*(\mathscr{V}_{\Gamma})) = c_k(\mathscr{V}_{\Gamma}^{\Sigma})$ by [25]. Moreover, the compatibility of the construction for different Σ (see, e.g., [27, Section 4.3.1]) shows that we have a well-defined embedding $\theta: \operatorname{H}^*(X^c) \to \operatorname{Gr}_*^W \operatorname{H}^*(M_{\Gamma}^*)$.

It remains to show that the classes are $G(\mathbb{Q})$ -invariant and the embedding is functorial. The direct limit $\operatorname{colim}_{\Sigma,\Gamma} \operatorname{H}^*(M_{\Gamma}^{\Sigma})$ over all pairs (Σ,Γ) where Σ is admissible for Γ is a $G(\mathbb{Q})$ -module, and contains $\operatorname{Gr}^W_* \operatorname{H}^*(\mathcal{M}^*)$ as a $G(\mathbb{Q})$ -submodule. Standard properties of the canonical extensions listed in [27, Section 4.3] show that the Chern classes are $G(\mathbb{Q})$ -invariants in $\operatorname{colim}_{\Sigma,\Gamma}\operatorname{H}^*(M_{\Gamma}^{\Sigma})$ and hence in $\operatorname{Gr}^W_*\operatorname{H}^*(\mathcal{M}^*)$. Finally, functoriality follows from [27, Section 4.3.4] and the definition of the map θ given in Lemma C.1.

Example 3.7. The representation of K on the top exterior power of \mathfrak{p} , where $\mathfrak{g} = \mathfrak{k} + \mathfrak{p}$ is the Cartan decomposition given by K, gives a special automorphic bundle called the Baily–Borel bundle. This extends as a line bundle \mathscr{L}^{bb} over M^*_{Γ} , and some power of \mathscr{L}^{bb} is the $\mathscr{O}(1)$ in the Baily–Borel projective embedding (see [32, Prop. 3.4(b)]). So \mathscr{L}^{bb} is

ample and the Chern class $c_1^{bb}:=c_1^*(\mathscr{L})=c_1\left(\mathscr{L}^{bb}\right)$ fixes a generator

$$\left(c_1^{bb}\right)^n \in \operatorname{Gr}_{2n}^W \mathrm{H}^{2n}(\mathscr{M}^*) = \mathrm{IH}^{2n}(\mathscr{M}^*)$$
(3.13)

in top degree.

Example 3.8. If $G(\mathbb{R})^{nc} = SU(1,n)$, then $X^c = SU(1+n)/S(U(1) \times U(n)) \cong \mathbb{P}^n$. So the invariant part of $\operatorname{Gr}^W_* \operatorname{H}^*(\mathscr{M}^*)$ is $\mathbb{C}\left[c_1^{bb}\right]/\left(\left(c_1^{bb}\right)^{n+1}\right)$. (See, e.g., [35, Section 1.2] for an intrinsic description of \mathscr{L}^{bb} .)

Example 3.9. If $G(\mathbb{R})^{nc} = Spin(2,n)$, the compact symmetric space dual to X is

$$X^{c} = Spin(2+n)/Spin(2) \times_{\{\pm I\}} Spin(n) = SO(2+n)/SO(2) \times SO(n),$$

which is a quadric in \mathbb{P}^{n+1} . The complex cohomology ring of quadrics is well known. Let E_2 and E_n be the vector bundles on X^c corresponding to the natural representations of $SO(2) \times SO(n)$ of dimension 2 and *n*, respectively. When *n* is odd, the complex cohomology is generated by the Euler class (or first Chern class) $c_1 = c_1(E_2) \in \mathrm{H}^2(X^c) - \mathrm{that}$ is, $\mathrm{H}^*(X^c) = \mathbb{C}[c_1]/(c_1^{n+1})$. When n = 2d is even, the complex cohomology of X^c is generated as a ring by c_1 and the Euler class $c_d = c_d(E_n) \in \mathrm{H}^{n=2d}(X^c)$, with the relations $c_1^{n+1} = 0, c_d^2 = (-1)^d c_1^{2d}$, and $c_1 c_d = 0$.

For later use, we remark that if $n = 2d \ge 4$ and $X_a^c \subset X^c$ is the inclusion of quadrics coming from $SO(2,a) \subset SO(2,n)$ for a < n, then $c_d|_{X_a^c} = 0$. Indeed, $0 = (c_1c_d)|_{X_{n-1}^c} = c_1|_{X_{n-1}^c} \cdot c_d|_{X_{n-1}^c} = c_1 \cdot (c_d|_{X_{n-1}^c})$. Since $c_1 \cdot$ is injective on $H^*(X_{n-1}^c) = \mathbb{C}[c_1]/(c_1^n)$ in degrees < 2(n-1), we must have $c_d|_{X_{n-1}^c} = 0$ if $d \ge 2$.

3.3. Cycle classes and an injectivity criterion

We will prove a criterion for the nonvanishing of Res between Shimura varieties and apply it to prove Theorem 3.1 and Corollary 3.2. Since we are only interested in cohomology with complex coefficients, we will ignore Tate twists henceforth and write $H^*(X)$ for $H^*(X,\mathbb{C})$.

Suppose now that $\iota: M^*_{H,\Gamma_H} \to M^*_{\Gamma}$ is the extension to minimal compactifications of a morphism of Shimura varieties (compare §2.4), and let $n = \dim M_{\Gamma, m} = \dim M_{H,\Gamma_H}$. Let

$$\xi_{\Gamma} := \left(\cap [M_{\Gamma}^*] \right)^{-1} \left(\iota_* \left[M_{H,\Gamma_H}^* \right] \right) \in \operatorname{Gr}_{2(n-m)}^W \operatorname{H}^{2(n-m)}(M_{\Gamma}^*)$$

be the cycle class defined earlier in Proposition 3.3, ignoring Tate twists and simplifying the notation (in the notation of that proposition, this would be $\xi_{M_{\Gamma}^*,M_{H,\Gamma_H}^*}$). It is easily checked that if $\Gamma' \subset \Gamma$ is normal, then $\xi_{\Gamma} = |\Gamma/\Gamma'|^{-1} \sum_{\gamma \in \Gamma/\Gamma'} \gamma \cdot \xi_{\Gamma'}$, where $\xi_{\Gamma'} = (\cap [M_{\Gamma}^*])^{-1} \left(\iota'_* \left(\left[M_{H,\Gamma'_H}^* \right] \right) \right)$ for $\iota' : M_{H,\Gamma'_H}^* \to M_{\Gamma'}^*$ at level Γ' .

We will also consider the closed immersion $\iota^c: X_H^c \to X^c$, which gives the cycle class

$$\xi_{X_H^c} := (\cap [X^c])^{-1} \left(\iota_*^c [X_H^c] \right),$$

which is nonzero since X_H^c is a subvariety of the algebraic variety X^c .

1672

Proposition 3.10. The $G(\mathbb{Q})$ -submodule of $\operatorname{IH}^*(\mathscr{M}^*)$ generated by ξ_{Γ} contains the cycle class $\xi_{X_{H}^c} \in \operatorname{H}^{2(n-m)}(X^c)$ of X_{H}^c in X^c .

Proof. The $G(\mathbb{Q})$ -submodule $V \subset \operatorname{Gr}_{2(n-m)}^W \operatorname{H}^{2(n-m)}(\mathscr{M}^*) \subset \operatorname{IH}^{2(n-m)}(\mathscr{M}^*)$ generated by ξ_{Γ} admits a decomposition $V = V^0 \oplus V^1$, where V^1 has no invariants or coinvariants and V^0 is contained in the summand of invariants $\operatorname{H}^{2(n-m)}(X^c) \subset \operatorname{Gr}_{2m}^W \operatorname{H}^{2m}(M_{\Gamma}^*)$. (This is because of semisimplicity in Proposition 3.6.) Write $\xi_{\Gamma} = \xi_{\Gamma}^0 + \xi_{\Gamma}^1$, with $\xi_{\Gamma}^i \in V^i$. Since V^1 has no coinvariants, $\xi_{\Gamma}^0 \cdot \alpha = \xi_{\Gamma} \cdot \alpha$ for any $\alpha \in \operatorname{H}^{2m}(X^c)$, so that

$$[M_{\Gamma}^{*}] \cap \left(\xi_{\Gamma}^{0} \cdot \alpha\right) = [M_{\Gamma}^{*}] \cap \left(\xi_{\Gamma} \cdot \alpha\right)$$
$$= [M_{\Gamma}^{*}] \cap \xi_{\Gamma} \cap \alpha$$
$$= \iota_{*} \left[M_{H,\Gamma_{H}}^{*}\right] \cap \alpha$$
$$= \iota_{*} \left(\left[M_{H,\Gamma_{H}}^{*}\right] \cap \iota^{*}(\alpha)\right).$$
(3.14)

1673

On the other hand,

$$[X^{c}] \cap \left(\xi_{X_{H}^{c}} \cdot \alpha\right) = [X^{c}] \cap \xi_{X_{H}^{c}} \cap \alpha$$
$$= \iota_{*}^{c} [X_{H}^{c}] \cap \alpha$$
$$= \iota_{*}^{c} ([X_{H}^{c}]] \cap \iota^{c*}(\alpha)).$$
(3.15)

Now the pullbacks ι^{c*} and ι^* are compatible, while under the isomorphism $\mathrm{H}^{2n}(X^c) \cong \mathrm{Gr}_{2n}^W \mathrm{H}^{2n}(M_{\Gamma}^*)$ the two nonzero linear forms $[M_{\Gamma}^*] \cap : \mathrm{Gr}_{2n}^W \mathrm{H}^{2n}(M_{\Gamma}^*) \to \mathrm{H}_0(M_{\Gamma}^*) = \mathbb{C}$ and $[X^c] \cap : \mathrm{H}^{2n}(X^c) \to \mathrm{H}_0(X^c) = \mathbb{C}$ are necessarily proportional; and the same holds for M_{H,Γ_H}^* and X_H^c . It follows from equations (3.14) and (3.15) that for $\alpha \in \mathrm{H}^{2m}(X^c)$, we have

$$[X^c] \cap \left(\xi_{\Gamma}^0 \cdot \alpha\right) \sim [X^c] \cap \left(\xi_{X_H^c} \cdot \alpha\right),$$

where ~ means up to a fixed nonzero constant independent of α . Thus $\xi_{\Gamma}^0 \cdot \alpha \sim \xi_{X_H^c} \cdot \alpha$ for any α , and so $\xi_{\Gamma}^0 \sim \xi_{X_H^c}$ by Poincaré duality for X^c .

Theorem 3.11. If $\alpha \in \operatorname{Gr}_{i}^{W} \operatorname{H}^{i}(\mathscr{M}^{*})$ and $\operatorname{Res}(\alpha) = 0$, then $\alpha \cdot \xi_{X_{H}^{c}} = 0$.

Proof. Suppose that $\alpha \in \operatorname{Gr}_i^W \operatorname{H}^i(M_{\Gamma}^*) \subset \operatorname{Gr}_i^W \operatorname{H}^i(\mathscr{M}^*)$ is such that $\operatorname{Res}(\alpha) = 0$. Set $g \in G(\mathbb{Q})$ and choose Γ' normal in Γ with $\Gamma' \subset \Gamma \cap g^{-1}\Gamma g$. Let $\gamma_1, \ldots, \gamma_r$ be the representatives for cosets of Γ' in Γ , let $p: M_{\Gamma'}^* \to M_{\Gamma}^*$, and let $\iota': M_{H,\Gamma'_H}^* \to M_{\Gamma'}^*$ be the natural map at level Γ' . Then we have

$$p^{-1}\left(\iota\left(M_{H,\Gamma_{H}}^{*}\right)\right) = \bigcup_{i} \gamma_{i} \cdot \iota'\left(M_{H,\Gamma_{H}'}^{*}\right)$$

If $\operatorname{Res}(\alpha) = 0$ for $\alpha \in \operatorname{Gr}_i^W \operatorname{H}^i(M_{\Gamma}^*)$, then $(g\gamma_i^{-1})^* \alpha = (\gamma_i^{-1})^* g^* \alpha$ restricts to zero on $\iota'\left(M_{H,\Gamma'_H}^*\right)$ for each i – that is, $\iota'^*\left((\gamma_i^{-1})^* g^*(\alpha)\right) = 0$ for each i. Using equation (3.4), we have

$$0 = \iota'_* \left(\iota'^* \left(\left(\gamma_i^{-1} \right)^* g^* \alpha \right) \cap \left[M_{H, \Gamma'_H}^* \right] \right)$$

$$= \left(\gamma_i^{-1} \right)^* g^* \alpha \cap \iota'_* \left(\left[M_{H, \Gamma'_H}^* \right] \right)$$

$$= \left[M_{\Gamma'}^* \right] \cap \left(\left(\gamma_i^{-1} \right)^* g^* \alpha \cdot \xi_{\Gamma'} \right)$$

$$= \left[M_{\Gamma'}^* \right] \cap \left(g^* \alpha \cdot \gamma_i^* \xi_{\Gamma'} \right).$$

(3.16)

By Proposition 3.3(2), we have that $g^* \alpha \cdot \gamma_i^* \xi_{\Gamma'} = 0$. Summing over Γ / Γ' gives

$$0 = g^* \alpha \cdot \xi_{\Gamma} = \alpha \cdot \left(g^{-1}\right)^* \xi_{\Gamma}$$

Since this holds for all $g \in G(\mathbb{Q})$, Proposition 3.10 implies that $\alpha \cdot \xi_{X_H^c} = 0$.

Proof of Theorem 3.1. In the case $SU(1,m) \subset SU(1,n)$, we have $X^c = \mathbb{P}^n$, so that $\xi_{X_H^c} \sim (c_1^{bb})^{n-m}$, where c_1^{bb} is the first Chern class of the ample Baily–Borel line bundle in formula (3.13). So $\cdot c_1^{bb}$ is injective in degrees < n on $\bigoplus_i \operatorname{Gr}_i^W \operatorname{H}^i(M_{\Gamma}^*) \subset \operatorname{IH}^*(M_{\Gamma}^*)$ because of the hard Lefschetz property for intersection cohomology [6, 42], and hence $\cdot \xi_{X_H^c}$ is injective in degrees $i \leq m$. Theorem 3.11 implies the injectivity of Res in degrees $\leq m$.

In the case $SO(2,m) \subset SO(2,n)$, the previous argument can be applied to the simply connected covers $\tilde{H} \subset \tilde{G}$. Now we claim $\xi_{X_{H}^{c}} \sim (c_{1}^{bb})^{n-m}$. If $n \neq 2m$ this is clear, since $\operatorname{Gr}_{2(n-m)}^{W}\operatorname{H}^{2(n-m)}(M_{\Gamma}^{*}) = \mathbb{C}(c_{1}^{bb})^{n-m}$, whereas if n = 2m, then it holds because $c_{m}|_{X_{H}^{c}} = 0$ (see Example 3.9). It follows that $\cdot \xi_{X_{H}^{c}}$ is injective on $\bigoplus_{i} \operatorname{Gr}_{i}^{W}\operatorname{H}^{i}(M_{\Gamma}^{*})$ in degrees $\leq m$ because of the hard Lefschetz property of $c_{1}^{bb} \cdot$ on $\operatorname{IH}^{*}(M_{\Gamma}^{*})$. Theorem 3.11 implies the injectivity of Res in degrees $\leq m$.

Proof of Corollary 3.2. The complex hyperbolic case follows easily from the observation that

$$\mathrm{H}^{k}_{!}(M_{\Gamma}) = \mathrm{Gr}^{W}_{k}\mathrm{H}^{k}_{c}(M_{\Gamma}) \qquad \text{for} k \leq n$$

(see the proof of [36, Proposition 1.6]) and the fact that $\operatorname{Gr}_k^W \operatorname{H}_c^k(M_{\Gamma}) \subset \operatorname{Gr}_k^W \operatorname{H}^k(M_{\Gamma}^*)$ for all k (Remark 3.4).

Now consider the orthogonal case. First note that

$$\mathbf{H}_{!}^{i}(M_{\Gamma}) = \operatorname{im}\left(\mathbf{H}_{c}^{i}(M_{\Gamma}) \to \operatorname{IH}^{i}(M_{\Gamma}^{*})\right) = \operatorname{Gr}_{i}^{W}\mathbf{H}_{c}^{i}(M_{\Gamma}) \qquad \text{for} i \leq n-1.$$
(3.17)

The first equality holds because $\operatorname{IH}^{i}(M_{\Gamma}^{*}) \to \operatorname{H}^{i}(M_{\Gamma})$ is injective for $i \leq n-1$ (and an isomorphism for $i \leq n-2$), because the boundary has dimension 1, and the second holds because $\operatorname{Gr}_{i}^{W}\operatorname{H}_{c}^{i}(M_{\Gamma}) \hookrightarrow \operatorname{IH}^{i}(M_{\Gamma}^{*})$ for all i (see Remark 3.4). On the other hand,

$$\operatorname{Gr}_{i}^{W}\operatorname{H}_{c}^{i}(M_{\Gamma}) \subset \operatorname{Gr}_{i}^{W}\operatorname{H}^{i}(M_{\Gamma}^{*})$$

for all *i* (by Remark 3.4 again). Thus $\mathrm{H}^{i}_{!}(\mathscr{M}) \subset \mathrm{Gr}^{W}_{i}\mathrm{H}^{i}(\mathscr{M}^{*})$ for $i \leq n-1$, and similarly for \mathscr{M}_{H} . So the corollary follows from Theorem 3.1 in degrees $\leq m-1$ in the case $rk_{\mathbb{Q}}(G) = 2$. If $rk_{\mathbb{Q}}(H) = 1$, then this can be improved slightly, because the singularities of $M^{*}_{H,\Gamma_{H}}$ are isolated and so $\mathrm{H}^{i}_{!}(\mathscr{M}) \subset \mathrm{Gr}^{W}_{i}\mathrm{H}^{i}(\mathscr{M}^{*})$ holds for i = m also.

1674

Remark 3.12. The use of the embedding $\mathrm{H}^*(X^c) \subset \mathrm{Gr}^W_*\mathrm{H}^*(\mathscr{M}^*)$ (of Proposition 3.6) in the proof of Theorem 3.1 can be avoided in the unitary case and in the orthogonal case (except possibly when n = 2m). As noted in Example 3.7, the first Chern class of the Baily–Borel bundle can be made sense of in $\mathrm{Gr}^W_*\mathrm{H}^*(\mathscr{M}^*)$, and the invariants in $\mathrm{H}^{2(n-m)}(M^*_{\Gamma})$ are reduced to $\mathbb{C}(c_1^{bb})^{n-m}$. So the argument can be run with $\mathrm{H}^*(X^c)$ replaced by the subring $\bigoplus_{0 \leq i \leq n} \mathbb{C}(c_1^{bb})^i$.

Remark 3.13. The foregoing arguments can be modified to treat nontrivial coefficients, using the fact that the local system E on M_{Γ} underlies a pure polarizable variation of Hodge structure (see [31, §4] or, for a more canonical approach in the context of Shimura varieties, see [19]). This allows us to use mixed Hodge modules and arguments with weights.

Remark 3.14. The criterion of [47] in the compact case has been used in [9] to prove a number of other results about restriction using computations in the compact dual. The analogues for the top weight quotient of $H^*(\mathcal{M}^*)$ in general then follow immediately using the criterion of Theorem 3.11 instead. It seems likely that (suitably formulated) they should extend to $H^*(\mathcal{M})$ using the methods of later sections.

The following example shows that these bounds can sometimes be improved on:

Example 3.15. Consider the case of $SO(2,2) \subset SO(2,n)$ for $n \geq 3$, and assume $rk_{\mathbb{Q}}(H) = 2$, so that $H = SL(2) \times SL(2)$. Theorem 3.1 gives the injectivity of $\operatorname{Gr}_2^W \operatorname{H}^2(\mathscr{M}^*) \to I_H^G \operatorname{Gr}_2^W \operatorname{H}^2(\mathscr{M}^*_H)$. Since $\operatorname{H}^2_!(M_{\Gamma}) = \operatorname{IH}^2(M_{\Gamma}^*) = \operatorname{Gr}_2^W \operatorname{H}^2_c(M_{\Gamma})$, the map $\operatorname{Gr}_2^W \operatorname{H}^2(\mathscr{M}^*_{\Gamma}) \to \operatorname{Gr}_2^W(\mathscr{M}^*_{H,\Gamma_H})$ factors as

$$\operatorname{Gr}_{2}^{W}\operatorname{H}_{c}^{2}(M_{\Gamma}) \to \operatorname{Gr}_{2}^{W}\operatorname{H}_{c}^{2}(M_{H,\Gamma_{H}}) \to \operatorname{Gr}_{2}^{W}\operatorname{H}^{2}(M_{H,\Gamma_{H}}^{*})$$

Now $M_{H,\Gamma_H} = X_1 \times X_2$ is a product of two modular curves, so $\operatorname{Gr}_2^W \operatorname{H}_c^2(X_1 \times X_2) = \operatorname{Gr}_1^W \operatorname{H}_c^1(X_1) \otimes \operatorname{Gr}_1^W \operatorname{H}_c^1(X_2)$ injects into $\operatorname{Gr}_2^W \operatorname{H}^2(X_1^* \times X_2^*)$, and so the second map is injective. It follows that Res is always injective on $\operatorname{H}_1^2(\mathcal{M})$, improving Corollary 3.2 slightly.

Remark 3.16. In fact, when m = 2 and n = 3, Res is injective on all of $H^2(\mathcal{M})$ by a result of Weissauer [51]. This is not covered by our results, since $H^2_!(\mathcal{M})$ is a proper subspace of $H^2(\mathcal{M})$.

Remark 3.17. The basic idea of this section is that in the presence of some functoriality, semisimplicity, and duality, one can use the averaging argument for the cycle class. This can also be applied to the reductive Borel–Serre (RBS) compactification, to get a slight generalization of Theorems 3.11 and 3.1.

The RBS compactification is a (nonalgebraic) compactification of M_{Γ} dominating M_{Γ}^* – that is, the identity of M_{Γ} extends to $M_{\Gamma}^{rbs} \to M_{\Gamma}^*$. The cohomology $\mathrm{H}^i(M_{\Gamma}^{rbs})$ carries a mixed Hodge structure like that of a proper variety – that is, with weights $\leq i$ in degree i – and the top weight quotient $\mathrm{Gr}_i^W \mathrm{H}^i(M_{\Gamma}^{rbs})$ is the image of a natural map $\mathrm{H}^i(M_{\Gamma}^{rbs}) \to \mathrm{IH}^i(M_{\Gamma}^*)$. For $\iota: M_{H,\Gamma_H} \to M_{\Gamma}$ there is no continuous map $M_{H,\Gamma_H}^{rbs} \to M_{\Gamma}^{rbs}$ extending $M_{H,\Gamma_H} \to M_{\Gamma}$, but nevertheless there is a natural pullback map $\mathrm{H}^*(M_{\Gamma}^{rbs}) \to \mathrm{H}^*(M_{H,\Gamma_H}^{rbs})$, which is a homomorphism of mixed Hodge structures. (See, e.g., the survey [37], where these results are discussed.) Theorem 3.1 can be improved to injectivity of the induced map Res on $\mathrm{Gr}_i^W \mathrm{H}^i(\mathcal{M}^{rbs}) = \mathrm{colim}_{\Gamma}\mathrm{Gr}_i^W \mathrm{H}^i(\mathcal{M}_{\Gamma}^{rbs})$ for $i \leq m$. Since the canonical mapping $\mathrm{H}^i(\mathcal{M}_{\Gamma}^*) \to \mathrm{IH}^*(\mathcal{M}_{\Gamma}^*)$ factors through $\mathrm{H}^*(\mathcal{M}_{\Gamma}^{rbs})$, it follows that $\mathrm{Gr}_i^W \mathrm{H}^i(\mathcal{M}^*) \subset \mathrm{Gr}_i^W \mathrm{H}^i(\mathcal{M}^{rbs})$. (However, in general $\mathrm{H}^i(\mathcal{M}^*) \to \mathrm{H}^*(\mathcal{M}^{rbs})$ is not injective.) It can be shown using methods of Eisenstein series that in the case at hand, this inclusion is proper, so we would have an improvement of Theorem 3.1.

4. Lie-algebra cohomology computations

4.1. Kostant's theorem

We recall results of [29]. Fix a complex semisimple Lie group G, a maximal torus $T \subset G$, and a Borel subgroup $B \supset T$, and let $\Phi = \Phi(T,G)$ be the root system, Φ^+ the positive roots determined by B, $\Phi^- = -\Phi^+$ the negative roots, $\rho = \frac{1}{2} \sum_{\alpha \in \Phi^+} \alpha$ the half-sum of positive roots, and W = W(T,G) the Weyl group of T in G. Let P be a standard parabolic subgroup of G, N its unipotent radical, and \mathfrak{n} its Lie algebra. The Weyl group of the Levi L = P/N is a subgroup $W_L \subset W$, and we let W^P be the set of minimal length coset representatives of $W_L \setminus W$. For each $w \in W^P$, the associated set of positive roots

$$\Phi(w) = \{ \alpha \in \Phi(T, G) : \alpha > 0, w^{-1}\alpha < 0 \} = \Phi^+ \cap w\Phi^-,$$

which has cardinality $\ell(w)$. For a dominant $\lambda \in X^*(T)$, the weights $w(\lambda + \rho) - \rho$ for $w \in W^P$ are dominant for L and distinct. The mapping $w \mapsto \Phi(w)$ sets up a bijection between W^P and the subsets S of $\Phi(\mathfrak{n}) = \{\alpha \in \Phi(T,G) : \mathfrak{g}^{\alpha} \subset \mathfrak{n}\}$ for which both S and $\Phi^+ - S$ are closed under + [29, 5.10] (recall that $\alpha + \beta$ is $\alpha + \beta$ when this is a root and empty otherwise).

Let E_{λ} be the irreducible finite-dimensional *G*-representation with highest weight $\lambda \in X^*(T)$ with respect to *B*. The Lie-algebra cohomology $H^*(\mathfrak{n}, E_{\lambda})$ is the cohomology of $\wedge^*\mathfrak{n}^* \otimes E_{\lambda}$ with the Lie-algebra differential. The natural *P*-module structure on $\wedge^*\mathfrak{n}^* \otimes E_{\lambda}$ descends to an L = P/N-module structure in cohomology. For an *L*-dominant weight $\mu \in X^*(T)$, let E^L_{μ} denote the irreducible finite-dimensional algebraic representation of *L* with highest weight μ . Then by [29, Theorem 5.14] there is a multiplicity-free decomposition of *L*-modules

$$\mathbf{H}^{k}(\mathbf{n}, E_{\lambda}) = \bigoplus_{w \in W^{P}, \ell(w) = k} E^{L}_{w(\lambda + \rho) - \rho}.$$
(4.1)

Kostant also identified a highest weight vector in each summand. Let \mathfrak{n}^- be the nilradical of the Lie algebra of the parabolic subgroup opposite to P. The Killing form gives isomorphisms $\mathfrak{n}^- \cong \mathfrak{n}^*$ and $\wedge^i \mathfrak{n}^- \cong \wedge^i \mathfrak{n}^*$. Choose a nonzero vector e_α in the root space \mathfrak{g}^α for each $\alpha \in \Phi(T, G_{\mathbb{C}})$, and for $w \in W^P$ define $e_w := \bigwedge_{\alpha \in \Phi(w)} e_{-\alpha} \in \wedge^{\ell(w)} \mathfrak{n}^-$. Let $v_{w\lambda} \in E_\lambda$ be a weight vector for the extremal weight $w\lambda$. Then under the identification of \mathfrak{n}^* with \mathfrak{n}^- , the element

$$e_w \otimes v_{w\lambda} \in \wedge^{\ell(w)} \mathfrak{n}^- \otimes E_\lambda \tag{4.2}$$

is closed in $\wedge^* \mathfrak{n}^* \otimes E_{\lambda}$ and its cohomology class is a highest weight vector for the summand $E_{w(\lambda+\rho)-\rho}^L$ in equation (4.1) [29, Theorem 5.14]; it is a harmonic representative for a natural Laplacian. A lowest weight vector is given by $\wedge_{\alpha \in \Phi(w)} e_{-w_0^L(\alpha)} \otimes v_{w_0^L w \lambda}$, where w_0^L is the longest element of $W_L \subset W$ [29, Remark 8.2]. In fact, taking the sum of the *L*-submodules of $\wedge^* \mathfrak{n}^- \otimes E_{\lambda}$ generated by the $e_w \otimes v_{w\lambda}$ as w runs over W^P gives (using the identification $\wedge^* \mathfrak{n}^* \cong \wedge^* \mathfrak{n}^-$) a canonical *L*-equivariant inclusion $H^*(\mathfrak{n}, E_{\lambda}) \subset \wedge^* \mathfrak{n}^* \otimes E_{\lambda}$ inducing the identity in cohomology and compatible with products (see [29, Theorem 5.7]).

4.2. Restriction maps in n-cohomology

Now consider the situation where $\iota: H \to G$ is a homomorphism of real semisimple groups with finite kernel. Then for a parabolic P of G with Levi L = P/N, we have the parabolic $P_H = \iota^{-1}(P)$ of H with unipotent radical $N_H = \iota^{-1}(N)$ and Levi $L_H = P_H/N_H$. Let $\mathfrak{n} = \operatorname{Lie} N(\mathbb{R})$ and $\mathfrak{n}_H = \operatorname{Lie} N_H(\mathbb{R})$. For a finite-dimensional $G(\mathbb{C})$ -representation E and E_H a summand of $E|_{H(\mathbb{C})}$, the restriction map

$$\mathrm{H}^*(\mathfrak{n}_{\mathbb{C}}, E) \longrightarrow \mathrm{H}^*(\mathfrak{n}_{H,\mathbb{C}}, E) \longrightarrow \mathrm{H}^*(\mathfrak{n}_{H,\mathbb{C}}, E_H)$$

is $L_H(\mathbb{C})$ -equivariant. Consider the map

$$Res_{\mathfrak{n}}: \mathrm{H}^{*}(\mathfrak{n}_{\mathbb{C}}, E) \longrightarrow \prod_{m \in L(\mathbb{C})} \mathrm{H}^{*}(\mathfrak{n}_{H,\mathbb{C}}, E_{H})$$

with coordinate indexed by $m \in L(\mathbb{C})$ given by precomposing the previous map with the adjoint action of m. Note that the kernel of Res_n is an $L(\mathbb{C})$ -module; in particular, for each irreducible summand $E^L_{w(\lambda+\rho)-\rho}$, we have that Res_n is injective on $E^L_{w(\lambda+\rho)-\rho} \iff \operatorname{Res}_n$ is nonzero on $E^L_{w(\lambda+\rho)-\rho} \iff \operatorname{Res}_n(e_w \otimes v_{w\lambda}) \neq 0$.

Let us assume that the restriction of finite-dimensional irreducible representations by $H(\mathbb{C}) \subset G(\mathbb{C})$ is multiplicity-free. Choose maximal tori $T_H \subset T$ and Borel subgroups $B_H \subset B$ (of $H\mathbb{C}) \subset G(\mathbb{C})$), and for an irreducible representation E with highest weight $\lambda \in X^*(T)$, let E_H be the summand of $E|_{H(\mathbb{C})}$ with highest weight $\lambda|_{T_H} \in X^*(T_H)$. The following propositions are proved by explicit elementary calculations with roots and weights using Kostant's theorem and take up the rest of this section. In each case, the restriction from $G(\mathbb{C})$ to $H(\mathbb{C})$ is multiplicity-free by classical results ([22, Section 8.1.1]).

Proposition 4.1. Let G = SO(d,1) and H = SO(c,1) for $2 \le c < d$ embedded in the standard way in G. Let P = LN be a proper parabolic subgroup of G. Then Res_n is injective in degrees $i \le c/2$ except in the case (d,c,i) = (2k+1,2k,k). In this case, $H^k(n,E)$ has two L-irreducible summands and Res_n is injective on either one.

Proposition 4.2. Let G = SU(n,1) and H = SU(m,1) for $2 \le m < n$ embedded in the standard way in G. Let P = MW be a proper parabolic subgroup of G. Then $\text{Res}_{\mathfrak{w}}$ is injective on $H^i(\mathfrak{w}, E)$ in degrees i < m.

Proposition 4.3. Let G = SO(2,n) and H = SO(2,m), embedded in G in the standard way for $2 \le m < n$, and $E = \mathbb{C}$. If P = MW is the stabilizer of an isotropic plane, then $\operatorname{Res}_{\mathfrak{w}}$ is injective on $\operatorname{H}^{i}(\mathfrak{w}, \mathbb{C})$ for $i \le m - 2$.

To treat $SO(1,n) \subset SU(1,n)$, we will need the following:

Proposition 4.4. Let G = SU(n,1) and H = SO(n,1), embedded in the standard way in G with $n \ge 2$. Let P = MW be a proper parabolic subgroup of G. Then $\text{Res}_{\mathfrak{w}}$ is injective on $\text{H}^{i,0}(\mathfrak{w},\mathbb{C})$ in degrees i < n.

The bigrading in Proposition 4.4 refers to the Hodge structure coming from the identification of $\mathrm{H}^{i}(\mathfrak{w},\mathbb{C})$ with the link cohomology $\mathrm{H}^{i}\left(i_{\{x\}}^{*}j_{*}\mathbb{C}\right)$, where $j: M_{\Gamma} \hookrightarrow M_{\Gamma}^{*}$ and $i_{\{x\}}: \{x\} \hookrightarrow M_{\Gamma}^{*}$ is the inclusion of the cusp corresponding to P (see Proposition 2.7 or [36, Lemma 1.2].) It can also seen from the decomposition (4.1) [36, Remark 1.11].

The rest of this section will be taken up with the proofs of these propositions.

4.3. Proof of Proposition 4.3

To make computations we will fix some notation for roots. We may assume $\mathfrak{g} = \mathfrak{so}(2,n) = \mathfrak{so}(J)$, where

$$J = \begin{pmatrix} & & & 1 \\ & & 1 & \\ & I_{n-2} & & \\ & 1 & & & \\ 1 & & & & \end{pmatrix}.$$

Fix a Cartan subalgebra \mathfrak{s} of $\mathfrak{so}(n-2)_{\mathbb{C}}$ and let $\mathfrak{t} \subset \mathfrak{g}_{\mathbb{C}}$ be the Cartan subalgebra defined by

$$\mathfrak{t} := \{ diag(a, d, C, -d, -a) : a, d \in \mathbb{C}, C \in \mathfrak{s} \}.$$

Then \mathfrak{t} is defined and maximally split over \mathbb{R} , and the subspace $\mathfrak{a}_{\mathbb{C}} \subset \mathfrak{t}$ given by C = 0is the complexification of the Lie algebra $\mathfrak{a} \subset \mathfrak{g}$ of a maximal \mathbb{R} -split Cartan in \mathfrak{g} . Let $\alpha_1, \alpha_2 \in \mathfrak{t}^*$ be defined by

$$\alpha_1(\operatorname{diag}(a,d,C,-d,-a)) = a,$$

$$\alpha_2(\operatorname{diag}(a,d,C,-d,-a)) = d.$$
(4.3)

The relative roots are $\Phi(\mathfrak{a}_{\mathbb{C}},\mathfrak{g}_{\mathbb{C}}) = \{\pm \alpha_1, \pm \alpha_2, \pm (\alpha_1 - \alpha_2), \pm (\alpha_1 + \alpha_2)\}.$

Now choose for \mathfrak{s} the Cartan subalgebra of block-diagonal matrices

$$\mathfrak{s} = \left\{ diag\left(\begin{pmatrix} 0 & b_1 \\ -b_1 & 0 \end{pmatrix}, \dots, \begin{pmatrix} 0 & b_k \\ -b_k & 0 \end{pmatrix} \right) : b_1, \dots, b_k \in \mathbb{C} \right\}$$
(4.4)

when n-2=2k is even and

$$\mathfrak{s} = \left\{ diag\left(\begin{pmatrix} 0 & b_1 \\ -b_1 & 0 \end{pmatrix}, \dots, \begin{pmatrix} 0 & b_k \\ -b_k & 0 \end{pmatrix}, 0 \right) : b_1, \dots, b_k \in \mathbb{C} \right\}$$
(4.5)

when n-2=2k+1 is odd. Let $\eta_1,\ldots,\eta_k\in\mathfrak{s}^*$ be defined by

$$\eta_i \left(diag \left(\begin{pmatrix} 0 & b_1 \\ -b_1 & 0 \end{pmatrix}, \dots, \begin{pmatrix} 0 & b_k \\ -b_k & 0 \end{pmatrix} \right), (0) \right) = \sqrt{-1} b_i, \tag{4.6}$$

where (0) means the entry is omitted when n-2 is even and the entry is zero when n-2 is odd. Fix the positive system $\Phi^+(\mathfrak{s},\mathfrak{so}(n-2)_{\mathbb{C}})$ with simple roots $\{\eta_i - \eta_{i+1} : 1 \leq i < k\} \sqcup$

 $\{\eta_k\}$ for n-2=2k+1 odd and $\{\eta_i - \eta_{i+1} : 1 \le i < k\} \sqcup \{\eta_k + \eta_{k-1}\}$ for n-2=2k even, and take the positive system in $\Phi(\mathfrak{t},\mathfrak{g}_{\mathbb{C}})$ containing it and compatible with $\Phi^+(\mathfrak{a}_{\mathbb{C}},\mathfrak{g}_{\mathbb{C}}) = \{\alpha_1, \alpha_2, \alpha_1 - \alpha_2, \alpha_1 + \alpha_2\}$. An explicit computation shows that the positive roots are

$$\Phi^+(\mathfrak{t},\mathfrak{g}_{\mathbb{C}}) = \Phi^+(\mathfrak{s},\mathfrak{so}(n-2)_{\mathbb{C}}) \sqcup \{\alpha_1 \pm \eta_i, \alpha_2 \pm \eta_i : 1 \le i \le k\} \sqcup \{\alpha_1 - \alpha_2, \alpha_1 + \alpha_2\}$$

for n-2=2k even and

$$\Phi^+(\mathfrak{t},\mathfrak{g}_{\mathbb{C}}) = \Phi^+(\mathfrak{s},\mathfrak{so}(n-2)_{\mathbb{C}}) \sqcup \{\alpha_1 \pm \eta_1, \alpha_2 \pm \eta_i : 1 \le i \le k\} \sqcup \{\alpha_1, \alpha_2, \alpha_1 + \alpha_2, \alpha_1 - \alpha_2\}$$

for n-2 = 2k+1 odd.

The parabolic P = MW in the proposition is the stabilizer of an isotropic plane, which may be assumed to be the obvious plane $\mathbb{R}e_1 + \mathbb{R}e_2$ in \mathbb{R}^{n+2} . We will need the set $\Phi(\mathfrak{w}) = \{\alpha \in \Phi(\mathfrak{t},\mathfrak{g}_{\mathbb{C}}) : \mathfrak{g}_{\mathbb{C}}^{\alpha} \subset \mathfrak{w}_{\mathbb{C}}\}$. Using the foregoing description and explicit matrix descriptions, we have

$$\Phi(\mathfrak{w}) = \{\alpha_1 \pm \eta_i\}_{1 \le i \le k} \sqcup \{\alpha_2 \pm \eta_i\}_{1 \le i \le k} \sqcup \{\alpha_1 + \alpha_2\}$$

if n-2=2k is even and

$$\Phi(\mathfrak{w}) = \{\alpha_1 \pm \eta_i\}_{1 \le i \le k} \sqcup \{\alpha_2 \pm \eta_i\}_{1 \le i \le k} \sqcup \{\alpha_1, \alpha_2, \alpha_1 + \alpha_2\}$$

if n-2=2k+1 is odd.

Now let us prove the proposition. We may assume that $\mathfrak{h} \subset \mathfrak{g}$ is given by the subspace $\mathbb{R}^{m+2} = (\mathbb{R}e_{m+1} + \cdots + \mathbb{R}e_n)^{\perp} \subset \mathbb{R}^{n+2}$ – that is, that $\mathfrak{h} = \mathfrak{so}(2,m)$ is embedded in $\mathfrak{g} = \mathfrak{so}(2,n)$ in a way that the $m+1, m+2, \ldots, n$ rows and columns are zero. We will consider the cases n even and n odd separately.

First assume n-2=2k is even. Then we have

$$\Phi(\mathfrak{w}) = \{\alpha_1 \pm \eta_i\}_{1 \le i \le k} \sqcup \{\alpha_2 \pm \eta_i\}_{1 \le i \le k} \sqcup \{\alpha_1 + \alpha_2\}.$$

For $w \in W^P$ of length $\leq m-2 = 2k - (n-m)$, the set of roots $\Phi(w) \subset \Phi(w)$ has cardinality $\leq 2k - (n-m)$, so that $\Phi^+ - \Phi(w)$ contains at least n-m elements which belong to $\{\alpha_2 \pm \eta_i\}_{1 \leq i \leq k}$, which has cardinality 2k. Since $\Phi^+ - \Phi(w)$ is closed under $\dot{+}$ and $\alpha_1 - \alpha_2 \in \Phi^+ - \Phi(w)$, we may choose sets I_+ and I_- in $\{1, \ldots, k\}$ such that the following are true:

- 1. I_+ and I_- are disjoint and $|I_+ \sqcup I_-| = n m$.
- 2. For $i \in I_+$, we have $\{\alpha_1 + \eta_i, \alpha_2 + \eta_i\} \subset \Phi^+ \Phi(w)$.
- 3. For $i \in I_-$, we have $\{\alpha_1 \eta_i, \alpha_2 \eta_i\} \subset \Phi^+ \Phi(w)$.

(These sets are not unique, but any choice suffices for our purposes.) Let \mathfrak{h}' be the copy of $\mathfrak{so}(2,m)$ given by the embedding of the subspace

$$\left(\bigoplus_{i\in I_+} \mathbb{R}e_{2i+2} \oplus \bigoplus_{i\in I_-} \mathbb{R}e_{2i+1}\right)^{\perp} \subset \mathbb{R}^{n+2};$$

that is, row and column 2i+2 are zero for $i \in I_+$ and row and column 2i+1 are zero for $i \in I_-$. Then the restriction of the harmonic representative $e_w = \wedge_{\alpha \in \Phi(w)} e_{-\alpha}$ in formula (4.2) to $\mathfrak{n}_{H'}$ is nonzero, and equals (up to a nonzero scalar) the harmonic representative of a similar class in $\mathfrak{n}_{H'}$. Since the subspace is conjugate to the subspace $(\mathbb{R}e_{m+1} + \cdots + \mathbb{R}e_n)^{\perp}$

by an element $m \in M(\mathbb{C}) = Spin(n-2,\mathbb{C})$, we see that \mathfrak{h}' is conjugate to \mathfrak{h} by $m \in M(\mathbb{C})$, and the highest weight vector e_w restricts nontrivially to $\operatorname{Ad}(m^{-1})(\mathfrak{h})$. This proves the proposition in this case.

Next assume n-2 = 2k+1 is odd. Then

$$\Phi(\mathfrak{w}) = \{\alpha_1 \pm \eta_i\}_{1 \le i \le k} \sqcup \{\alpha_2 \pm \eta_i\}_{1 \le i \le k} \sqcup \{\alpha_1, \alpha_2, \alpha_1 + \alpha_2\}.$$

For $w \in W^P$ of length $\leq m-2 = 2k+1-(n-m)$, the set of roots $\Phi(w) \subset \Phi(\mathfrak{w})$ has cardinality $\leq 2k+1-(n-m)$, so that $\Phi^+ - \Phi(w)$ contains at least n-m elements from the set $\{\alpha_2 \pm \eta_i : 1 \leq i \leq k\} \sqcup \{\alpha_2\}$ of cardinality 2k+1. As in the previous case, $\alpha_1 - \alpha_2 \in \Phi^+ - \Phi(w)$, and $\Phi(w)$ is closed under $\dot{+}$, so we may choose (not necessarily unique) sets of indices I_+ and I_- such that the following are true:

- 1. I_+ and I_- are disjoint, $|I_+ \sqcup I_-| = n m 1$ if $\alpha_2 \in \Phi^+ \Phi(w)$, and $|I_+ \sqcup I_-| = n m$ if $\alpha_2 \notin \Phi^+ \Phi(w)$.
- 2. For $i \in I_+$, we have $\{\alpha_1 + \eta_i, \alpha_2 + \eta_i\} \subset \Phi^+ \Phi(w)$.
- 3. For $i \in I_-$, we have $\{\alpha_1 \eta_i, \alpha_2 \eta_i\} \subset \Phi^+ \Phi(w)$.

As before, if $\alpha_2 \notin \Phi^+ - \Phi(w)$ we can define the subspace

$$\left(\bigoplus_{i\in I_+} \mathbb{R}e_{2i+2} \oplus \bigoplus_{i\in I_-} \mathbb{R}e_{2i+1}\right)^{\perp} \subset \mathbb{R}^{n+2}$$

of dimension m, and the harmonic representative e_w restricts nontrivially to the corresponding $\mathfrak{h}' = \mathfrak{so}(2,m)$ in \mathfrak{g} . If $\alpha_2 \in \Phi^+ - \Phi(w)$, then one adds on $\mathbb{R}e_n$ to the subspace and e_2 restricts nontrivially to the corresponding $\mathfrak{h}' = \mathfrak{so}(2,m)$. In either case, since \mathfrak{h}' is conjugate to \mathfrak{h} by $m \in M(\mathbb{C})$, we have proved the proposition. \Box

4.4. Proof of Proposition 4.1

For $E = \mathbb{C}$, the injectivity in degrees $i \leq c/2$ except in the exceptional case is immediate from the fact that \mathfrak{n}^* is the natural representation of the factor SO(d-1) of L, and so $\mathrm{H}^i(\mathfrak{n},\mathbb{C}) = \wedge^i \mathfrak{n}^*$ are irreducible. This can be easily generalized to the case of general E, but we give a computational proof using Kostant's theorem, as we will have to verify slightly more.

We may assume $\mathfrak{g} = \mathfrak{so}(1,d) = \mathfrak{so}(J)$, where

$$J = \begin{pmatrix} & & 1 \\ & I_{d-1} & \\ 1 & & \end{pmatrix}.$$

Fix a Cartan subalgebra \mathfrak{s} of $\mathfrak{so}(d-2)_{\mathbb{C}}$ and let $\mathfrak{t} \subset \mathfrak{g}_{\mathbb{C}}$ be the Cartan subalgebra defined by

$$\mathfrak{t} := \{ diag(a, C, -a) : a, d \in \mathbb{C}, C \in \mathfrak{s} \}.$$

Then \mathfrak{t} is defined and maximally split over \mathbb{R} , and the subspace $\mathfrak{a}_{\mathbb{C}} \subset \mathfrak{t}$ given by C = 0 is the complexification of the Lie algebra $\mathfrak{a} \subset \mathfrak{g}$ of a maximal \mathbb{R} -split subspace in \mathfrak{g} . Let

1680

 $\alpha \in \mathfrak{t}^*$ be defined by

$$\alpha(diag(a, C, -a)) = a. \tag{4.7}$$

We choose for \mathfrak{s} the same Cartan subalgebra of block-diagonal matrices in $\mathfrak{so}(d-1)_{\mathbb{C}}$ specified earlier in equations (4.4) and (4.5) and use the same roots η_i and the same positive system used there. An explicit computation shows that the positive roots are

$$\Phi^+(\mathfrak{t},\mathfrak{g}_{\mathbb{C}}) = \Phi^+(\mathfrak{s},\mathfrak{so}(d-1)_{\mathbb{C}}) \sqcup \{\alpha \pm \eta_i : 1 \le i \le k\}$$

for d-1=2k even and

$$\Phi^+(\mathfrak{t},\mathfrak{g}_{\mathbb{C}}) = \Phi^+(\mathfrak{s},\mathfrak{so}(d-1)_{\mathbb{C}}) \sqcup \{\alpha \pm \eta_i : 1 \le i \le k\} \sqcup \{\alpha\}$$

for d-1 = 2k+1 odd. We also have

$$\Phi(\mathfrak{n}) = \{ \alpha \pm \eta_i : 1 \le i \le k \}$$

in the case d-1=2k even and

$$\Phi(\mathfrak{n}) = \{\alpha \pm \eta_i : 1 \le i \le k\} \sqcup \{\alpha\}$$

in the case d-1 = 2k+1 odd.

We will list the relevant $w \in W^P$ and the sets $\Phi(w)$. We will consider the even and odd cases separately.

First assume d-1 = 2k is even. Let $\alpha_1, \alpha_2, \ldots, \alpha_{k+1}$ be the set of simple roots of $\mathfrak{so}(1,d)_{\mathbb{C}}$ determined by the positive system already fixed – that is, $\alpha_1 := \alpha - \eta_1$, $\alpha_i = \eta_{i-1} - \eta_i$ for $2 \le i \le k$, and $\alpha_{k+1} = \eta_{k-1} + \eta_k$. The minimal length representatives in W^P of length $\le k$ are

$$\{s_0, s_1, \ldots, s_k, t_k\},\$$

where $s_0 := 1$, $s_j := s_{\alpha_1} \cdots s_{\alpha_j}$ for $1 \le j \le k$ has length j, and $t_k = s_{k-1}s_{\alpha_{k+1}}$ has length k (compare [17, Section VI.3.1]; as usual, s_{α_i} denotes the reflection in α_i). The set $\Phi(w)$ is easily computed for these representatives: $\Phi(1) = \emptyset$ and $\Phi(s_j) =$ $\{\alpha_1, \alpha_1 + \alpha_2, \dots, \alpha_1 + \dots + \alpha_j\} = \{\alpha - \eta_1, \alpha_1 - \eta_2, \dots, \alpha - \eta_j\}$ for $j \le k$, whereas $\Phi(t_k) =$ $\{\alpha_1, \dots, \alpha_1 + \dots + \alpha_{k-1}, \alpha_1 + \dots + \alpha_{k-1} + \alpha_{k+1}\} = \{\alpha - \eta_1, \alpha_1 - \eta_2, \dots, \alpha - \eta_{k-1}, \alpha + \eta_k\}.$

Now suppose d-1 = 2k+1 is odd. Let $\alpha_1, \alpha_2, \ldots, \alpha_{k+1}$ be the simple roots of $\mathfrak{so}(1,d)_{\mathbb{C}}$ determined by the positive system – that is, $\alpha_1 := \alpha - \eta_1$, $\alpha_i = \eta_{i-1} - \eta_i$ for $2 \le i \le k$, and $\alpha_{k+1} = \eta_k$. The minimal length representatives in W^P of length $\le k$ are

$$\{s_0, s_1, \ldots, s_k\},\$$

where $s_0 := 1$, and $s_j := s_{\alpha_1} \cdots s_{\alpha_j}$ for $1 \le j \le k$ has length j (compare [17, Section VI.4.4], where this set is denoted PW). The set $\Phi(s_j)$ has the same description for $j \le k$ as before.

Now consider the setup of the proposition. We have $\mathfrak{h} = \mathfrak{so}(1,c)$ for $c \leq d-1$, embedded in the standard way – that is, using the subspace $\mathbb{R}^{c+1} \subset \mathbb{R}^{d+1}$ spanned by $e_1, e_2, \ldots, e_c, e_{d+1}$. To show that $\operatorname{Res}_{\mathfrak{n}}$ is injective in a given degree *i*, it will suffice to show that the harmonic representative (i.e., *L*-highest weight vector) $e_w \otimes v_{w\lambda}$ restricts nontrivially in $\operatorname{H}^i(\mathfrak{n}_H, E_H)$ for each $w \in W^P$ of length *i*. In the case at hand, for j < c/2 there is a unique element in W^P of length *j*, namely s_j . The *L*-highest weight vector $e_{s_j} \otimes v_{s_j\lambda} =$

 $\bigwedge_{1 \leq i \leq j} e_{-(\alpha - \eta_i)} \otimes v_{s_j\lambda} \text{ maps in } \wedge^j \mathfrak{n}_H^* \otimes E_H \text{ to (a nonzero multiple of) the harmonic representative } \bigwedge_{1 \leq i \leq j} e_{-(\alpha - \eta_i)}^H \otimes v_{s_j^H\lambda_H}^H, \text{ where } v_{s_j^H\lambda_H}^H \text{ is the } s_j^H(\lambda_H) \text{-weight vector of } E_H, \text{ and hence is nonvanishing in cohomology. This proves that } \operatorname{Res}_{\mathfrak{n}}(e_w \otimes v_{w\lambda}) \neq 0, \text{ and hence that } \operatorname{Res}_{\mathfrak{n}} \text{ is injective for } j < c/2. \text{ The same proof works if } j \leq c/2, \text{ as long as we are not in the exceptional case } (d,c,i) = (2k+1,2k,k).$

In the remaining case we are considering $\operatorname{Res}_{\mathfrak{n}}$ on $\operatorname{H}^{k}(\mathfrak{n}, E)$ for d = 2k + 1, c = 2k, i = k. In this case there are two *L*-irreducible summands, with highest weight vectors $e_{s_k} \otimes v_{s_k\lambda}$ and $e_{t_k} \otimes v_{t_k\lambda}$, respectively. In the embedding $\mathfrak{so}(2k+1,\mathbb{C}) \subset \mathfrak{so}(2k+2,\mathbb{C})$ the weight space $\mathfrak{h}^{-\alpha_k}$ is embedded diagonally in the weight spaces $\mathfrak{g}^{-\alpha_k}$ and $\mathfrak{g}^{-\alpha_{k+1}}$. Under the restriction from *T* to T_H we have $\alpha_k|_{T_H} = \alpha_{k+1}|_{T_H} = \eta_{k-1}^H = \alpha_k^H$. Thus the vector $e_{s_k} \otimes v_{s_k\lambda} =$ $\bigwedge_{1 \leq i \leq k} e_{-(\alpha - \eta_i)} \otimes v_{s_k\lambda}$ goes to (a nonzero multiple of) the vector $\bigwedge_{1 \leq i \leq k-1} e_{-(\alpha - \eta_i)}^H \wedge e_{-\eta_{k-1}}^H \otimes v_{s_k\lambda}^H$, which is nonzero since η_{k-1} is not one of $\alpha - \eta_i$, $i \leq k-1$. A similar argument applies to $e_{t_k} \otimes v_{t_k\lambda}$.

4.5. Proof of Proposition 4.2

This was proved for $E = \mathbb{C}$ in [36, §1.6], and the elements of W^P are explicitly listed there. The proof extends to general coefficients exactly as in the previous proof.

4.6. Proof of Proposition 4.4

A tedious computational proof is possible, but we will argue differently. Recall the notation P = MW for the parabolic in SU(1,n) and $P_H = P \cap H = LN$ for the parabolic in SO(1,n). For k < n we have a diagram

Here, as in §2.2, $\mathfrak{w}, \mathfrak{v}, \mathfrak{u}$ are the Lie algebras of the real points of $W(\mathbb{R}), V(\mathbb{R}), U(\mathbb{R})$. The second vertical map is induced by $\mathfrak{n} \subset \mathfrak{w}$. The first vertical map is induced by the identification of \mathfrak{v} with the χ -eigenspace for A in \mathfrak{w} ; since $A \subset P_H$ and it acts by χ on \mathfrak{u} , we have $\mathfrak{n} \subset \mathfrak{v}$ (compare §2.2 for notation). The top row comes from the long exact sequence for the boundary divisor in the toroidal compactification, and is exact in degrees k < n and is a sequence of Hodge structures (see [36, Lemma 1.3]). The bigrading on $\wedge^k \mathfrak{v}^*_{\mathbb{C}}$ comes from $\mathfrak{v}^*_{\mathbb{C}} = (\mathfrak{v}^*_{\mathbb{C}})^{1,0} + (\mathfrak{v}^*_{\mathbb{C}})^{0,1}$, given by the complex structure on $\mathfrak{v} = \mathfrak{w}/\mathfrak{u}$ (itself given by the central $U(1) \subset M(\mathbb{R})$), and the (k, 0)-subspace $\wedge^k (\mathfrak{v}^*_{\mathbb{C}})^{1,0}$ maps isomorphically onto $H^{k,0}(\mathfrak{w}, \mathbb{C})$. This is because the first term in the sequence has Hodge types $(k-1,1), \ldots, (1,k-1)$, because \mathfrak{u}^* amounts to a Tate twist (see the proof of [36, Lemma 1.3]). Now the composition $(\mathfrak{v}^*_{\mathbb{C}})^{1,0} \hookrightarrow \mathfrak{v}^*_{\mathbb{C}} \to \mathfrak{n}^*_{\mathbb{C}}$ is an isomorphism, since the kernel of $\mathfrak{v}^*_{\mathbb{C}} \to \mathfrak{n}^*_{\mathbb{C}}$ is nonzero, and since $H^{k,0}(\mathfrak{w}, \mathbb{C})$ is M-irreducible (see [36, Remark 1.11]), Res_{\mathfrak{w}} is injective on $H^{k,0}(\mathfrak{w}_{\mathbb{C}}, \mathbb{C})$ for k < n.

5. Congruence real hyperbolic manifolds

We restate the main result (Theorem 1.1) for congruence hyperbolic manifolds:

Theorem 5.1. Suppose that $H \subset G$ are semisimple groups of the same \mathbb{Q} -rank such that $H(\mathbb{R})^{nc} \subset G(\mathbb{R})^{nc}$ is $SO(1,c) \subset SO(1,d)$ for $2 \leq c < d$, and that neither H nor G is triality. Then Res : $\mathrm{H}^{i}(\mathcal{M}_{G}, E) \to I_{H}^{G}\mathrm{H}^{i}(\mathcal{M}_{H}, E_{H})$ is injective for $i \leq c/2$.

In the compact case (and $E = \mathbb{C}$), this is [12, Theorem 1.5] of Bergeron and Clozel, who, using the Burger–Sarnak method [18] following Harris and Li [28], deduced it from Arthur's endoscopic classification [2] of automorphic representations for orthogonal groups. Their arguments can be adapted to the noncompact case to prove injectivity on the interior cohomology $\mathrm{H}^{i}_{!}(\mathscr{M}, E)$ for $i \leq c/2$. Since the state of the literature on this adaptation to the noncompact case is less than satisfactory, we will sketch the argument in some detail, although the ingredients are all well known. Combined with an elementary argument at infinity using Proposition 4.1, this proves the theorem in general. Before starting the proofs, we will need to recall some general facts.

Recall the classification of unitarizable (\mathfrak{g}, K) -modules with cohomology with coefficients in E for $G(\mathbb{R}) = SO(d, 1)$ from [17, Section VI.4] or [41, Section 1.3]. (We refer to [17] for more details; this reference deals with $SO_0(d, 1)$ and $E = \mathbb{C}$, but it is easy to extend to the general case using, for example, translation functors as in [17, Section VI.0].) Let $0 \leq i_E \leq \lfloor d/2 \rfloor$ be the minimal degree for which there exists a unitary representation V with $H^*(\mathfrak{g}, K, V \otimes E) \neq \{0\}$ (so $i_{\mathbb{C}} = 0$ and $i_E = \lfloor d/2 \rfloor$ if λ is regular). For each degree $i_E \leq i \leq \lfloor d/2 \rfloor$, there is a unique unitary cohomology in exactly degrees i and d-i if d is odd or i < d/2, and if d = 2k it has cohomology in degree k. In the case d = 2k, the representation π_k of SO(1,d) is a discrete series representation and its restriction to $SO_0(1,d)$ is a sum of two discrete series representations. If d = 2k + 1 is odd, the representation π_k is tempered. This completes the list of unitarizable (\mathfrak{g}, K) -modules with cohomology with coefficients in E. (All these depend on E, but to keep the notation simple we do not indicate this.) When we use these objects for $H(\mathbb{R})$, we will write π_i^H .

We recall some well-known facts about noncompact arithmetic quotients, for which we refer to Appendix A. There is a decomposition $L^2(\Gamma \setminus G(\mathbb{R})) = L^2_{dis}(\Gamma \setminus G(\mathbb{R})) \oplus$ $L^2_{cts}(\Gamma \setminus G(\mathbb{R}))$ into discrete and continuous spectra and a further decomposition $L^2_{dis}(\Gamma \setminus G(\mathbb{R})) = L^2_{cusp}(\Gamma \setminus G(\mathbb{R})) \oplus L^2_{res}(\Gamma \setminus G(\mathbb{R}))$ into cuspidal and residual spectra. For $? \in \{cusp, dis, cts\}$, let

$$\mathrm{H}_{?}^{*}(\Gamma, E) = \mathrm{H}^{*}\left(\mathfrak{g}, K, L_{?}^{2}\left(\Gamma \setminus G(\mathbb{R})\right) \otimes E\right),$$

where the (\mathfrak{g}, K) -cohomology of a unitary $G(\mathbb{R})$ -representation (π, V) is understood to be that of the space V^{∞} of smooth vectors. The natural map

$$\mathrm{H}^*\left(\mathfrak{g}, K, L^2\left(\Gamma \backslash G(\mathbb{R})\right) \otimes E\right) \longrightarrow \mathrm{H}^*(\Gamma, E)$$

induced by $L^2(\Gamma \setminus G(\mathbb{R}))^{\infty} \subset C^{\infty}(\Gamma \setminus G(\mathbb{R}))$ is injective on $\mathrm{H}^*_{cusp}(\Gamma, E)$ and zero on $\mathrm{H}^*_{cts}(\Gamma, E)$ (see Appendix A for a proof of this well-known fact).

Lemma 5.2. For Γ arithmetic in SO(1,d), we have the following:

- (1) $\operatorname{H}^*_{cusp}(\Gamma, E) = \operatorname{H}^*_!(\Gamma, E).$
- (2) $\operatorname{H}^{i}_{dis}(\Gamma, E) \to \operatorname{H}^{i}(\Gamma, E)$ is injective in degrees $i \leq d/2$ and an isomorphism for $i \leq \lfloor d/2 \rfloor 1$.

Proof. This follows from methods of Harder [26] recalled in Appendix B and also Rohlfs and Speh [41]. According to [41, Theorem 1.5.1], the cohomology in degrees $i < k = \lfloor d/2 \rfloor$ is all square-integrable, and by the results in [41, Section 1.4], the noncuspidal squareintegrable classes are generated using residues of Eisenstein series. These classes restrict nontrivially to the boundary (see the proof of [41, Proposition 1.4.4] or Lemma B.1), so they do not belong to interior cohomology. This proves (1) of the lemma in degrees $i < k = \lfloor d/2 \rfloor$. Moreover, the restriction to the boundary is injective on the residual classes, proving (2) in degrees $i < k = \lfloor d/2 \rfloor$. Statement (1) follows in degrees $i > d - \lfloor d/2 \rfloor$ by duality. This leaves degree k when d = 2k is even and degrees k, k + 1 when d = 2k + 1 is odd. In both cases, the only contributions to $H_1^*(\Gamma, E)$ (or to $H_{dis}^i(\Gamma, E)$) in these degrees are from tempered representations (namely the discrete series when d = 2k is even and the tempered representation π_k when d = 2k + 1 is odd), so they are cuspidal by a well-known observation of Wallach [50], and then (1) and (2) are clear for these contributions.

Proposition 5.3. Assume that neither G nor H is triality. Then Res is injective on $H_i^i(\mathcal{M}, E)$ for $i \leq c/2$.

Proof. We sketch how to adapt the argument of [7, 8, 12, 28] to the noncompact case. The main points are as follows:

(1) For $i \leq c/2$, the abstract restriction of π_i to (\mathfrak{h}, K_H) contains π_i^H as a direct summand and multiplicity 1 holds – that is, dim Hom $_{(\mathfrak{h}, K_H)}(\pi_i|_H, \pi_i^H) = 1$. Moreover, the induced map

$$\mathrm{H}^{i}(\mathfrak{g}, K, \pi_{i} \otimes E) \to \mathrm{H}^{i}(\mathfrak{h}, K_{H}, \pi_{i}^{H} \otimes E_{H})$$

$$(5.1)$$

is an isomorphism of one-dimensional spaces. This is proved in [7, Theorem 3.4] (see also [28, §1,§6] and [8, Théorème 5.3]). The references treat the case $E = \mathbb{C}$ but the proof works in general; alternatively, one can use translation functors as in [17, Section VI.0] to reduce the general case to this one.

(2) Let $R: C^{\infty}(\Gamma \setminus G(\mathbb{R})) \to C^{\infty}(\Gamma_H \setminus H(\mathbb{R}))$ denote restriction of functions and $R^*: H^*(\Gamma, E) \to H^*(\Gamma_H, E_H)$ the induced map in cohomology. Given an irreducible summand π of $L^2_{cusp}(\Gamma \setminus G(\mathbb{R}))$ with smooth vectors $\pi^{\infty} = \pi_i$, the image $R(\pi_i)$ consists of bounded functions, so we may consider its closure $R(\pi)$ in $L^2(\Gamma_H \setminus H(\mathbb{R}))$. Suppose that an irreducible summand σ of $L^2(\Gamma_H \setminus H(\mathbb{R}))$ with $\sigma^{\infty} = \pi_i^H$ appears as a direct summand of $R(\pi)$. We claim that $R^*: H^i(\Gamma, E) \to H^i(\Gamma_H, E_H)$ is injective on the summand $H^i(\mathfrak{g}, K, \pi \otimes E)$. To see this, note that $R(\pi) = R(\pi)_{dis} \oplus R(\pi)_{cts}$, where $R(\pi)_? = R(\pi) \cap L^2_?(\Gamma \setminus G(\mathbb{R}))$, for ? = dis, cts, and the map R^* on $H^i(\mathfrak{g}, K, \pi \otimes E)$ is induced by the composition

$$\pi^{\infty} \to R(\pi)^{\infty} \to R(\pi)^{\infty}_{dis} \to C^{\infty}\left(\Gamma_H \setminus H(\mathbb{R})\right).$$

(We use the fact that $\mathrm{H}^*(\mathfrak{h}, K_H, R(\pi)_{cts} \otimes E_H) \to \mathrm{H}^*(\Gamma_H, E_H)$ is zero since $R(\pi)_{cts}$ is a summand of $L^2_{cts}(\Gamma_H \setminus H(\mathbb{R}))$; see Appendix A for a proof of this fact.) The last map

1685

induces an injection in degree $i \leq c/2$ (by Lemma 5.2(2)). The map $\pi^{\infty} \to R(\pi)_{dis}^{\infty}$ is nonzero in cohomology because the composition $\pi^{\infty} \to R(\pi)_{dis}^{\infty} \to \sigma^{\infty}$ is a nonzero multiple of the map $\pi_i|_H \to \pi_i^H$ in (1) (by multiplicity 1), and hence induces the nontrivial map (5.1) in cohomology. This proves that R^* is nonzero on $H^*(\mathfrak{g}, K, \pi \otimes E)$, and hence injective.

(3) The Burger–Sarnak argument shows that given π as in (2), and assuming a certain isolation hypothesis on π_i^H (recalled later), we can arrange for a summand σ as in (2), perhaps after replacing H by a conjugate – that is, replacing the map R by $R_g : C^{\infty}(\Gamma \setminus G(\mathbb{R})) \to C^{\infty}(g\Gamma g^{-1} \cap H \setminus H(\mathbb{R}))$ for some $g \in G(\mathbb{Q})$. This is [28, Proposition 3.1] and the refinement [7, Proposition 3.2]. The observation of [18] is that matrix coefficients of the cuspidal representation on $\Gamma \setminus G(\mathbb{R})$ are, when restricted to $H(\mathbb{R})$, the limits, uniform on compacta, of finite sums of matrix coefficients of $H(\mathbb{R})$ appearing in the spaces $L^2(g^{-1}\Gamma g \cap H \setminus H(\mathbb{R}))$ for $g \in G(\mathbb{Q})$. A key point explained in [28] (see the remarks at the bottom of their p. 93) is that this applies to cuspidal cohomology on a noncompact quotient, because cuspidal functions are of rapid decrease, hence uniformly continuous on $\Gamma \setminus G(\mathbb{R})$, and this suffices for the argument in [18]. Thus under the isolation hypothesis, if π_i^H is weakly contained in $R(\pi)$, then there is a direct summand σ of $R_g(\pi)$ as in (2), and so by (2), R_g^* is injective on $H^i(\mathfrak{g}, K, \pi \otimes E)$.

(4) The isolation hypothesis required in (3) is that π_i^H is isolated in $\{\pi_i^H\} \cup \{(\rho, V_\rho) \in \widehat{H}_{Aut} : d \equiv 0 \text{ on } C^i(\mathfrak{g}, K, V_\rho \otimes E)\}$. (We refer to [7, 28] for the precise definitions of \widehat{H}_{Aut} and the relevance of this condition.) This was shown in [7] to follow from a uniform (in Γ) lower bound for the first nonzero eigenvalue of the Laplacian on *i*-forms on M_{Γ} [7]. This eigenvalue bound was later proved in [12, Theorem 1.3] using Arthur's endoscopic classification [2] of representations. (At this point triality forms must be excluded, although they do not occur if the Q-rank is 1; see Remark 5.5.)

(5) Putting (1)–(4) together, we conclude that Res is injective on the cuspidal cohomology for $i \leq c/2$, and hence, by Lemma 5.2, on $\mathrm{H}^{i}_{!}(\mathscr{M}, E)$ for $i \leq c/2$.

We will make some remarks as to the necessity of the contortions in the previous proof after finishing the proof of the theorem.

Proof of Theorem 5.1. We will use the standard commutative diagram

$$0 \longrightarrow \mathrm{H}^{i}_{!}(\mathscr{M}, E) \longrightarrow \mathrm{H}^{i}(\mathscr{M}, E) \longrightarrow I^{G}_{P}\mathrm{H}^{i}(\mathfrak{n}, E)$$

$$\downarrow^{\mathrm{Res}} \qquad \qquad \downarrow^{\mathrm{Res}} \qquad \qquad \downarrow^{\mathrm{Res}_{\infty}}$$

$$0 \longrightarrow I^{G}_{H}\mathrm{H}^{i}_{!}(\mathscr{M}_{H}, E) \longrightarrow I^{G}_{H}\mathrm{H}^{i}(\mathscr{M}_{H}, E) \longrightarrow I^{G}_{H}I^{H}_{P_{H}}\mathrm{H}^{i}(\mathfrak{n}_{H}, E_{H}).$$

$$(5.2)$$

This diagram comes from the properness of $M_{H,\Gamma_H} \to M_{\Gamma}$, or can be seen using the fact that the cusp (i.e., reductive Borel–Serre) compactification is functorial for $H \subset G$ in this case. The identification of the boundary cohomology with $I_P^G \mathrm{H}^i(\mathfrak{n}, E)$ is standard. It suffices to prove the injectivity of Res_{∞} for $i \leq c/2$. By the transitivity of induction, $I_H^G I_{P_H}^H = I_P^G I_{P_H}^P$, so it suffices to prove the injectivity of $\mathrm{H}^i(\mathfrak{n}, E) \to I_{P_H}^P \mathrm{H}^i(\mathfrak{n}_H, E_H)$. Now the action of $N(\mathbb{Q})$ on $\mathrm{H}^i(\mathfrak{n}, E)$ is trivial, so this factors through $(I_{P_H}^P \mathrm{H}^i(\mathfrak{n}_H, E_H))^{N(\mathbb{Q})} = I_{L_H}^L \mathrm{H}^i(\mathfrak{n}_H, E_H)$, and it is enough to prove the injectivity of $\mathrm{H}^i(\mathfrak{n}, E) \to I_{L_H}^L \mathrm{H}^i(\mathfrak{n}_H, E)$. This follows from the injectivity of $\mathrm{Res}_{\mathfrak{n}}$ in degrees $i \leq c$ proved in Proposition 4.1, except possibly in the case (d, c, i) = (2k + 1, 2k, k).

The remaining case is restriction from SO(1,2k+1) to SO(1,2k) in degree i = k. It suffices to prove that $\operatorname{Res}_{\infty}$ is injective on the image of $\operatorname{H}^{k}(\mathcal{M}, E) \to I_{P}^{G}\operatorname{H}^{k}(\mathfrak{n}, E)$. This map is induced by the $P(\mathbb{Q})$ -map $\mathrm{H}^{k}(\mathcal{M}, E) \to \mathrm{H}^{k}(\mathfrak{n}, E)$ given by restriction to a deleted neighborhood of the cusp given by P = LN, and the image of $\mathrm{H}^{k}(\mathcal{M}, E) \to I_{E}^{G}\mathrm{H}^{k}(\mathfrak{n}, E)$ is $I_P^G U$, where U is the image of $H^k(\mathcal{M}) \to H^k(\mathfrak{n}, E)$. There is a nondegenerate duality pairing on $\mathrm{H}^{k}(\mathfrak{n}, E)$ given by the cup product and the self-duality of E. Now U is a maximal isotropic subspace for the duality pairing on $\mathrm{H}^{k}(\mathfrak{n}, E)$ – that is, $U^{\perp} = U$ – and it is also $L(\mathbb{C}) = \mathbb{C}^* \times SO(2k,\mathbb{C})$ -stable. Since $\mathrm{H}^k(\mathfrak{n}, E) = U_+ \oplus U_-$ is a sum of two inequivalent $SO(2k,\mathbb{C})$ -modules of the same dimension (they form a single $O(2k,\mathbb{C})$ module), either $U = U_+$ or $U = U_-$, and the image $I_P^G U$ of $\mathrm{H}^k(\mathcal{M}) \to I_P^G \mathrm{H}^k(\mathfrak{n}, E)$ is either $I_P^G U_+$ or $I_P^G U_-$. In the notation of §4.4, U_+ and U_- are the $SO(2k,\mathbb{C})$ -modules with highest weight vectors $e_{s_k} \otimes v_{s_k\lambda}$ and $e_{t_k} \otimes v_{t_k\lambda}$, respectively. Now the restriction $\operatorname{Res}_{\mathfrak{n}}: \operatorname{H}^{k}(\mathfrak{n}, E) \to \prod_{m} \operatorname{H}^{k}(\mathfrak{n}_{H}, E_{H})$ induced by $\mathfrak{n}_{H} \subset \mathfrak{n}$ is nonzero on either summand, because it is nonzero on these highest weight vectors, as was proved in Proposition 4.1. It follows that $I_P^G \mathrm{H}^k(\mathfrak{n}, E) \to I_P^G I_{P_H}^P \mathrm{H}^k(\mathfrak{n}_H, E_H)$ is injective on each of $I_P^G U_{\pm}$ individually, and hence on the image of $\mathrm{H}^{k}(\mathcal{M}, E) \to I_{P}^{G}\mathrm{H}^{k}(\mathfrak{n}, E)$, whichever of these modules it is. This completes the proof of the theorem.

Remark 5.4 (on the Burger–Sarnak method for noncompact quotients). In general, it is not clear to us that the argument of [7, 8, 12, 28] can be adapted to treat noncuspidal interior cohomology classes on a general arithmetic quotient without a better understanding of the latter, for example, using Eisenstein series. Since the argument for injectivity in cohomology treats one summand π of L^2 at a time, one needs to know that π^{∞} , or at least π , contains some uniformly continuous functions, the diagonal matrix coefficients of which can then be used in the Burger–Sarnak argument. If π is cuspidal, then the functions in π^{∞} are of rapid decrease, hence uniformly continuous, and this suffices. It is not clear to us that the automorphic representatives of noncuspidal interior cohomology classes are uniformly continuous on $\Gamma \setminus G(\mathbb{R})$ – they are not of rapid decrease, as they would then be cuspidal – or, indeed, that the summand π^{∞} contributing to such cohomology contains any uniformly continuous functions.

In the case at hand, Lemma 5.2 shows that there is no noncuspidal interior cohomology, and so this problem does not occur. For the congruence ball quotients discussed in [36], [13, §3], and the next section one, can show that the noncuspidal interior cohomology consists of nonprimitive classes (see the discussion in the next section), and the analogue of Proposition 5.3 for $SU(1,m) \subset SU(1,n)$ can be proved similarly. However, for SO(2,n)the situation is more complicated, and something more is required. In any case, we will not use automorphic arguments to treat interior cohomology in the cases SU(1,n) and SO(2,n), since the geometric arguments of §3 are available.

Remark 5.5. A triality form over a totally real field F which becomes SO(1,7) over \mathbb{R} for some real embedding of F is necessarily anisotropic over \mathbb{Q} . This follows by looking

at Tits indices (see the table on [46]; for details, see [14, p. 337f]). So in the Q-rank 1 case of the theorem, we may ignore triality altogether.

6. Congruence complex hyperbolic manifolds

The first main result for congruence complex hyperbolic manifolds is the following:

Theorem 6.1. Suppose that $H \subset G$ are groups of the same \mathbb{Q} -rank and $H(\mathbb{R})^{nc} \subset G(\mathbb{R})^{nc}$ is the inclusion $SU(1,m) \subset SU(1,n)$ with $2 \leq m < n$. Then Res is injective on $\mathrm{H}^{i}_{!}(\mathcal{M}_{G})$ for $i \leq m$ and on $\mathrm{H}^{i}(\mathcal{M}_{G})$ for i < m.

This follows immediately from Corollary 3.2 and Proposition 4.2. This is simpler than the proof in [36].

The rest of this section consists of the proof of the following, which is Theorem 1.2:

Theorem 6.2. Suppose that $H \subset G$ are groups of the same \mathbb{Q} -rank and $H(\mathbb{R})^{nc} \subset G(\mathbb{R})^{nc}$ is the inclusion $SO(1,n) \subset SU(1,n)$ with n > 2. Then Res is injective on $\mathrm{H}^{i,0}(\mathcal{M}_G)$ for $i \leq n/2$.

Proof. The proof is broadly the same as that of Theorem 5.1: Given the injectivity on $\mathrm{H}_{!}^{i,0}(\mathscr{M})$ for $i \leq n/2$, the analogue for this situation of diagram (5.2), the strictness of the Hodge filtration, and Proposition 4.4 combine to prove the theorem.

The proof of injectivity on $H_!^{i,0}(\mathcal{M})$ for $i \leq n/2$ follows the outline of the proof of Proposition 5.3, with step (1) there replaced by the following:

(1') For each i < n, there is a unique cohomological (\mathfrak{g}, K) -module $J_{i,0}$ with cohomology in bidegree (i,0). This is immediate from the classification of (\mathfrak{g},K) -modules with cohomology for SU(1,n) [17, Sections VI.4.7–VI.4.12]. For $i \le n/2$, the abstract restriction of $J_{i,0}$ to (\mathfrak{h}, K_H) contains π_i^H (the unique cohomological representation for SO(1,n)with $\mathrm{H}^i(\mathfrak{h}, K_H, \pi_i^H) \ne 0$; see §5) as a direct summand with multiplicity 1 – that is, $\dim \mathrm{Hom}_{(\mathfrak{h}, K_H)}(J_{i,0}|_H, \pi_i^H) = 1$ – and the induced map

$$\mathrm{H}^{i}(\mathfrak{g}, K, J_{i,0}) \longrightarrow \mathrm{H}^{i}(\mathfrak{h}, K_{H}, \pi_{i}^{H})$$

is an isomorphism of one-dimensional spaces. This is [8, Théorème 5.6].

Given (1'), steps (2)-(5) in the proof of Proposition 5.3 work verbatim to prove that Res is injective on $\mathrm{H}^{i,0}_{cusp}(\mathscr{M})$ for $i \leq n/2$. We are using here the agreement of the two possible Hodge structures on $\mathrm{H}^{i}_{!}(\mathscr{M})$, the first coming from geometry (hence having good properties for the boundary exact sequence and in diagram (5.2)) and the second from the inclusion $\mathrm{H}^{i}_{!}(M_{\Gamma}) \subset \mathrm{IH}^{i}(M_{\Gamma}^{*})$ and the L^{2} Hodge theory on the latter coming from equation (3.12) (hence agreeing with the Hodge types in (\mathfrak{g}, K) -cohomology). This agreement is known by [52] because M_{Γ}^{*} has isolated singularities.

Lemma 6.3 completes the proof of the theorem.

It remains to prove the following:

Lemma 6.3. For Γ arithmetic in SU(1,n), we have $\mathrm{H}^{*,0}_{!}(M_{\Gamma}) = \mathrm{H}^{*,0}_{cusp}(M_{\Gamma})$.

Proof. The proof is similar to that of Lemma 5.2 for SO(1,n). Suppose first that i < n. Since $\operatorname{IH}^{i}(M_{\Gamma}^{*}) = \operatorname{H}^{i}(M_{\Gamma})$ for i < n, we are actually dealing with the cohomology

$$\mathrm{H}^{i}\left(\mathfrak{g},K,L_{dis}^{2}\left(\Gamma\backslash G(\mathbb{R})\right)\right)=\bigoplus_{\pi\subset L_{dis}^{2}}\mathrm{H}^{i}(\mathfrak{g},K,\pi)$$

where the sum is over irreducible closed summands, only finitely many of which make a nonzero contribution to the sum. For π to contribute to the (i,0) summand of cohomology, we must have $\pi^{\infty} = J_{i,0}$. For such a summand π , the theory of Eisenstein series gives a mapping $I_{i,0} \to L^2_{dis}$ onto $\pi^{\infty} = J_{i,0}$. Here $I_{i,0}$ denotes the standard module of which $J_{i,0}$ is the Langlands quotient [17, Section VI.4.8]. Now the minimal degree in which $J_{i,0}$ has cohomology is i, so we are in the situation of Appendix B, and applying Lemma B.1 gives Lemma 6.3 for i < n. (Note that the assumption (*) required in Lemma B.1 holds, since $I_{i,0}$ has cohomology in degrees 2n - i, 2n - i - 1, and $I_{i,0} \to J_{i,0}$ induces an isomorphism in degree 2n - i; see [17, p. 133].)

The statement in degrees i > n follows by duality. Finally, the equality in degree n holds because the component at infinity of a class in $\mathrm{H}^{n,0}_{!}(M_{\Gamma})$ is of discrete series type, and hence is already cuspidal by [50].

Remark 6.4. The proof of the lemma shows more generally that for $i+j \leq n$, we have

$$\mathrm{H}^{i,j}_{!}(M_{\Gamma})^{prim} = \mathrm{H}^{i,j}_{cusp}(M_{\Gamma})^{prim},$$

where the primitive is taken with respect to an invariant Lefschetz class. Thus the Burger– Sarnak method can be applied to prove injectivity on $\mathrm{H}^{i,j}_{!}(M_{\Gamma})^{prim}$ for the restriction by $SU(1,m) \subset SU(1,n)$. Using the action of the Lefschetz operator, this can be used to give another proof of Theorem 6.1 for complex hyperbolic manifolds.

7. Orthogonal Shimura varieties

The main theorem in this case is the following:

Theorem 7.1. Suppose that $H \subset G$ are of the same \mathbb{Q} -rank and that $H(\mathbb{R})^{nc} \subset G(\mathbb{R})^{nc}$ is the inclusion $SO(2,m) \subset SO(2,n)$ with $n > m \ge 2$. Then Res is injective on $\mathrm{H}^{i}(\mathcal{M}_{G}, E)$ for $i \le m-1$.

We will argue as if the Q-ranks of both G and H are 2 and indicate how the argument simplifies in the Q-rank 1 case. The proof is by a kind of induction on the stratification of the minimal compactification, going from injectivity on interior cohomology $\mathrm{H}^{i}_{\mathrm{I}}(\mathcal{M}_{G})$ (proved earlier as Corollary 3.2) to injectivity on a larger subspace $\mathrm{Gr}^{W}_{i}\mathbb{H}^{i}_{c}(\mathcal{M}^{i}_{G}, j^{*}_{*}\mathbb{C})$ of $\mathrm{H}^{i}(\mathcal{M}_{G})$ (defined later), which takes into account some contributions from the onedimensional boundary strata, and then to the injectivity on all of $\mathrm{H}^{i}(\mathcal{M}_{G})$ by taking into account some contributions from the cusps (using the Lefschetz property for real hyperbolic manifolds from §5). A similar, but simpler, argument was used in [36] for ball quotients.

To simplify the notation somewhat, we will write \mathscr{M} for \mathscr{M}_G . (We continue to write \mathscr{M}_H for the Shimura variety associated to H, of course.)

1688

7.1. Some exact sequences related to the boundary

We will introduce some notation which will be useful. Recall the stratification of M_{Γ}^* discussed in §2, and denote the inclusions by

$$M_{\Gamma} \xrightarrow{j_{\Gamma}^{1}} M_{\Gamma}^{1} \xrightarrow{j_{\Gamma}^{0}} M_{\Gamma}^{*},$$

with $j_{\Gamma}^0 \circ j_{\Gamma}^1 = j_{\Gamma}$. This gives two cohomology long exact sequences:

1. The distinguished triangle associated with $j_{\Gamma^*}^1 \mathbb{C}$ on M_{Γ}^1 and the open-closed decomposition $M_{\Gamma} \subset \overset{j_{\Gamma}^1}{\longrightarrow} M_{\Gamma}^1 \overset{i_{\Gamma}^1}{\longleftarrow} Z_{\Gamma}^1$ is

$$j_{\Gamma !}^{1}\mathbb{C} \longrightarrow j_{\Gamma *}^{1}\mathbb{C} \longrightarrow i_{\Gamma *}^{1}i_{\Gamma }^{1*}j_{\Gamma *}^{1}\mathbb{C} \xrightarrow{+1}$$

and gives

$$\cdots \longrightarrow \mathrm{H}^{i}_{c}(M_{\Gamma}) \longrightarrow \mathbb{H}^{i}_{c}\left(M^{1}_{\Gamma}, j^{1}_{\Gamma*}\mathbb{C}\right) \longrightarrow \mathbb{H}^{i}_{c}\left(Z^{1}_{\Gamma}, i^{1*}_{\Gamma}j^{1}_{\Gamma*}\mathbb{C}\right) \longrightarrow \cdots$$

2. The distinguished triangle associated with $j_{\Gamma*}\mathbb{C}$ and $M^1_{\Gamma} \xrightarrow{j^0_{\Gamma}} M^*_{\Gamma} \xrightarrow{i^0_{\Gamma}} Z^0_{\Gamma}$ is

$$j^0_{\Gamma!} j^1_{\Gamma*} \mathbb{C} \longrightarrow j_{\Gamma*} \mathbb{C} \longrightarrow i^0_{\Gamma*} i^{0*}_{\Gamma} j^1_{\Gamma*} \mathbb{C} \xrightarrow{+1}$$

and gives

$$\cdots \longrightarrow \mathbb{H}^{i}_{c}\left(M^{1}_{\Gamma}, j^{1}_{\Gamma*}\mathbb{C}\right) \longrightarrow \mathrm{H}^{i}(M_{\Gamma}) \longrightarrow \mathbb{H}^{i}\left(Z^{0}_{\Gamma}, i^{0*}_{\Gamma}j_{\Gamma*}\mathbb{C}\right) \longrightarrow \cdots$$

Both are long exact sequences of mixed Hodge structures by [42].

These exact sequences are natural with respect to passing to subgroups of Γ of finite index, and this leads us to introduce the following suggestive notation:

$$\mathbb{H}_{c}^{i}\left(\mathscr{M}^{1}, j_{*}^{1}\mathbb{C}\right) := \operatorname{colim}_{\Gamma}\mathbb{H}_{c}^{i}\left(M_{\Gamma}^{1}, j_{\Gamma*}^{1}\mathbb{C}\right), \\
\mathbb{H}_{c}^{i}\left(\mathscr{Z}^{1}, i^{1*}j_{*}^{1}\mathbb{C}\right) := \operatorname{colim}_{\Gamma}\mathbb{H}_{c}^{i}\left(Z_{\Gamma}^{1}, i_{\Gamma}^{1*}j_{\Gamma*}^{1}\mathbb{C}\right), \\
\mathbb{H}^{i}\left(\mathscr{Z}^{0}, i^{0*}j_{*}\mathbb{C}\right) := \operatorname{colim}_{\Gamma}\mathbb{H}^{i}\left(Z_{\Gamma}^{0}, i_{\Gamma}^{0*}j_{\Gamma*}\mathbb{C}\right),$$
(7.1)

where all colimits are over congruence subgroups. These are smooth $G(\mathbb{Q})$ -modules, and the map $\mathrm{H}^*_c(\mathscr{M}) \to \mathrm{H}^*(\mathscr{M})$ factors through $\mathbb{H}^i_c(\mathscr{M}^1, j^1_*\mathbb{C}) \longrightarrow \mathrm{H}^i(\mathscr{M})$. The exact sequences give exact sequences

$$\cdots \longrightarrow \mathrm{H}^{i}_{c}(\mathscr{M}) \longrightarrow \mathbb{H}^{i}_{c}\left(\mathscr{M}^{1}, j_{*}^{1}\mathbb{C}\right) \longrightarrow \mathbb{H}^{i}_{c}\left(\mathscr{Z}^{1}, i^{1*}j_{*}^{1}\mathbb{C}\right) \longrightarrow \cdots$$
(7.2)

and

$$\cdots \longrightarrow \mathbb{H}^{i}_{c}\left(\mathscr{M}^{1}, j_{*}^{1}\mathbb{C}\right) \longrightarrow \mathbb{H}^{i}\left(\mathscr{M}\right) \longrightarrow \mathbb{H}^{i}\left(\mathscr{Z}^{0}, i^{0*}j_{*}\mathbb{C}\right) \longrightarrow \cdots,$$
(7.3)

which are exact sequences of (colimits of) mixed Hodge structures.

We also note the following useful consequence of the purity lemma (Lemma 3.5):

$$\mathbf{H}_{!}^{i}(\mathscr{M}) = \operatorname{im}\left(\operatorname{Gr}_{i}^{W}\mathbf{H}_{c}^{i}(\mathscr{M}) \to \operatorname{Gr}_{i}^{W}\mathbb{H}_{c}^{i}\left(\mathscr{M}^{1}, j_{*}^{1}\mathbb{C}\right)\right) \qquad \text{for} i \leq n-1.$$
(7.4)

Indeed, Lemma 3.5 implies that $\mathrm{H}_{!}^{i}(M_{\Gamma}) = \mathrm{im}\left(\mathrm{Gr}_{i}^{W}\mathrm{H}_{c}^{i}(M_{\Gamma}) \to \mathrm{Gr}_{i}^{W}\mathbb{H}_{c}^{i}\left(M_{\Gamma}^{1}, j_{*}^{1}\mathbb{C}\right)\right)$ for $i \leq n-1$, because $\mathrm{Gr}_{i}^{W}\mathbb{H}_{c}^{i}\left(M_{\Gamma}^{1}, j_{\Gamma}^{1}\mathbb{C}\right) \subset \mathrm{Gr}_{i}^{W}\mathrm{H}^{i}(M_{\Gamma})$ by the second exact sequence given. Since Gr_{i}^{W} commutes with the colimits, we get equation (7.4).

7.2. Proof of Theorem 7.1

Now consider the situation of $H \subset G$ and the morphism $M^*_{H,\Gamma_H} \to M^*_{\Gamma}$. The stratifications are compatible, in the sense that the stratification of M^*_{H,Γ_H} is the pullback of that of M^*_{Γ} – that is, the relevant diagrams relating strata are Cartesian. It follows that both the exact sequences are functorial – that is, there are $H(\mathbb{Q})$ -module maps from each exact sequence for G to the corresponding one for H. Frobenius reciprocity gives commutative diagrams of $G(\mathbb{Q})$ -modules with exact rows:

from the first sequence and the similar diagram with exact rows

from the second. Taking Gr^W_i and using equation (7.4) in the first diagram gives a commutative diagram

in which the upper row is exact for $i \leq n-1$ and the lower row is exact for $i \leq m-1$. (We have used the purity of $\mathrm{H}_{!}^{i}(\mathscr{M})$.) Similarly, taking Gr_{i}^{W} and using equation (7.4) in the second diagram gives a commutative diagram with exact rows for $i \leq m-1$:

We have used the purity of $\mathrm{H}^{i}(\mathscr{M})$ in degrees $\leq m-1$, which follows from the fact that $\mathrm{IH}^{i}(\mathcal{M}_{\Gamma}^{*}) = \mathrm{H}^{i}(\mathcal{M}_{\Gamma})$ in degrees $i \leq n-2$.

We see from these diagrams that Theorem 7.1, namely the injectivity of Res on $\mathrm{H}^{i}(\mathcal{M})$ in degrees $\leq m-1$, follows from the conjunction of Corollary 3.2 (injectivity on interior cohomology in degrees $\leq m-1$) and the following two statements:

Proposition 7.2. The map $\operatorname{Res}_{\infty}^{1}$ in diagram (7.5) is injective in degrees $i \leq m-1$.

Proposition 7.3. The map $\operatorname{Res}_{\infty}^{0}$ in diagram (7.6) is injective in degrees $i \leq m-1$.

The rest of this subsection will be taken up with the proofs of these two propositions. The first will use Proposition 4.3, and the second will use the Lefschetz property for real hyperbolic manifolds in Theorem 5.1.

Proof of Proposition 7.2. Recall that Proposition 7.2 asserts the injectivity of

$$\operatorname{Res}_{\infty}^{1}:\operatorname{Gr}_{i}^{W}\mathbb{H}_{c}^{i}\left(\mathscr{Z}^{1},i^{1*}j_{*}^{1}\mathbb{C}\right)\longrightarrow I_{H}^{G}\operatorname{Gr}_{i}^{W}\mathbb{H}_{c}^{i}\left(\mathscr{Z}_{H}^{1},i_{H}^{1*}j_{H*}^{1}\mathbb{C}\right)$$

in degrees $i \leq m-1$. We will deduce from Proposition 4.3 the a priori stronger assertion that the map

$$\mathbb{H}_{c}^{i}\left(\mathscr{Z}^{1}, i^{1*}j_{*}^{1}\mathbb{C}\right) \longrightarrow I_{H}^{G}\mathbb{H}_{c}^{i}\left(\mathscr{Z}_{H}^{1}, i_{H}^{1*}j_{H*}^{1}\mathbb{C}\right)$$
(7.7)

is injective in this range; since this is a morphism of mixed Hodge structures, the statement about the *i*th graded follows. By definition,

$$\mathbb{H}_{c}^{i}\left(\mathscr{Z}^{1}, i^{1*} j_{*}^{1} \mathbb{C}\right) = \operatorname{colim}_{\Gamma} \mathbb{H}_{c}^{i}\left(Z_{\Gamma}^{1}, i_{\Gamma}^{1*} j_{\Gamma*}^{1} \mathbb{C}\right).$$

Choose a rational boundary component F of dimension 1 and let P = MW be its stabilizer, which is the maximal parabolic stabilizing an isotropic plane in V. The stratum of M_{Γ}^* given by F is $S_{\Gamma} := \Gamma_{M_h} \setminus F$, and it is a component of Z_{Γ}^1 ; let $i_{S_{\Gamma}} : S_{\Gamma} \hookrightarrow M_{\Gamma}^1$ be the inclusion. Then we have natural identifications

$$\mathbb{H}_{c}^{i}\left(\mathscr{Z}^{1}, i^{1*}j_{*}^{1}\mathbb{C}\right) = I_{P}^{G}\left(\operatorname{colim}_{\Gamma}\mathbb{H}_{c}^{i}(S_{\Gamma}, i_{S_{\Gamma}}^{*}j_{\Gamma*}\mathbb{C}_{M_{\Gamma}})\right) \\
= \bigoplus_{k} I_{P}^{G}(\operatorname{colim}_{\Gamma}\mathbb{H}_{c}^{i-k}(S_{\Gamma}, \mathbb{H}^{k}(i_{S_{\Gamma}}^{*}j_{\Gamma*}\mathbb{C}_{M_{\Gamma}}))) \\
= \bigoplus_{k} I_{P}^{G}(\operatorname{colim}_{\Gamma}\mathbb{H}_{c}^{i-k}(S_{\Gamma}, \mathbb{H}^{k}(\mathfrak{w}, \mathbb{C}))) \\
= \bigoplus_{k} I_{P}^{G}\mathbb{H}_{c}^{i-k}(\mathscr{M}_{M}, \mathbb{H}^{k}(\mathfrak{w}, \mathbb{C})).$$
(7.8)

Here the first equality is an elementary argument keeping track of the cusps (see [35, Lemma 3.3] for a similar argument for ball quotients) and the second is given by Proposition 2.7. We have also used the fact that $M_{\ell}(\mathbb{R})$ is compact, so that $\Gamma_{M_{\ell}A} = \{e\}$ for neat Γ . There is a similar expression for the $H(\mathbb{Q})$ -module $\mathbb{H}_c^i(\mathscr{Z}_H^1, i_H^{1*}j_{H*}^1\mathbb{C})$ and hence for the target of expression (7.7), namely

$$I_{H}^{G}\mathbb{H}_{c}^{i}\left(\mathscr{Z}_{H}^{1},i_{H}^{1*}j_{H*}^{1}\mathbb{C}\right) = \bigoplus_{k} I_{H}^{G}I_{P_{H}}^{H}\left(\operatorname{colim}_{\Gamma_{H}}\mathbb{H}_{c}^{i-k}\left(S_{\Gamma_{H}},\mathbb{H}^{k}(\mathfrak{w}_{H},\mathbb{C})\right)\right).$$

By the transitivity of induction $I_{H}^{G}I_{P_{H}}^{H} = I_{P}^{G}I_{P_{H}}^{P}$ we are reduced to showing that

$$\operatorname{colim}_{\Gamma} \operatorname{H}^{i-k}_{c} \left(S_{\Gamma}, \operatorname{H}^{k}(\mathfrak{n}, \mathbb{C}) \right) \longrightarrow I^{P}_{P_{H}} \operatorname{colim}_{\Gamma_{H}} \operatorname{H}^{i-k}_{c} \left(S_{\Gamma_{H}}, \operatorname{H}^{k}(\mathfrak{n}_{H}, \mathbb{C}) \right)$$

is injective for $i \leq m-1$. Now the action of $W(\mathbb{Q})$ on the left-hand side is trivial, and

$$\left(I_{P_{H}}^{P}\mathbf{H}_{c}^{i-k}\left(S_{\Gamma},\mathbf{H}^{k}(\mathfrak{n}_{H},\mathbb{C})\right)\right)^{W(\mathbb{Q})}=I_{M_{H}}^{M}\mathbf{H}_{c}^{i-k}\left(S_{\Gamma_{H}},\mathbf{H}^{k}(\mathfrak{n}_{H},\mathbb{C})\right),$$

so we must show that

$$\operatorname{colim}_{\Gamma} \operatorname{H}^{i-k}_{c} \left(S_{\Gamma}, \operatorname{H}^{k}(\mathfrak{n}, \mathbb{C}) \right) \longrightarrow I^{M}_{M_{H}} \operatorname{colim}_{\Gamma_{H}} \operatorname{H}^{i-k}_{c} \left(S_{\Gamma_{H}}, \operatorname{H}^{k}(\mathfrak{n}_{H}, \mathbb{C}) \right)$$

is injective for $i \leq m-1$. But now note that $S_{\Gamma} = S_{\Gamma_H}$ for neat Γ (M and M_H differ only in the compact factor), and the case i = k does not occur (because $\mathrm{H}^0_c(S_{\Gamma}, V) = 0$ for any V), so this follows from Proposition 4.3.

Proof of Proposition 7.3. Recall that the proposition asserts the injectivity of

$$\operatorname{Res}_{\infty}^{0}:\operatorname{Gr}_{i}^{W}\mathbb{H}^{i}\left(\mathscr{Z}^{0},i^{0*}j_{*}\mathbb{C}\right)\longrightarrow I_{H}^{G}\operatorname{Gr}_{i}^{W}\mathbb{H}^{i}\left(\mathscr{Z}_{H}^{0},i_{H}^{0*}j_{H*}\mathbb{C}\right)$$

in degrees $i \leq m-1$. We will reduce this to the Lefschetz property for congruence hyperbolic manifolds – that is, Theorem 5.1.

Let P = MW be the stabilizer of an isotropic line I in V, and F be the associated rational boundary component. Let $i_{s_{\Gamma}} : \{s_{\Gamma}\} \hookrightarrow M_{\Gamma}^1$ be the stratum given by F in M_{Γ}^* . Then by Proposition 2.7 there is an isomorphism in the derived category

$$i_{S_{\Gamma}}^* j_{\Gamma*} \mathbb{C} = \bigoplus_k \mathrm{H}^k \left(i_{S_{\Gamma}}^* j_{\Gamma*} \mathbb{C} \right) [-k],$$

and moreover,

$$\mathrm{H}^{i}\left(i_{s_{\Gamma}}^{*}j_{\Gamma*}\mathbb{C}\right) = \bigoplus_{k} \mathrm{H}^{i-k}\left(\Gamma_{M},\wedge^{k}\mathfrak{u}_{\mathbb{C}}^{*}\right),$$

where we have used the facts that $\Gamma_M = \Gamma_{M_\ell}$ (assuming Γ is neat) is in SO(1, n-1) and $\mathfrak{w} = \mathfrak{u}$ is abelian, so that $\mathrm{H}^*(\mathfrak{w}, \mathbb{C}) = \wedge^* \mathfrak{u}^*$. By Proposition 2.7, the k-summand is pure of weight 2k, so that

$$\operatorname{Gr}_{i}^{W}\operatorname{H}^{i}\left(i_{s_{\Gamma}}^{*}j_{\Gamma*}\mathbb{C}\right) = \operatorname{H}^{i/2}\left(\Gamma_{M}, \wedge^{i/2}\mathfrak{u}_{\mathbb{C}}^{*}\right)$$

if i is even and zero if i is odd.

We then have

$$\begin{aligned} \operatorname{Gr}_{i}^{W} \mathbb{H}^{i}\left(\mathscr{Z}^{0}, i^{0*} j_{*} \mathbb{C}\right) &= I_{P}^{G} \operatorname{colim} \mathrm{H}^{i/2}\left(\Gamma_{M}, \wedge^{i/2} \mathfrak{u}_{\mathbb{C}}^{*}\right) \\ &= I_{P}^{G} \mathrm{H}^{i/2}\left(\mathscr{M}_{M}, \wedge^{i/2} \mathfrak{u}_{\mathbb{C}}^{*}\right) \end{aligned}$$

for *i* even and zero for *i* odd. This discussion applies also to $\operatorname{Gr}_{i}^{W} \mathbb{H}^{i}\left(\mathscr{Z}_{H}^{0}, i_{H}^{0*} j_{H*} \mathbb{C}\right)$, giving

$$\begin{aligned} \operatorname{Gr}_{i}^{W} \mathbb{H}^{i}\left(\mathscr{Z}^{0}, i^{0*} j_{*} \mathbb{C}\right) &= I_{P_{H}}^{H} \operatorname{colim} \operatorname{H}^{i/2}\left(\Gamma_{M_{H}}, \wedge^{i/2} \mathfrak{u}_{H,\mathbb{C}}^{*}\right) \\ &= I_{P}^{G} \operatorname{H}^{i/2}\left(\mathscr{M}_{M_{H}}, \wedge^{i/2} \mathfrak{u}_{H,\mathbb{C}}^{*}\right) \end{aligned}$$

for i even and zero for i odd. By the transitivity of induction, the injectivity of ${\rm Res}^0_\infty$ in degree i is reduced to that of

$$\mathrm{H}^{i/2}\left(\mathscr{M}_{M},\wedge^{i/2}\mathfrak{u}_{\mathbb{C}}^{*}\right)\longrightarrow I_{P_{H}}^{P}\mathrm{H}^{i/2}\left(\mathscr{M}_{M_{H}},\wedge^{i/2}\mathfrak{u}_{H,\mathbb{C}}^{*}\right).$$

The action of $W(\mathbb{Q}) = U(\mathbb{Q})$ on the source is trivial, so this factors through the $U(\mathbb{Q})$ -invariants of the target – that is,

$$\left(I_{P_{H}}^{P}\operatorname{colim} \mathrm{H}^{i/2}\left(\Gamma_{M_{H}},\wedge^{i/2}\mathfrak{u}_{H,\mathbb{C}}^{*}\right)\right)^{U(\mathbb{Q})}=I_{M_{H}}^{M}\operatorname{colim} \mathrm{H}^{i/2}\left(\Gamma_{M_{H}},\wedge^{i/2}\mathfrak{u}_{H,\mathbb{C}}^{*}\right).$$

We are thus reduced to the injectivity of

$$\operatorname{colim} \mathrm{H}^{i/2}\left(\Gamma_{M},\wedge^{i/2}\mathfrak{u}_{\mathbb{C}}^{*}\right)\longrightarrow I_{M_{H}}^{M}\operatorname{colim} \mathrm{H}^{i/2}\left(\Gamma_{M_{H}},\wedge^{i/2}\mathfrak{u}_{H,\mathbb{C}}^{*}\right).$$

Suppose $i \leq m-1$. Then $\wedge^{i/2}\mathfrak{u}_{\mathbb{C}}^*$ is irreducible and $\wedge^{i/2}\mathfrak{u}_{H,\mathbb{C}}^*$ is the M_H -summand containing the M-highest weight vector, so the injectivity of the previous map follows from the Lefschetz property in Theorem 5.1. (The subgroup M_H is never triality.) This concludes the proof.

This concludes the proof of Theorem 7.1.

Appendix A. Some facts about the L^2 cohomology of arithmetic manifolds

Let G be a semisimple algebraic group over \mathbb{Q} , K a maximal compact subgroup of $G(\mathbb{R})$, $X = G(\mathbb{R})/K$, and $\Gamma \subset G(\mathbb{Q})$ a congruence subgroup. (Everything in this appendix should apply to any arithmetic subgroup, but we will use results from [21], which is written in an adelic context, so we make this assumption at the outset.)

The L^2 cohomology of Γ with coefficients in a finite-dimensional algebraic representation E of $G(\mathbb{C})$ is

$$\mathrm{H}^{*}_{(2)}(\Gamma, E) = \mathrm{H}^{*}\left(\mathfrak{g}, K, L^{2}\left(\Gamma \setminus G(\mathbb{R})\right) \otimes E\right).$$

(The (\mathfrak{g}, K) -cohomology of a $G(\mathbb{R})$ -representation (π, V) is, by definition, that of the space V^{∞} of smooth vectors.) This is not the usual definition, which requires looking at the complex of L^2 differential forms with L^2 weak differential (or smooth L^2 forms with L^2 differential), but they are known to agree [16, Prop. 5.4]. The decompositions

$$\begin{split} L^{2}\left(\Gamma\backslash G(\mathbb{R})\right) &= L^{2}_{dis}\left(\Gamma\backslash G(\mathbb{R})\right) \oplus L^{2}_{cts}\left(\Gamma\backslash G(\mathbb{R})\right) \\ &= L^{2}_{cusp}\left(\Gamma\backslash G(\mathbb{R})\right) \oplus L^{2}_{res}\left(\Gamma\backslash G(\mathbb{R})\right) \oplus L^{2}_{cts}\left(\Gamma\backslash G(\mathbb{R})\right) \end{split}$$

of the L^2 spectrum into cuspidal, residual, and continuous parts (given by Langlands' theory of Eisenstein series) induce decompositions

$$\begin{aligned} \mathbf{H}^*_{(2)}(\Gamma, E) &= \mathbf{H}^*_{dis}(\Gamma, E) \oplus \mathbf{H}^*_{cts}(\Gamma, E) \\ &= \mathbf{H}^*_{cusp}(\Gamma, E) \oplus \mathbf{H}^*_{res}(\Gamma, E) \oplus \mathbf{H}^*_{cts}(\Gamma, E), \end{aligned}$$

where $\mathrm{H}_{?}^{*}(\Gamma, E) = \mathrm{H}^{*}(\mathfrak{g}, K, L_{?}^{2}(\Gamma \setminus G(\mathbb{R})) \otimes E)$. The summand $\mathrm{H}_{dis}^{*}(\Gamma, E)$ is identified with the (finite-dimensional) space of *E*-valued L^{2} harmonic forms for the invariant metric (a result due to Borel and Garland; see [16, Prop. 4.4(i)]), whereas the summand $\mathrm{H}_{cts}^{*}(\Gamma, E)$ either vanishes (e.g., when *G* is equal-rank) or is infinite-dimensional [16]. The inclusion $L^{2}(\Gamma \setminus G(\mathbb{R}))^{\infty} \subset C^{\infty}(\Gamma \setminus G(\mathbb{R}))$ induces a natural map

$$\mathrm{H}^*_{(2)}(\Gamma, E) \longrightarrow \mathrm{H}^*(\Gamma, E)$$

which is injective on cuspidal cohomology. Proposition A.1 is well known to experts, but for lack of a suitable reference we give a proof, which is a simple matter of applying results of [21] and [16]. It goes beyond the existing literature [16] only in the cases where $G(\mathbb{R})$ does not have a discrete series – for example, GL(n) – and we need to use it in the main body of the paper for the case SO(1,d), d odd.

Proposition A.1. The map $\mathrm{H}^*_{(2)}(\Gamma, E) \to \mathrm{H}^*(\Gamma, E)$ is zero on $\mathrm{H}^*_{cts}(\Gamma, E)$.

Proof. Following Franke [21], let $S_1(\Gamma \setminus G(\mathbb{R})) \subset L^2(\Gamma \setminus G(\mathbb{R}))$ be the submodule of smooth functions which are uniformly in L^2 – that is, smooth functions f such that $Df \in L^2$ for all $D \in U(\mathfrak{g})$. The inclusion $S_1 \subset L^2$ induces isomorphisms in cohomology (by [21, Theorem 3]), so $S_1 \subset C^\infty$ induces the map in question. It factors as

$$S_1(\Gamma \setminus G(\mathbb{R})) \subset S_{\log}(\Gamma \setminus G(\mathbb{R})) \subset C^{\infty}(\Gamma \setminus G(\mathbb{R})),$$

where $S_{\log} = S_{\log}(\Gamma \setminus G(\mathbb{R}))$ is the space of functions which are uniformly L^2 up to logarithmic terms (see [21, §5] or [49, Section 6.1]). It suffices to show that $\mathrm{H}^*_{cts}(\Gamma, E)$, as a summand of $\mathrm{H}^*(\mathfrak{g}, K, S_1 \otimes E)$, goes to zero in $\mathrm{H}^*(\mathfrak{g}, K, S_{\log} \otimes E)$. We will do this using further reductions to the bounded spectra $S_{1,b} \subset S_1$ and $S_{\log,b} \subset S_{\log}$ with respect to the Casimir operator, a notion introduced in [21, Section 5.1] (compare [49, Section 6.3]). We will show the following:

- 1. $S_{1,b} \subset L^2$ induces an isomorphism $\mathrm{H}^*(\mathfrak{g}, K, S_{1,b} \otimes E) = \mathrm{H}^*_{(2)}(\Gamma, E)$.
- 2. $S_{\log,b} \subset S_{\log}$ induces an isomorphism in cohomology.
- 3. $S_{1,b} \subset S_{\log,b}$ induces zero on $\mathrm{H}^*_{cts}(\Gamma, E)$.

This will prove the proposition.

(1) To show that $S_{1,b} \subset L^2$ induces an isomorphism in cohomology, we use Langlands spectral decomposition of L^2 . There are compatible direct sum decompositions

$$S_{1,b} = \bigoplus_{\{R\}} S_{1,b,\{R\}} \subset L^2 = \bigoplus_{\{R\}} L^2_{\{R\}}$$

indexed by associate classes of parabolic subgroups. The R = G summands are $L_{dis}^{2,\infty}$ and L_{dis}^{2} . (In the usual statement of $\{R\}$ -decompositions, the R = G summand is the cuspidal

1695

part; we are using the obvious rearrangement.) The inclusion $L_{dis}^{2,\infty} \subset L_{dis}^2$ induces an isomorphism by definition, so we must show the same for $S_{1,b,\{P\}} \subset L_{\{P\}}^2$ for proper P. For P = MAN, we have (compare [21, Theorem 11])

$$S_{1,b,\{P\}} = \left(\operatorname{Ind}_{P}^{G} L_{c}^{2}(i\mathfrak{a}^{*}) \otimes A_{2}^{M} \right)^{W_{M}},$$

where A_2^M is the space of L^2 automorphic forms on $\Gamma_M \setminus M(\mathbb{R})$, and the Weyl group W_M of M acts by intertwining operators. The $\{P\}$ -summand of L^2 has the following description (see, e.g., [16, Section 4.3]). For each irreducible summand V of $L^2_{dis}(\Gamma_M \setminus M(\mathbb{R}))$, we have the direct integral

$$E_{P,V} = \int_{\mathfrak{a}^*}^{\oplus} \operatorname{Ind}_P^G \mathbb{C}_{\rho+i\mu} \otimes V.$$

Then $L^2_{\{P\}}$ is the invariants under the action of W_M by intertwining operators on the (countable) Hilbert-space direct sum of $E_{P,V}$ as V varies over all irreducible summands. One could, equivalently, restrict to a subset of V modulo W_M -equivalence and take the sum of direct integrals like this one over the positive Weyl chamber \mathfrak{a}^{*+} , which is the formulation in [16, Section 4.3].

Now $\mathrm{H}^*(\mathfrak{g}, K, E_{P,V} \otimes E) = \{0\}$ unless $E_{P,V}$ shares K-types with $\wedge^*(\mathfrak{g}/\mathfrak{k}) \otimes E^*$ and the Casimir acts by the correct scalar, so it follows that there is a finite set $\{V_i\}_{i \in I}$ of V such that the cohomology becomes a finite sum

$$\mathbf{H}^{*}\left(\mathfrak{g}, K, L^{2}_{\{P\}} \otimes E\right) = \left(\bigoplus_{i \in I} \mathbf{H}^{*}\left(\mathfrak{g}, K, E_{P, V_{i}} \otimes E\right)\right)^{W_{M}}$$

(see [16, Prop. 4.4(ii)]). The same applies to $S_{1,b,\{P\}}$ – that is, we may replace A_2^M by $\bigoplus_{i \in I} A_2^M \cap V_i^\infty$ and get the same cohomology. The computation of a single summand $\mathrm{H}^*(\mathfrak{g}, K, E_{P,V_i} \otimes E)$ is contained in [16, Theorem 3.4], and the answer is similar to the usual computation for induced representations in [17, Theorem III.3.3]: There is a unique $s \in W^P$ such that

$$\mathrm{H}^{*}\left(\mathfrak{g}, K, E_{P, V_{i}} \otimes E\right) = \mathrm{H}^{*-\ell(s)}\left(\mathfrak{m}, K_{M}, V_{i} \otimes E_{s(\lambda+\rho)-\rho}^{M}\right) \otimes \mathrm{H}^{*}\left(\mathfrak{a}, \int_{\mathfrak{a}^{*}}^{\oplus} \mathbb{C}_{i\mu} d\mu\right),$$

where $E_{s(\lambda+\rho)-\rho}^{M}$ is the restriction to M of $E_{s(\lambda+\rho)-\rho}^{MA}$ (notation as in Kostant's theorem in §4.1). The parallel computation for $\operatorname{Ind}_{P}^{G}L_{c}^{2}(i\mathfrak{a}^{*}) \otimes V_{i}$ (by the same arguments as in the proof of [16, Theorem 3.4]) gives the same expression, with $\int_{\mathfrak{a}^{*}}^{\oplus} \mathbb{C}_{i\mu} d\mu$ replaced by $L_{c}^{2}(i\mathfrak{a}^{*})$. The assertion that $S_{1,b,\{P\}} \subset L_{\{P\}}^{2}$ induces an isomorphism now boils down to the assertion that (for each P) the inclusion of $L_{c}^{2}(i\mathfrak{a}^{*}) = \operatorname{colim}_{\Omega\subset\mathfrak{a}^{*}}\int_{\Omega}^{\oplus} \mathbb{C}_{i\mu} d\mu$ (the colimit taken over compact Ω) into the direct integral $\int_{\mathfrak{a}^{*}}^{\oplus} \mathbb{C}_{i\mu} d\mu$ is an isomorphism in \mathfrak{a} -cohomology. This elementary fact follows, for example, from [16], which shows that this is already true of $\int_{\Omega}^{\oplus} \mathbb{C}_{i\mu} d\mu \subset \int_{\mathfrak{a}^{*}}^{\oplus} \mathbb{C}_{i\mu} d\mu$ if $0 \in \Omega$.

(2) To show that the inclusion $S_{\log,b} \subset S_{\log}$ induces an isomorphism in cohomology, we simply combine [21, Theorem 10] and the spectral sequence in [21, Theorem 7]. (This may not be the simplest or most natural proof, but it is certainly the shortest to write down here!)

(3) We are reduced to considering the inclusion $S_{1,b} \subset S_{\log,b}$. Now $S_{\log,b}$ has a spectral decomposition analogous to that of $S_{1,b}$, in which the $\{P\}$ -summand for P = MAN is

$$\left(\operatorname{Ind}_P^G D'_c(i\mathfrak{a}^*) \otimes A_2^M\right)^{W_M},$$

where $D'_c(\mathfrak{ia}^*)$ is the space of compactly supported distributions (compare [21, Theorem 12]). The P = G summand is still $L^{2,\infty}$. So the fact that $S_{1,b,\{P\}} \subset S_{\log,b,\{P\}}$ induces zero for P proper boils down to the fact that the inclusion of \mathfrak{a} -modules $L^2_c(\mathfrak{ia}^*) \subset D'_c(\mathfrak{ia}^*)$ induces zero in \mathfrak{a} -cohomology. This is immediate: The first has cohomology in degrees in [1, dim \mathfrak{a}] (by [16, Prop. 3.2], as already remarked), and the latter has cohomology only in degree 0 (e.g., by [21, Lemma 1]; this reduces to the fact that the complex $D'_c(\mathbb{R}) \xrightarrow{x} D'_c(\mathbb{R})$ has cohomology only in degree 0). This concludes the proof of the proposition.

The preceding proof used three results [21, Theorems 10, 12, and 13]) whose proofs constitute the technical heart of Franke's work. It is possible that they can be avoided, but the method of proof gives rather more, as we now show. The results which follow are not used in the body of the paper, but will be useful elsewhere. The following is a corollary of the proof of the proposition:

Corollary A.2. If E is rationally defined (i.e., has a rational structure preserved by $G(\mathbb{Q})$) then the square-integrable cohomology, which is (by definition) the image of $\mathrm{H}^*_{(2)}(\Gamma, E) \to \mathrm{H}^*(\Gamma, E)$, is a rational subspace.

Proof. By [33, Theorem A], the cohomology of S_{\log} is isomorphic to the lower middle weighted cohomology of [23], and this has a rational structure ([23, Chapter IV]) compatible with the map to $H^*(\Gamma, E)$.

In fact, we can refine this statement somewhat using the same methods. Recall that there is a subspace $S_{-\log}(\Gamma \setminus G(\mathbb{R})) \subset S_{\log}(\Gamma \setminus G(\mathbb{R}))$ defined by using a condition dual to the one defining S_{\log} (see [21, §5] or [49, Section 6.1]).

Proposition A.3. The image of $H^*(\mathfrak{g}, K, S_{-\log} \otimes E) \to H^*(\mathfrak{g}, K, S_{\log} \otimes E)$ is identified with $H^*_{dis}(\Gamma, E)$ or, equivalently, with the space of E-valued L^2 harmonic forms.

Proof. This was proved in [33, Theorem B] under the assumption that G is equal-rank, in which case the map in the proposition is an isomorphism and both groups compute the L^2 cohomology. In general, we argue as follows. By [21, Theorem 10] and the spectral sequence of [21, Theorem 7], we know that $S_{\pm \log,b} \subset S_{\pm \log}$ induce isomorphisms in (\mathfrak{g}, K) cohomology, so it is enough to consider the inclusion $S_{-\log,b} \subset S_{\log,b}$. For these spaces there are compatible decompositions

$$S_{-\log,b} = \bigoplus_{\{P\}} S_{-\log,b,\{P\}} \subset S_{\log,b} = \bigoplus_{\{P\}} S_{\log,b,\{P\}}$$

indexed by associate classes of parabolic subgroups. The $\{P\}$ -summand for $S_{-\log,P}$ is

$$\left(\operatorname{Ind}_P^G C_c^{\infty}(i\mathfrak{a}^*) \otimes A_2^M\right)^{W_M}$$

The $\{P\}$ -summand for $S_{\log,b}$ was recalled in the proof of the previous proposition and amounts to replacing $C_c^{\infty}(\mathfrak{ia}^*)$ by $D_c'(\mathfrak{ia}^*)$ in this expression; the map $S_{-\log,b,\{P\}} \subset S_{\log,b,\{P\}}$ is induced by $C_c^{\infty}(\mathfrak{ia}^*) \subset D_c'(\mathfrak{ia}^*)$. The map in \mathfrak{a} -cohomology induced by this inclusion is zero, since $\mathrm{H}^*(\mathfrak{a}, C_c^{\infty}(\mathfrak{ia}^*))$ is concentrated in degree dim \mathfrak{a} while $\mathrm{H}^*(\mathfrak{a}, D_c'(\mathfrak{ia}^*))$ is concentrated in degree 0. By the standard computations of cohomology for induced representations (recalled earlier in the proof of the previous proposition), we see that $S_{-\log,b,\{P\}} \subset S_{\log,b,\{P\}}$ is zero in cohomology for $P \neq G$. Since the P = G summands are both identified with $L_{dis}^{2,\infty}$, the first statement follows. \Box

The previous corollary is refined by the following:

Corollary A.4. If E is rationally defined, then the space of square-integrable E-valued harmonic forms on $\Gamma \setminus X$ has a canonical (Betti) rational structure.

Proof. By [33, Theorem A], the groups in the proposition are the upper and lower middle weighted cohomology groups, which have natural \mathbb{Q} -structures ([23, Chapter IV]).

Remark A.5. In contrast to the corollary, the space of cuspidal harmonic forms, which is simply the cuspidal cohomology, should not be expected to be Betti rational in general, for example, for Sp(4). Of course, it is well known to be Betti rational in the case of GL(n)[21, 7.6] and some related cases – for example, for SO(1,d) it follows from Lemma 5.2.

Remark A.6. When $X = G(\mathbb{R})/K$ has an Hermitian structure, something much stronger than the corollary is true thanks to equation (3.12), namely that the space of L^2 harmonic forms is part of a mixed realization over the number field of definition of $\Gamma \setminus X$ (the reflex field, if we work in the context of Shimura varieties). In particular, it has both Betti and de Rham rational structures.

Appendix B. Residual Eisenstein cohomology in corank 1

We summarize here some very well-known facts (essentially going back to [26]) on the construction of cohomology via residual Eisenstein series from cuspidal data on maximal parabolic subgroups. In the body of the paper these are applied to the rank 1 groups SO(1,n) and SU(1,n).

For a $G(\mathbb{R})$ -representation V, the smooth vectors are denoted V^{∞} and $\mathrm{H}^*(\mathfrak{g}, K, V)$ is the (\mathfrak{g}, K) -cohomology of V^{∞} . For a (\mathfrak{g}, K) -module or $G(\mathbb{R})$ -representation V, let

$$d_{\min}(V) = \min\left\{i: \mathrm{H}^{i}(\mathfrak{g}, K, V) \neq 0\right\}, \qquad d_{\max}(V) = \max\left\{i: \mathrm{H}^{i}(\mathfrak{g}, K, V) \neq 0\right\},$$

assuming these make sense and are finite.

Let $P = LN \subset G$ be a maximal parabolic subgroup and A the Q-split part of the center of L. Let $\sigma \subset L^2_{cusp}(\Gamma_L A(\mathbb{R})^0 \setminus L(\mathbb{R}))$ be a cuspidal automorphic form on L. For $\lambda \in \mathfrak{a}_{\mathbb{C}}^*$, let $I_{\lambda} = \operatorname{Ind}_P^G \otimes \mathbb{C}_{\lambda}$ be the (normalized) induced representation. The theory of Eisenstein series produces a (\mathfrak{g}, K) -homomorphism to the space of automorphic forms,

$$\mathbf{E}: I_{\lambda}^{\infty} \to \mathscr{A}(\Gamma \backslash G(\mathbb{R})),$$

which is meromorphic in $\lambda \in \mathfrak{a}_{\mathbb{C}}^*$. If λ is a pole of **E**, meaning that the Eisenstein series $E(\phi, \lambda)$ has a pole for generic ϕ in the space of σ , then P is self-associate and the pole

is real and simple if $Re(\lambda) \in (\mathfrak{a}^*)^+$. Taking the residue at such a λ defines a residual Eisenstein operator

$$\mathbf{E}^*: I^{\infty}_{\lambda} \to \mathscr{A}(\Gamma \backslash G(\mathbb{R})).$$

By [21], the (\mathfrak{g}, K) -cohomology of $\mathscr{A}(\Gamma \setminus G(\mathbb{R})) \otimes E$ is $\mathrm{H}^*(\Gamma, E)$.

Lemma B.1. Suppose that P, σ are as discussed and that \mathbf{E} has a pole at λ with $Re(\lambda) \in (\mathfrak{a}^*)^+$, so that λ is real. Suppose that the Langlands quotient J_{λ} of I_{λ} is cohomological with respect to a coefficient representation E, and further that

(*) $d_{\max}(I_{\lambda} \otimes E) = d_{\max}(J_{\lambda} \otimes E)$ and $\mathrm{H}^{d_{\max}(I_{\lambda} \otimes E)}(\mathfrak{g}, K, I_{\lambda} \otimes E) \to \mathrm{H}^{d_{\max}(J_{\lambda} \otimes E)}(\mathfrak{g}, K, J_{\lambda} \otimes E)$ is an isomorphism

(see Remark B.2). Then the map in cohomology induced by $\mathbf{E}^*(I_{\lambda}^{\infty}) \subset \mathscr{A}$ in degree $d_{\min}(J_{\lambda} \otimes E)$ is injective and the nonzero classes in the subspace

$$\mathrm{H}^{d_{\min}(J_{\lambda}\otimes E)}(\mathfrak{g}, K, \mathbf{E}^{*}(I_{\lambda}^{\infty})\otimes E) \subset \mathrm{H}^{d_{\min}(J_{\lambda}\otimes E)}(\Gamma, E)$$

restrict nontrivially to the boundary – that is, do not belong to $H^*(\Gamma, E)$.

Proof. We write I, J for I_{λ}, J_{λ} and ignore the coefficients E, as they are not relevant. The residue of a cuspidal Eisenstein series at a point of the positive Weyl chamber is well known to be square-integrable, so that $\mathbf{E}^*(I^{\infty}) \subset \mathscr{A} \cap L^2_{dis}$, and as an abstract representation, $\mathbf{E}^*(I^{\infty})$ is the Langlands quotient J^{∞} . Taking the constant term of automorphic forms along P defines a mapping

$$I^{\infty} \xrightarrow{\mathbf{E}^*} \mathscr{A} \longrightarrow I^{*,\infty},$$

where $I^* = \operatorname{Ind}_P^G \sigma^* \otimes \mathbb{C}_{-\lambda}$ is the contragredient of I. (The usual expression for the constant term defines maps $I_{\lambda} \to \mathscr{A} \to I_{\lambda} \oplus I_{\lambda}^*$ for generic λ , but for the residual operator at a pole only the second term is nonvanishing.) The composite is a nonzero multiple of the standard interwining operator $I \to I^*$, the image of which is the Langlands quotient J, and the factoring is exactly $I^{\infty} \to J^{\infty} \subset I^{*,\infty}$.

Now by assumption $d_{\max}(I) = d_{\max}(J)$, and so by duality ([17, Proposition I.7.6] for the irreducible unitary module J and [17, Theorem III.3.3 and Proposition I.7.6] for I) we have

$$d_{\min}(J) = \dim X - d_{\max}(J) = \dim G(\mathbb{R})/K - d_{\max}(I) = d_{\min}(I^*)$$

and $\mathrm{H}^{d_{\min}(J)}(\mathfrak{g}, K, J) \cong \mathrm{H}^{d_{\min}(J)}(\mathfrak{g}, K, I^*)$. Moreover, for a class in $\mathrm{H}^{d_{\min}(J)}(\mathfrak{g}, K, \mathbf{E}^*(I))$ in $\mathrm{H}^*(\Gamma, \mathbb{C})$, the induced mapping

$$C^*(\mathfrak{g}, K, \mathbf{E}^*(I^\infty)) \to C^*(\mathfrak{g}, K, I^{*,\infty})$$

gives, via the identification of $\mathrm{H}^*(\mathfrak{g}, K, I^{*,\infty})$ with a summand of the cohomology of the *P*-boundary, the restriction of the class to the *P*-boundary. (This is contained in [26] in a differential-geometric language and in [45] in representation-theoretic terms.) The restriction is therefore nonzero, because $\mathrm{H}^{d_{\min}(J)}(\mathfrak{g}, K, J) \cong \mathrm{H}^{d_{\min}(J)}(\mathfrak{g}, K, I^*)$, and so $\mathrm{H}^{d_{\min}(J)}(\mathfrak{g}, K, \mathbf{E}^*(I)) \to \mathrm{H}^*(\Gamma, \mathbb{C})$ is injective. The classes in this subspace survive on restriction to the boundary, so they are not in interior cohomology. \Box

Remark B.2. In fact, (*) always holds for a unitary cohomological Langlands quotient, but rather than prove this general fact, we just check it in the cases of interest where the lemma is applied.

Appendix C. Chern classes of automorphic vector bundles

We are in the situation of §3.2: *G* is semisimple and simply connected, $X = G(\mathbb{R})/K$ is Hermitian, and $M_{\Gamma} = \Gamma \setminus X$. Fix a smooth toroidal compactification $M_{\Gamma} \hookrightarrow M_{\Gamma}^{\Sigma}$ in which the boundary is a simple normal crossings divisor [3]. Let Rep(*H*) denote the category of finite-dimensional representations of a compact group *H*. With *E* in Rep(*K*) are associated the homogeneous bundle \mathscr{E}^c on X^c , the automorphic vector bundle \mathscr{E}_{Γ} on $M_{\Gamma} = \Gamma \setminus X$, and the canonical extension $\mathscr{E}_{\Gamma}^{\Sigma}$ of \mathscr{E}_{Γ} to M_{Γ}^{Σ} . The following statement is well known to experts; we borrow the proof from [34, Lemma 3.7.2]):

Lemma C.1. There is an injective ring homomorphism θ : $\mathrm{H}^*(X^c, \mathbb{Q}) \to \mathrm{H}^*(M^{\Sigma}_{\Gamma}, \mathbb{Q})$ with $\theta(c_k(\mathscr{E}^c)) = (-1)^k c_k(\mathscr{E}^{\Sigma}_{\Gamma})$ for all $E \in \mathrm{Rep}(K)$.

Proof. Following a suggestion of N. Fakhruddin we will use K-theory to prove this. Let $K^0(-)$ denote the topological K-theory of a space and $ch: K^0(-) \to H^*(-,\mathbb{Q})$ the Chern character homomorphism. We write R(H) for the representation ring of a compact group H – that is, the Grothendieck group of the category $\operatorname{Rep}(H)$.

We first show that the ring homomorphism $R(K) \to \mathrm{H}^*(M^{\Sigma}_{\Gamma}, \mathbb{Q})$ defined by $V \mapsto ch(\mathscr{V}^{\Sigma}_{\Gamma})$ and extended \mathbb{Q} -linearly defines a ring homomorphism

$$\kappa: K^0(X^c) \otimes \mathbb{Q} \to \mathrm{H}^*\left(M^{\Sigma}_{\Gamma}, \mathbb{Q}\right). \tag{C.1}$$

Since $X^c = G(\mathbb{R})^c / K$ with $G(\mathbb{R})^c$ simply connected (it is the maximal compact of the simply connected group $G(\mathbb{C})$) and $K \subset G(\mathbb{R})^c$ is a subgroup of maximal rank, the construction $V \mapsto \mathscr{V}^c$ gives an isomorphism

$$R(K) \otimes_{R(G(\mathbb{R})^c)} \mathbb{Z} \to K^0(X^c),$$

where \mathbb{Z} is an $R(G(\mathbb{R})^c)$ -module via the dimension homomorphism (by [40, Theorem 3]). Since the left-hand side is the quotient of R(K) by the ideal generated by ker(dim : $R(G(\mathbb{R})^c) \to \mathbb{Z})$ and ch is a ring homomorphism, to show that κ is well defined it suffices to show that $ch\left(\mathscr{E}^{\Sigma}\right) = \dim E$ if $E \in \operatorname{Rep}(G(\mathbb{R})^c)$. The degree 0 term of the Chern character of a bundle is its rank, so it suffices to check that $c_k\left(\mathscr{E}_{\Gamma}^{\Sigma}\right) = 0$ for k > 0 for such E. Now Mumford showed that the kth Chern form of the invariant or Nomizu connection defines a current on M_{Γ}^{Σ} which represents (up to a factor of $(2\pi\sqrt{-1})^k$) the Chern class $c_k\left(\mathscr{E}_{\Gamma}^{\Sigma}\right)$ ([32, Theorem 3.1 and Theorem 1.4]). But if E is a $G(\mathbb{R})^c$ -representation, the curvature 2-form of the Nomizu connection vanishes identically (see, e.g., [25, Proposition 5.3]), hence so do its Chern forms for k > 0. Thus $c_k\left(\mathscr{E}_{\Gamma}^{\Sigma}\right) = 0$ for k > 0 and $ch\left(\mathscr{E}_{\Gamma}^{\Sigma}\right) = \dim E$, and we have κ as in formula (C.1). It is a ring homomorphism because canonical extension is compatible with the tensor product [27, Section 4.2] and the Chern character is multiplicative. Since X^c is a flag variety, it has only even-degree cohomology, so the Chern character gives an isomorphism $ch: K^0(X^c) \otimes \mathbb{Q} \to \mathrm{H}^*(X^c, \mathbb{Q})$ (compare [4, 2.4]). Now define

$$\theta := \kappa \circ ch^{-1} \circ \sigma,$$

where $\sigma: \mathrm{H}^*(X^c, \mathbb{Q}) \to \mathrm{H}^*(X^c, \mathbb{Q})$ is defined by $\sigma(\alpha) = (-1)^{\mathrm{deg}(\alpha)/2} \alpha$. Since $\mathrm{H}^*(X^c, \mathbb{Q})$ is concentrated in even degrees, this makes sense, and σ is a ring homomorphism. Note that $\theta(\sigma(ch(\mathscr{E}^c))) = ch(\mathscr{E}_{\Gamma}^{\Sigma})$, from which it follows that $\theta(c_k(\mathscr{E}^c)) = (-1)^k c_k(\mathscr{E}_{\Gamma}^{\Sigma})$. This implies that θ is injective (i.e., nonzero) in top degree $2n = 2 \dim_{\mathbb{C}} X$: Choose a nonzero monomial $c_{k_1}(\mathscr{E}_{\Gamma}^c) \cdots c_{k_r}(\mathscr{E}_{\Gamma}^c)$ with $\sum_i k_i = n$; it spans $\mathrm{H}^{2n}(X^c, \mathbb{Q})$. By Mumford's version of proportionality [32, Theorem 3.2],

$$\left[M_{\Gamma}^{\Sigma}\right] \cap \theta\left(c_{k_{1}}\left(\mathscr{E}_{1}^{c}\right)\cdots c_{k_{r}}\left(\mathscr{E}_{r}^{c}\right)\right) = (-1)^{n} \cdot C \cdot \left[X^{c}\right] \cap c_{k_{1}}\left(\mathscr{E}_{1}^{c}\right)\cdots c_{k_{r}}\left(\mathscr{E}_{r}^{c}\right) \neq 0,$$

where C is a nonzero constant independent of k_i, \mathscr{E}_i . It follows that θ is injective: For nonzero $\alpha \in \mathrm{H}^i(X^c)$, choose $\beta \in \mathrm{H}^{2n-i}(X^c)$ such that $\alpha \cdot \beta \neq 0$. Then $0 \neq \theta(\alpha \cdot \beta) = \theta(\alpha) \cdot \theta(\beta)$, so that $\theta(\alpha) \neq 0$.

Acknowledgments We thank Najmuddin Fakhruddin, Dipendra Prasad, and T. N. Venkataramana for helpful discussions, DAE for support under PIC 12-R&D-TFR-5.01-0500, and the referee for helpful comments.

Competing Interest. The author(s) declare none.

References

- J. ARTHUR, Unipotent automorphic representations: Conjectures, Astérisque 171-172 (1989), 13-71.
- J. ARTHUR, The Endoscopic Classification of Representations: Orthogonal and Symplectic Groups, AMS Colloquium Publications, 61 (American Mathematical Society, Providence, RI, 2013).
- [3] A. ASH, D. MUMFORD, M. RAPOPORT AND Y.-S. TAI, Smooth Compactification of Locally Symmetric Varieties, 2nd ed. (Cambridge University Press, Cambridge, UK, 2010).
- [4] M. F. ATIYAH AND F. HIRZEBRUCH, Vector bundles and homogeneous spaces, in Differential Geometry, Proceedings of Symposia in Pure Mathematics, 3, pp. 7–38 (American Mathematical Society, Providence, RI, 1961).
- [5] W. BAILY AND A. BOREL, Compactification of arithmetic quotients of bounded symmetric domains, Ann. of Math. (2) 84 (1966), 442–528.
- [6] A. BEILINSON, J. BERNSTEIN AND P. DELIGNE, Faisceaux Pervers, Astérisque 100 (1982).
- [7] N. BERGERON, Lefschetz properties for arithmetic real and complex hyperbolic manifolds, Int. Math. Res. Not. IMRN 20 (2003) 1089–1122.
- [8] N. BERGERON, Propriétés de Lefschetz automorphes pour les groupes unitaires et orthogonaux, *Mém. Soc. Math. Fr. (N.S.)* **106** (2006).
- [9] N. BERGERON, Restriction de la cohomologie d'une variété de Shimura à ses sous-variétés, *Transform. Groups* 14 (2009), 41–86.
- [10] N. BERGERON, Hodge theory and cycle theory of locally symmetric spaces, in *Proceedings of the International Congress of Mathematicians 2018*, pp. 831–857 (World Scientific, Singapore, 2019).

- [11] N. BERGERON AND L. CLOZEL, Spectre automorphe des variétés hyperboliques et applications topologiques Astérisque **303** (2005).
- [12] N. BERGERON AND L. CLOZEL, Quelques conséquences des travaux d'Arthur pour le spectre et la topologie des variétés hyperboliques, *Invent. Math.* **192** (2013), 505–532.
- [13] N. BERGERON AND L. CLOZEL, Sur le spectre et la topologie des variétés hyperboliques de congruence: Les cas complexe et quaternionien, Math. Ann. 368 (2017), 1333–1358.
- [14] N. BERGERON AND L. CLOZEL, Sur la cohomologie des variétés hyperboliques de dimension 7 trialitaires, *Israel J. Math.* 222 (2017), 333–400.
- [15] R. E. BORCHERDS, Automorphic forms on $O_{s+2,2}(\mathbf{R})$ and infinite products, *Invent. Math.* **120** (1995), 161–213.
- [16] A. BOREL AND W. CASSELMAN, L² cohomology of locally symmetric manifolds of finite volume, Duke Math. J. 50 (1983), 625–647.
- [17] A. BOREL AND N. WALLACH, Continuous Cohomology, Discrete Subgroups, and Representations of Reductive Groups, 2nd ed., Mathematical Surveys and Monographs, 67 (American Mathematical Society, Providence, RI, (1999).
- [18] M. BURGER AND P. SARNAK, Ramanujan duals II, Invent. Math. 106 (1991), 1–11.
- [19] J. I. BURGOS AND J. WILDESHAUS, Hodge modules on Shimura varieties and their higher direct images in the Baily-Borel compactification, Ann. Sci. Éc. Norm. Supér. (4) 37 (2004), 363–413.
- [20] P. DELIGNE, Théorie de Hodge III, Publ. Math. Inst. Hautes Études Sci. 44 (1974), 5–77.
- [21] J. FRANKE, Harmonic analysis in weighted L₂ spaces, Ann. Sci. Éc. Norm. Supér. (4) 31 (1998), 181–279.
- [22] R. GOODMAN AND N. WALLACH, Symmetry, Representations, and Invariants, Graduate Texts in Mathematics, 255 (Springer, New York, NY, 2009).
- [23] M. GORESKY, G. HARDER, AND R. MACPHERSON, Weighted cohomology, Invent. Math. 116 (1994), 139–213.
- [24] M. GORESKY AND R. MACPHERSON, Stratified Morse Theory, Ergebnisse der Mathematik und ihrer Grenzgebiete, 14 (Springer-Verlag, New York, NY, 1988).
- [25] M. GORESKY AND W. PARDON, Chern classes of automorphic vector bundles, *Invent. Math.* 147 (2002), 561–612.
- [26] G. HARDER, On the cohomology of discrete arithmetically defined groups, in *Discrete Subgroups of Lie Groups (Bombay, 1973)*, pp. 129–160 (Oxford University Press, Bombay, 1975).
- [27] M. HARRIS, Functorial properties of toroidal compactifications of locally symmetric varieties, Proc. Lond. Math. Soc. (3) 59 (1989), 1–22.
- [28] M. HARRIS AND J.-S. LI, A Lefschetz property for subvarieties of Shimura varieties, J. Algebraic Geom. 7 (1998), 77–122.
- [29] B. KOSTANT, Lie algebra cohomology and the Borel-Weil theorem, Ann. of Math. (2) 74 (1961), 329–387.
- [30] E. LOOIJENGA, L^2 cohomology of locally symmetric varieties, *Compos. Math.* **23** (1988), 1–20.
- [31] E. LOOIJENGA AND M. RAPOPORT, Weights in the local cohomology of a Baily-Borel compactification, in *Complex Geometry and Lie Theory, Proceedings of Symposia in Pure Mathematics*, 53, pp. 223–260 (American Mathematical Society, Providence, RI, 1991).
- [32] D. MUMFORD, Hirzebruch's proportionality theorem in the noncompact case, *Invent. Math.* 42 (1977), 239–272.
- [33] A. NAIR, Weighted cohomology of arithmetic groups, Ann. of Math. (2) 150 (1999), 1–31.

A. N. Nair and A. Rai

- [34] A. NAIR, 'Chern classes of automorphic vector bundles and the reductive Borel-Serre compactification', Preprint, 2014, http://www.math.tifr.res.in/~arvind.
- [35] A. NAIR, Lefschetz properties for noncompact arithmetic ball quotients, J. Reine Angew. Math. 730 (2017), 163–198.
- [36] A. NAIR, Lefschetz properties for noncompact arithmetic ball quotients II, Manuscripta Math. 152 (2017), 443–457.
- [37] A. NAIR, Motivic and automorphic aspects of the reductive Borel-Serre compactification, in *Hodge Theory and L²Analysis (Volume in Honour of S. Zucker)*, Advanced Lectures in Mathematics, 39, pp. 473–485 (International Press, Somerville, MA, 2017).
- [38] T. ODA, A note on the Albanese variety of an arithmetic quotient of the complex hyperball, J. Fac. Sci. Univ. Tokyo 28 (1981), 481–486.
- [39] R. PINK, On *l*-adic sheaves on Shimura varieties and their higher direct images in the Baily-Borel compactification, *Math. Ann.* 292 (1992), 197–240.
- [40] H. V. PITTIE, Homogeneous vector bundles on homogeneous spaces, *Topology* 11 (1972), 199–203.
- [41] J. ROHLFS AND B. SPEH, Representations with cohomology in the discrete spectrum of SO(n,1)(ℤ) and Lefschetz numbers, Ann. Sci. Éc. Norm. Supér. (4) 20 (1987), 89–136.
- [42] M. SAITO, Mixed Hodge modules, Publ. Res. Inst. Math. Sci. 26 (1990), 221–333.
- [43] L. SAPER AND M. STERN, L² cohomology of arithmetic varieties, Ann. of Math. (2) 139 (1990), 1–69.
- [44] I. SATAKE, Algebraic Structures of Symmetric Domains, Iwanami Shoten, Tokyo; (Princeton University Press, Princeton, NJ, 1980).
- [45] J. SCHWERMER, Kohomologie arithmetisch definierte Gruppen und Eisensteinreihen, Lecture Notes in Mathematics, 988 (Springer-Verlag, New York, NY, 1983).
- [46] J. TITS, Classification of algebraic semisimple groups, in Algebraic Groups and Discontinuous Subgroups, Proceedings of Symposia in Pure Mathematics, 9, pp. 33–62 (American Mathematical Society, Providence, RI, 1966).
- [47] T. N. VENKATARAMANA, Cohomology of compact locally symmetric spaces, Compos. Math. 125 (2001), 221–253.
- [48] T. N. VENKATARAMANA, Cohomology of arithmetic groups and representations, in Proceedings of the International Congress of Mathematicians 2010, Vol. III, pp. 1366– 1375 (Hindustan Book Agency, New Delhi, 2010).
- [49] J.-L. WALDSPURGER, Cohomology des espaces des formes automorphes (d'aprés J. Franke), Séminaire Bourbaki 1995/1996, Exp. No. 809, Astérisque 241 (1997), 139–156.
- [50] N. WALLACH, On the constant term of a square-integrable automorphic form, in Operator Algebras and Group Representations, Vol. II (Neptun, 1980), Monographs and Studies in Mathematics, 18, pp. 227–237 (Pitman, Boston, MA, 1984).
- [51] R. WEISSAUER, Differentialformen zu Untergruppen der Siegelschen Modulgruppe zweiten Grades, J. Reine Angew. Math. 391 (1988), 100–156.
- [52] S. ZUCKER, The Hodge structures on the intersection homology of varieties with isolated singularities, Duke Math. J. 55 (1987), 603–616.